### NOVEL IMPEDANCE TUNER, PHASE SHIFTER, AND VECTOR MODULATORS USING RF MEMS TECHNOLOGY

## A THESIS SUBMITTED TO THE GRADUATE SCHOOL OF NATURAL AND APPLIED SCIENCES OF MIDDLE EAST TECHNICAL UNIVERSITY

ΒY

MEHMET ÜNLÜ

## IN PARTIAL FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY IN ELECTRICAL AND ELECTRONICS ENGINEERING

**MARCH 2009** 

#### Approval of the thesis:

# NOVEL IMPEDANCE TUNER, PHASE SHIFTER, AND VECTOR MODULATORS

#### USING RF MEMS TECHNOLOGY

submitted by **MEHMET ÜNLÜ** in partial fulfillment of the requirements for the degree of **Doctor of Philosophy in Electrical and Electronics Engineering Department, Middle East Technical University** by,

Prof. Dr. Canan Özgen Dean, Graduate School of <b>Natural and Applied Sciences</b>	
Prof. Dr. İsmet Erkmen	
Assoc Brof Dr. Simsek Domin	
Supervisor, Electrical and Electronics Eng. Dept., METU	
Prof. Dr. Tayfun Akın Co-Supervisor, <b>Electrical and Electronics Eng. Dept., METU</b>	
Examining Committee Members:	
Prof. Dr. Altunkan Hızal Electrical and Electronics Eng. Dept., METU	
Assoc. Prof. Dr. Şimşek Demir Electrical and Electronics Eng. Dept., METU	
Prof. Dr. Tayfun Akın Electrical and Electronics Eng. Dept., METU	
Assoc. Prof. Dr. Sencer Koç Electrical and Electronics Eng. Dept., METU	
Prof. Dr. Hayrettin Köymen Electrical and Electronics Eng. Dept., Bilkent University	

Date:

06.03.2009

I hereby declare that all information in this document has been obtained and presented in accordance with academic rules and ethical conduct. I also declare that, as required by these rules and conduct, I have fully cited and referenced all material and results that are not original to this work.

Name, Last Name: Mehmet ÜNLÜ

Signature :

## ABSTRACT

## NOVEL IMPEDANCE TUNER, PHASE SHIFTER, AND VECTOR MODULATORS USING RF MEMS TECHNOLOGY

Ünlü, Mehmet

Ph. D., Department of Electrical and Electronics EngineeringSupervisor: Assoc. Prof. Dr. Şimşek DemirCo-Supervisor: Prof. Dr. Tayfun Akın

March 2009, 268 pages

This thesis presents the theory, design, fabrication, and measurement results of novel reconfigurable impedance tuner, phase shifter, and vector modulators using the RF MEMS technology. The presented circuits are based on triple stub topology, and it is shown both theoretically and experimentally in this thesis that it is possible to control the insertion phase and amplitude of the input signal simultaneously using this topology. The presented circuits are implemented using an in-house, surface micromachining fabrication process developed at METU, namely METU RF MEMS Fabrication Process, which is implemented using six masks on quartz substrates. The RF MEMS impedance tuner is designed to operate in 6-20 GHz frequency band, and it covers the Smith Chart with 1331 impedance points. The measurement results of 729 impedance points of the fabricated impedance tuner show that a wide Smith Chart coverage is obtained in the entire band.

The RF MEMS phase shifter is designed to cover 0-360° range 10° steps at 15 GHz center frequency. The measurement results of the fabricated phase shifter show that the average phase error is 1.7°, the average insertion loss is -3.1 dB, and the average return loss is -19.3 dB for the measured 21 phase states. The phase shifter can also work up to 30 GHz and 40 GHz with average insertion losses of -5 dB and -8 dB, respectively. The designed RF MEMS vector modulator operates in 22.5-27.5 GHz band, and it has 3 amplitude and 8 phase states. The measurement results of the fabricated vector modulator show that the amplitude error is 0.5 dB, the phase error is 4°, and the return loss is -15 dB on average among the 24 measured states at each of 22.5, 25, and 27.5 GHz frequencies.

The results presented in this thesis show once more that the METU RF MEMS Fabrication Process is mature and successful on implementing complicated RF MEMS structures.

Keywords: RF MEMS, reconfigurable impedance tuner, phase shifter, vector modulator, MEMS switch.

# RF MEMS TEKNOLOJİSİ İLE YENİ EMPEDANS AYARLAYICI, FAZ KAYDIRICI VE VEKTÖR KİPLEYİCİLER

Ünlü, Mehmet Doktora, Elektrik ve Elektronik Mühendisliği Bölümü Tez Yöneticisi: Doç. Dr. Şimşek Demir Ortak Tez Yöneticisi: Prof. Dr. Tayfun Akın

Mart 2009, 268 sayfa

Bu tezde, RF MEMS teknolojisi kullanılarak gerçekleştirilen yeni ayarlanabilir empedans ayarlayıcı, faz kaydırıcı ve vektör kipleyicilerin teorisi, tasarımı, üretimi ve ölçüm sonuçları sunulmaktadır. Sunulan devreler üç saplamali bir topoloji kullanılarak tasarlanmış, ayrıca bu topoloji ile gelen işaretin genliğinin ve fazının aynı anda değiştirilebildiği bu tez kapsamında kuramsal ve deneysel olarak gösterilmiştir. Sunulan devrelerin tamamı, ODTÜ'de geliştirilen, altı maske ile kuvars taban üzerinde gerçekleştirilen ve ODTÜ RF MEMS Üretim Süreci olarak adlandırılan bir yüzey mikroişleme süreci ile üretilmiştir. Bahsi geçen RF MEMS empedans ayarlayıcı 6-20 GHz frekans bandında çalışmak üzere tasarlanmış ve Smith Abağı'nı 1331 empedans noktası ile tarayabilmektedir. Yapının ölçülen 729 empedans noktası göstermektedir ki, bu yapı ile Smith Abağı'nın çok büyük bir bölümü bahsi geçen bandın tamamında kapsanabilmektedir. RF MEMS faz kaydırıcı 15 GHz frekansında 0-360° faz aralığını 10°'lik adımlarla tarayabilecek şekilde tasarlanmıştır. Üretilmiş olan faz kaydırıcının ölçümü yapılan 21 durumundaki ortalama faz hatası 1.7°, araya girme kaybı -3.1 dB ve geriye dönüş kaybı -19.3 dB olarak elde edilmiştir. Ayrıca bu yapı; 30 GHz frekansına kadar ortalama -5 dB, 40 GHz frekansına kadar da ortalama -8 dB araya girme kaybı ile çalışabilmektedir. RF MEMS vektör kipleyici, 22.5-27.5 GHz frekans bandında, 3 adet genlik ve 8 adet faz durumu verebilecek şekilde tasarlanmıştır. Üretilmiş olan vektör kipleyicinin, 22.5, 25 ve 27.5 GHz frekanslarında yapılan 24'er durum ölçümünde ortalama, 0.5 dB genlik hatası, 4° faz hatası ve -15 dB geriye dönüş kaybı ile çalıştığı görülmüştür.

Bu tezde sunulan sonuçlar bir kez daha göstermektedir ki, ODTÜ RF MEMS Üretim Süreci olgundur ve karmaşık RF MEMS yapıların üretiminde başarıyla kullanılabilmektedir.

Anahtar Kelimeler: RF MEMS, empedans ayarlayıcı, faz kaydırıcı, vektör kipleyici, MEMS anahtar.

To My Family

### ACKNOWLEDGMENTS

First, I would like to express my sincere appreciation and thanks to both of my supervisors Prof. Şimşek Demir and Prof. Tayfun Akın. I would like to thank Prof. Şimşek Demir for following, attending, and supporting each step of my research wherever possible, and also caring for me more than I do for myself, not only in the academic sense, but also in all the remaining parts of life. I would like to thank Prof. Tayfun Akın for giving me the opportunity to join his group, providing this extraordinary research environment in my country, and introducing me the invaluable people in his group. He was so kind and patient to listen to me for hours whenever I need at the hard times of all parts of my life.

I would like to thank Prof. Sencer Koç for taking the time to solve the points that I had stuck and supporting for being on my committee. I also would like to thank Prof. Hayrettin Köymen for taking time to serve on my committee, and also for his innovative solutions. I am indebted to Prof. Altunkan Hızal for taking time to attend to all of my committee meetings even though he was not a member, providing the millimeterwave laboratory environment, and encouraging me during the whole thesis study. I would like to acknowledge Prof. Özlem Aydın Çivi for her help and support during this Ph. D. study.

Words fail me when I try to express my gratefulness to Dr. Kağan Topallı after 13 years. He gave me priceless support during this Ph. D. study, and a shoulder that I can hold on at all the hard times of my life. I would like to thank the members of the METU RF MEMS Group, especially Hüseyin Sağkol (TU Delft), Halil İbrahim Atasoy, and Çağrı Çetintepe for their support and friendship. I also would like to acknowledge İpek İstanbulluoğlu Turhan, Engin Ufuk Temoçin, Ömer Bayraktar, İlker

Comart, Ozan Doğan Gürbüz, Caner Güçlü, Ahmet Hayrattin Yüzer for their help and friendship.

There are many people who helped me along the way in numerous ways. I would like thank the former and present members of the METU MEMS and BIOMEMS Groups. I would specially like to thank Murat Tepegöz. He is much more than a friend for me, and I owe much to him. I also would like to thank Dr. Mustafa Seçmen, Dr. Mahmud Yusuf Tanrıkulu, and Dr. Said Emre Alper in all sincerity for their nice accompany during the past 8 years. I would like to acknowledge Dr. Özge Zorlu for his friendship and help.

This research would not have been possible without the support of a special person, Orhan Akar. I am deeply indebted to him for his help, never-ending support, and friendship. I also would like to thank all of the personnel both inside and outside the cleanroom at METU MEMS Center fabrication facilities for their support.

I would specially like to thank my parents, Mustafa Ünlü and Kübra Ünlü, for their everlasting love, encouragement, and support throughout my whole life. I would not have been here without them, and I understand their priceless value more on every passing day of my life. I also would like to thank my grandfather Kenan Kul, who is my first mentor, my uncle Emre Kul, and all the elder and younger members of my family, who supported me with their prayers.

Last but not the least, I would like to express my deepest gratitude for my wife Nuriye for her everlasting love, patience, encouragement, and support. This achievement is as much hers as it is mine. I also would like to thank my son Ahmet Selim for being joy of my family.

Finally, I would like to thank TÜBİTAK BİDEB for the scholarship during my Ph. D. study.

# **TABLE OF CONTENTS**

ABSTRACTiv
ÖZvi
ACKNOWLEDGMENTSix
TABLE OF CONTENTSxi
LIST OF TABLESxv
LIST OF FIGURES xviii
CHAPTERS
1. INTRODUCTION1
1.1. RF MEMS Impedance Tuners and Matching Networks7
1.2. RF MEMS Phase Shifters10
1.3. Vector Modulators13
1.4. Research Objectives and Organization of the Thesis
2. METU RF MEMS FABRICATION PROCESS AND ITS IMPROVEMENT 20
2.1. Introduction20
2.2. METU RF MEMS Fabrication Process21
2.2.1. Process Layers and Material Selection
2.2.2. Process Flow
2.3. The Optimization of METU RF MEMS Fabrication Process
2.3.1. Optimization of the Conductivity of the Ti/Au First Metal Layer 33
2.3.2. Optimization of the Stress of the Au Structural Layer
2.4. Conclusion

	AR	RECON	VFIGURABLE RF MEMS TRIPLE STUB IMPEDANCE TUNER FOR	
	6-20	) GHZ	APPLICATIONS	. 52
3	3.1.	Intro	duction	. 52
3	3.2.	RF M	IEMS Impedance Tuner Design	53
	3.2	2.1.	RF MEMS Switch (Unit Section) Design	54
	3.2	2.2.	T-Junction Design	60
	3.2	2.3.	The Simulations of the Triple Stub Impedance Tuner	.61
	3.2	2.4.	Case Studies	63
3	3.3.	The l	-abrication and the Measurement Results	66
	3.3	8.1.	The Packaging of the Impedance Tuner	66
	3.3	8.2.	The Design and the Implementation of the Multi-Pin ME	MS
			Automated Measurement Setup (MAMS)	.70
	3.3	8.3.	The Measurement Results of the RF MEMS Impedance Tuner	. 77
Э	3.4.	Conc	lusion	იე
4				. 95
	THE	THE	ORY OF PHASE AND AMPLITUDE CONTROL USING TRIPLE STUB	. 55
-	THE TOP		ORY OF PHASE AND AMPLITUDE CONTROL USING TRIPLE STUB	. 93
4	ТНЕ ТОР 4.1.	THE OLOC Intro	ORY OF PHASE AND AMPLITUDE CONTROL USING TRIPLE STUB	. 93 . 94 . 94
4	THE TOP 4.1. 4.2.	THE POLOC Intro The T	DRY OF PHASE AND AMPLITUDE CONTROL USING TRIPLE STUB GY duction Fheory of Phase Control Using Triple Stub Topology	. 93 . 94 . 94 . 95
4	THE TOP 4.1. 4.2. 4.2	THE POLOC Intro The T 2.1.	ORY OF PHASE AND AMPLITUDE CONTROL USING TRIPLE STUB GY duction Fheory of Phase Control Using Triple Stub Topology Lossless Case	. 94 . 94 . 95 . 95
4	THE TOP 1.1. 1.2. 4.2 4.2	THEO POLOC Intro The T 2.1. 2.2.	DRY OF PHASE AND AMPLITUDE CONTROL USING TRIPLE STUB GY duction Fheory of Phase Control Using Triple Stub Topology Lossless Case Lossy Case: Example Case Study: $t = \lambda/8$	. 94 . 94 . 95 . 95
4	THE TOP 4.1. 4.2 4.2 4.2 4.3.	THEO POLOC Intro The T 2.1. 2.2. The F	DRY OF PHASE AND AMPLITUDE CONTROL USING TRIPLE STUB GY duction Theory of Phase Control Using Triple Stub Topology Lossless Case Lossy Case: Example Case Study: $t = \lambda/8$ Phase / Amplitude Control Using the Triple Stub Topology	. 93 . 94 . 94 . 95 . 95 . 98
 	THE TOP 4.1. 4.2 4.2 4.3 4.3	THE POLOC Intro The T 2.1. 2.2. The P 3.1.	ORY OF PHASE AND AMPLITUDE CONTROL USING TRIPLE STUB GY duction Theory of Phase Control Using Triple Stub Topology Lossless Case Lossy Case: Example Case Study: $t = \lambda/8$ Phase / Amplitude Control Using the Triple Stub Topology	. 93 . 94 . 94 . 95 . 95 . 98 102
4	THE TOP 4.1. 4.2 4.2 4.3 4.3 4.3	THE POLOC Intro The 7 2.1. 2.2. The 8 3.1. 3.2.	DRY OF PHASE AND AMPLITUDE CONTROL USING TRIPLE STUB GY	. 93 . 94 . 95 . 95 . 98 102 106 113

	4.4.1.	The Design of the Single Metal Layer Triple Stub Topole	ogy Circuits
			116
	4.4.2.	The Measurements of the Single Metal Layer Triple Stu	b Topology
		Circuits	119
	4.5. Othe	er Possible Application Areas	127
	4.5.1.	IQ Divider	128
	4.5.2.	1: <i>k</i> Power Divider	129
	4.5.3.	Vector Modulator Type I	130
	4.5.4.	Vector Modulator Type II	131
	4.6. Cond	clusion	
5.	RF MEMS	SPHASE SHIFTER USING TRIPLE STUB TOPOLOGY	134
	5.1. Intro	oduction	134
	5.2. The	Design of the RF MEMS Phase Shifter	135
	5.2.1.	The T-Junction Design	136
	5.2.2.	The DMTL Interconnection Line Design	142
	5.2.3.	The DMTL Stub Design	164
	5.2.4.	The Overall RF MEMS Phase Shifter Design	170
	5.3. The	Fabrication and the Measurement Results	175
	5.3.1.	The DMTL Parameter Extraction	
	5.3.2.	The RF MEMS Phase Shifter Measurements	
	5.4. Cond	clusion	
6.	RF MEMS	SVECTOR MODULATOR USING TRIPLE STUB TOPOLOGY	202
	6.1. Intro	oduction	
	6.2. The	Design of the RF MEMS Vector Modulator	203
	6.2.1.	The DMTL stub Design	

6.2.2.	The Overall RF MEMS Vector Modulator Design	210
6.3. The	Fabrication and the Measurement Results	217
6.3.1.	The DMTL Parameter Extraction	218
6.3.2.	The RF MEMS Vector Modulator Measurements	224
6.4. Con	clusion	236
7. CONCLUS	SIONS AND FUTURE WORK	237
8. REFEREN	CES	243
9. DETAILED	D FABRICATION PROCESS FLOW	259
REFERENCES		243
APPENDIX		259
CURRICULUM	VITAE	263

# **LIST OF TABLES**

### TABLES

Table 1.1: The application areas of RF MEMS switches, varactors, and inductors [7].5
Table 2.1: The materials used in the METU RF MEMS fabrication process22
Table 2.2: The measured physical properties of the METU RF MEMS fabricationprocess materials.24
Table 2.3: The summary of the sputtering process parameters that are used for theelectrical conductivity optimization of the first metal layer
Table 2.4: The extracted material properties and losses for the optimized and old Ausputtering processes on quartz and glass substrates.36
Table 2.5: The measured material properties for the Au electroplating optimizationtrial processes.37
Table 2.6: The process conditions and the measured stress levels of the recipes that are measured on blank wafers for the structural layer stress optimization
Table 2.7: The process conditions and the measured buckling levels of the recipesthat are measured on released MEMS bridge for the structural layer stressoptimization.41
Table 3.1: The physical design parameters of the RF MEMS switch
Table 3.2: The circuit design parameters of the RF MEMS switch design
Table 3.3: The physical design parameters of the T-junction design. 61

Table 3.5: The summary one-to-one correspondence between the measured andthe simulated impedance point of the IT.87

Table 4.2: The lengths of the parts of the circuit (3) for different values oftransmission line attenuation constants.112

Table 5.2: The circuit design parameters of the DMTL UST-I and DMTL UST-II. ..... 151

Table 5.7: The final simulation results of the RF MEMS PS design for 0-360° range
with 10° steps
Table 5.8: The circuit model parameters of the DMTL UST-I, II, and III that are
extracted from the measurements181
Table 5.9: The summary of measured performance of the RF MEMS phase shifter.
Table 6.1: The circuit design parameters of the DMTL UST-IV. 206
Table 6.2: The physical design parameters of the final design of the DMTL UST-IV.
Table 6.3: The final simulation results of the RF MEMS VM design for 0-360° range
with 45° steps and -2 to -8 dB amplitude range with 3 dB steps
Table 6.4: The circuit model parameters of the DMTL UST-IV that are extracted from
the measurements
Table 6.5: The summary of measured performance of the RF MEMS VM232

# **LIST OF FIGURES**

FIGURES
Figure 1.1: The MEMS market forecast for the 2007-2012 period [2]2
Figure 1.2: The MEMS market forecast for the 2007-2012 period [6]
Figure 1.3: The block diagrams of the vector modulator topologies
Figure 2.1: (a) The flow of the METU RF MEMS fabrication process
Figure 2.2: The comparison of the simulated and the measured impedance coverage
of an RF MEMS impedance tuner for different unloaded line loss values
Figure 2.3: The comparison of the measured and simulated return loss performance
of a tunable RF MEMS antenna [49]31
Figure 2.4: Examples of the RF MEMS structures from previous runs that suffer from
buckling due to residual stress (a) Compressive stress level (b) Tensile stress level.32
Figure 2.5: The measurement results of the CPW test structures that are fabricated
on glass and quartz substrates with optimized and old first metal layer recipes (a)
S <sub>11</sub>   (b)  S <sub>21</sub>
Figure 2.6: The snapshots of the MEMS beams simulated using Coventorware 45
Figure 2.7: The optical profiler measurements and SEM pictures of the MEMS test
beams
Figure 2.8: The optical profiler measurements and SEM pictures of other MEMS
structures 50

Figure 3.1: The block diagram of the triple stub impedance tuner
Figure 3.2: (a) 3D view of the RF MEMS switch of the triple stub IT (z-dimension
exaggerated) (b) Top view of the RF MEMS switch of the triple stub IT55
Figure 3.3: The circuit model of the MEMS switch that is used for the design of the
triple stub IT
Figure 3.4: The comparison of HFSS simulations with that of the circuit model of the
RF MEMS switch that is used for the design of the triple stub IT59
Figure 3.5: The layout of the T-junction that is used in the design of the triple stub IT
Figure 3.6: The Smith Chart coverage of the designed IT in 6-20 GHz band for 1331 impedance points
Figure 3.7: The performance of the triple stub IT for 50 $\Omega$ -to-250 $\Omega$ impedance
transformation at 10 GHz (a) $ S_{11} $ (b) $ S_{21} $ (c) The loss of the tuner at this state 64
Figure 3.8: The performance of the triple stub IT for 50 $\Omega$ -to-20 $\Omega$ impedance
transformation at 20 GHz (a) $ S_{11} $ (b) $ S_{21} $ (c) The loss of the tuner at this state 65
Figure 3.9: The photographs of the fabricated IT67
Figure 3.10: The SEM pictures of the fabricated RF MEMS IT
Figure 3.11: The photographs of the packaged RF MEMS IT70
Figure 3.12: The bonding diagram of the RF MEMS IT for HYB02413 package70
Figure 3.13: The block diagram of the multi-pin MEMS automated measurement setup
Figure 3.14: Some of the possible waveforms that can be produced by the
"Waveform Generator Card"73

Figure 3.15: The photograph of the waveform generator card and its block diagram. 74

Figure 3.19: The measured upstate performance of the RF MEMS IT compared with the simulations (Stb1 = 0, Stb2 = 0, Stb3 =0) (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c)  $\angle S_{21}$  (d)  $\angle S_{11}$ .... 80

Figure 3.20: The measured performance of the RF MEMS IT compared with the simulations (Stb1 = 4, Stb2 = 2, Stb3 = 32) (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c)  $\angle S_{21}$  (d)  $\angle S_{11}$ .......82

Figure 3.21: The measured performance of the RF MEMS IT compared with the simulations (Stb1 = 64, Stb2 = 384, Stb3 = 24) (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c)  $\angle S_{21}$  (d)  $\angle S_{11}...83$ 

Figure 3.23: The 10 dB impedance coverage of the measured impedance points that can be matched by the fabricated RF MEMS IT (a) 10 GHz (b) 20 GHz. The impedance points in this figure are the conjugates of the ones given in Figure 3.22.

Figure 3.25: The measured performance of the triple stub IT for 50 $\Omega$ -to-150	Ω
impedance transformation at 20 GHz compared with the simulations (a) $ S_{11} $	(b)
S <sub>21</sub>   (c) The loss of the tuner at this state	92

Figure 4.2: The insertion phase, the insertion loss, and the return loss performances of the TST based phase shifter for different transmission line losses (a)  $S_{21}$  (b)  $S_{11}$ ..99

Figure 4.3: The performance of the TST based phase shifter for different interconnection line lengths (a)  $t = 0.0375\lambda$ ,  $t = \lambda/8$ ,  $t = \lambda/4$  (b)  $t = 3\lambda/8$ ,  $t = 0.4625\lambda$ .

Figure 4.5: The performance of the TST when it is set to three sample insertion phase/amplitude states at 15 GHz (a)  $|S_{21}|$  (b)  $\angle S_{21}$  (c)  $|S_{11}|$  (d) The loss of the TST. 106

Figure 4.7: The schematic of  $\lambda/2$  resonator with 0, 1, and 2 input/output ports... 109

Figure 4.9: The  $|S_{21}|$  and  $|S_{11}|$  of the resonator circuit (3)......112

Figure 4.14: The measurement results of the TST based single metal layer PS for (-1 dB, 180°) state compared with the simulations (a)  $|S_{21}|$  (b)  $\angle S_{21}$  (c)  $|S_{11}|$ . .... 123

Figure 4.15: The measurement results of the TST based single metal layer VM for (-6 dB, 90°) state compared with the simulations (a)  $|S_{21}|$  (b)  $\angle S_{21}$  (c)  $|S_{11}|$ . ..... 124

Figure 4.20: The block diagram of the proposed "Vector Modulator Topology I	" that
uses TSTs.	131

Figure 4.21: The block diagram of the proposed "Vector Modulator Topology II" that
uses TSTs
Figure 5.1: The block diagram of the RF MEMS phase shifter135
Figure 5.2: The layout of the designed T-junction. The dimensions are given in $\mu$ m.
Figure 5.3: The electrical model of the T-junction discontinuity138
Figure 5.4: The PS models that are used to compare the effects of the T-junction: (a)
The PS without T-junction (b) The PS with T-junction139
Figure 5.5: The comparison of the S-parameters of the PS for with and without T-
junction cases: (a) $ S_{11} $ (b) $ S_{21} $ (c) The error between the S-parameter of the two
cases
Figure 5.6: The circuit model of a unit section of a digital type DMTL144
Figure 5.7: The $S_{21}$ of the ideal TST based PS with different interconnection lines
lengths. The figures show two insertion phase values when the insertion loss is
minimum and when the insertion loss is equal to -3 dB149
Figure 5.8: The circuit model of the DMTL unit section that is used for the design of
the DMTL interconnection line
Figure 5.9: (a) 3D view of the DMTL UST-I (z-dimension exaggerated) (b) Top view of
the DMTL UST-I (c) 3D view of the DMTL UST-II (z-dimension exaggerated)154
Figure 5.10: The circuit model of the DMTL unit section that is used for the design of
the DMTL interconnection line
Figure 5.11: The comparison of the S-parameters of the circuit model and HFSS
simulations for the DMTL UST-I and DMTL UST-II. The graphs are the results of
cascaded 16 unit sections both for the circuit model and the HFSS simulation results

(a)-(c) DMTL UST-I (d)-(f) DMTL UST-II (g) The error between the S-parameters of the circuit model and the HFSS simulations of both unit sections for all cases. ..... 159

Figure 5.13: The layout of the DMTL interconnection line......162

Figure 5.15: 3D view of the DMTL UST-III (z-dimension exaggerated)......166

Figure 5.17: The S11 characteristics of the DMTL stub design for all possible 512states.169Figure 5.18: The layout of the final RF MEMS PS design.170Figure 5.19: The S21 performance of the designed RF MEMS PS for all 36 phase172Figure 5.20: The performance of the designed RF MEMS PS for all 36 phase states172Figure 5.20: The performance of the designed RF MEMS PS for all 36 phase states174Figure 5.21: The photographs of the fabricated RF MEMS PS.176

Figure 5.32: The measured  $S_{21}$  performance of the RF MEMS PS (SOLT calibration used) (a) 22.5 GHz (b) 25 GHz (c) 27.5 GHz (d) 30 GHz (e) 35 GHz (f) 40 GHz...... 198

Figure 5.33: The simulated  $S_{11}$  characteristics of the DMTL stub design for all possible 512 states with the extracted circuit model parameters at some measurement frequencies (a) 15 GHz (b) 25 GHz (c) 30 GHz (d) 40 GHz...... 200

Figure 6.9: The photograph of the packaged RF MEMS VM......221

Figure 6.10: The comparison of the measured S-parameters and the circuit model simulations for the 9 cascaded UST-IVs. (a) Magnitudes of the S-parameters in the upstate (b) Magnitudes of the S-parameters in the downstate (c) The phases of the

Figure 6.12: The measured upstate performance of the RF MEMS VM (SOLT calibration used) compared with the simulations (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c)  $\angle S_{21}$ ...... 226

Figure 6.14: The measured performance of the RF MEMS VM (SOLT calibration used) that is reconfigured to (-5 dB, -135°) @ 22.5 GHz compared with the simulations (Stb1 = 32, Stb2 = 1, Stb3 = 0, IntLine = 0) (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c)  $\angle S_{21}$ . 229

### CHAPTER 1

### **INTRODUCTION**

The way of making things smaller always attracted scientists' attention, especially after a high number of technological inventions were achieved in the twentieth century. Dr. Richard Feynman's famous talk, "There's Plenty of Room at the Bottom" [1] was maybe one of the best foresights about what the technology will become in today's world. Micro-Electro-Mechanical systems, namely MEMS, are one of exciting and emerging technologies that Dr. Feynman had foreseen its potential. MEMS technology is the way of doing things in the "micro" world that are used to be done in the "macro" world. The advantage brought is more than just reducing the size; MEMS are employed as miniature embedded systems that also use the advantages brought by the microelectronics. They enable higher level functions, they integrate a number of functions in one single package, and they decrease the costs using the benefits of the batch fabrication technology.

MEMS have been developed starting from 1970s, and today, they are becoming one of the mature fields of electronics with a high number of manufacturers and research institutes, and the demand on MEMS is increasing day by day. The MEMS market value is reported to be 7.1 billion USD in 2007, and it is expected to be 14 billion USD by 2012 [2]. The MEMS products include inkjet printer heads, pressure sensors, accelerometers, gyroscopes, microdisplays, microbolometers, Si microphones, MOEMS, microfluidics, micro probes, micro fuel cells, and RF MEMS. Figure 1.1 shows the MEMS market forecast for the 2007-2012 period [2].



#### MEMS market forecast 2007 - 2012 in value (USD million)

Figure 1.1: The MEMS market forecast for the 2007-2012 period [2].

One of the highly emerging fields of MEMS is the MEMS for radio frequency applications, shortly RF MEMS. Although the research on RF MEMS started much later than the other fields of MEMS, they gained a huge popularity. The first RF MEMS structure was demonstrated by Larson et al. in 1991 [3], which was a mechanical switch specifically designed for microwave applications. This work was followed by two outstanding switches in 1995, which were the metal-to-metal contact type Rockwell switch [4] and the capacitive type Texas Instruments switch [5] suitable for DC-60 GHz and 10-120 GHz applications, respectively. These works gave rise to the RF MEMS research all around the world with several universities and companies, which include the giant electronics manufacturers such as Motorola, Analog Devices, Samsung, Omron, NEC, and ST-Microelectronics. Today, RF MEMS technology is on the way to become one of the mature fields of microwave electronics, and the RF MEMS market is growing gradually day-by-day. Figure 1.2 presents the market forecast for the RF MEMS switches for the 2006-2011 period [6].



#### **RF MEMS** switch market forecast

Figure 1.2: The MEMS market forecast for the 2007-2012 period [6].

RF MEMS offer important advantages in the switching devices technology with respect to their semiconductor rivals [7]. The most important one is the cut-off frequency, which is about 40 THz, and this value is 20 times of its closest rival by 2001. The other important advantages brought by RF MEMS technology is high linearity, low insertion loss, high isolation, and low power consumption.

A number of RF MEMS components and systems have been developed all around the world using RF MEMS technology. These are mainly concentrated on the switches ([7]-[15]), the varactors ([7], [16], [17]) and monolithic and hybrid components and systems that use RF MEMS switches such as phase shifters ([7], [18]-[30]), reconfigurable impedance matching networks ([31]-[37]), reconfigurable filters ([7], [38]-[42]), and reconfigurable antennas ([43]-[52]). There are also components other than the switches and switch based structures that are outcomes of this technology. Research on static micromachined components such as inductors, transmission lines, high-Q resonators, filters, and antennas; thin film bulk acoustic resonators (FBAR) and filters; and micromechanical resonators and filters are carried out by universities and companies all around the world [53]-[55]. Among all of the RF MEMS components, FBARs are the first commercial ones [56].

The advantages brought by RF MEMS technology make these components strong candidates for a high number of applications in the microwave world, which are summarized in Table 1.1. Although the performances demonstrated by RF MEMS components are promising for these applications, these components should be ready for batch fabrication, which means that the problems of RF MEMS components should be solved. The main problems of RF MEMS components are high actuation voltage requirements, relatively low speed, reliability, power handling, packaging, and cost. The high actuation voltage requirement is solved by means of voltage upconverter circuits with the price of additional cost. There are already a high number of application areas for the RF MEMS switches with their current switching speeds, which are about 1-100 µs. To solve the remaining problems, the research on RF MEMS was directed to these issues all around the world. Today, it is reported that metal-to-metal contact switch of RadantMEMS [14], which is commercially available by now, worked for 900 billion cycles at +20 dBm RF power (cold-switched) without any failure [57]. This switch was hermetically packaged using glass frit, which means that the packaging problem was also solved. Another switch from RadantMEMS was demonstrated to work for 13 billion cycles at 40 dBm (again cold-switched), which was fabricated using a modified process for higher power handling operation. In [58], it is reported that a capacitive, shunt RF MEMS switch worked for more than 100 billion cycles +20 dBm RF power (hot-switched) without any failure. This switch was also packaged using wafer-level micro-encapsulation [59]. The cost is not a major issue for defense and aerospace applications; however, the cost including the packaging is still a problem for the cost-driven commercial applications. These developments show that RF MEMS is on the way for becoming a mature technology, and more companies will be providing RF MEMS components from batch fabrication soon [60].

Area	System	Number of Cycles (Billions)	Years
Phased arrays	Communication systems (ground)	1-10	2-10
	(space)	10-100	2 - 10
	(airborne)	10-100	2-10
Phased arrays	Radar systems (ground)	10-100	5-10
	(space)	10-100	5 - 10
	(missile)	0.2 - 10	1-5
	(airborne)	1 - 100	5 - 10
	(automotive)	1 - 10	5-10
Switching and recon- figurable networks	Wireless communication (portable)	0.01-4	2–3
	(base station)	0.1 - 100	5 - 10
	Satellite (communication and radar)	0.1 - 1	2 - 10
	Airborne (communication and radar)	0.1 - 10	2 - 10
	Instrumentation	10 - 100	10
Low-power oscilla-	Wireless communication (portable)	0.1	2–3
tors and amplifiers	Satellite (communication and radar)	0.1-1	2 - 10
(varactors, inductors)	Airborne (communication and radar)	0.1–10	2–10
RF MEMS Elements:	Switch, Varactor, Inductor		
Subsystems:	Switching networks <sup><i>a</i></sup>		
	Transmit/receive switches		
	Very high isolation switches (instrumentation)		
	Programmable attenuators		
	Phase shifters (digital and analog)		
	Reconfigurable antennas		
	Reconfigurable matching (or impedance) networks		
	Reconfigurable Butler matrices for multibeam systems		
	Tunable filters		
	Switched filter banks		
	Miniature microwave filters		
	Switched diversity antennas, oscillators, amplifiers		
	Low phase-noise oscillators (fixed and tunable)		
	High-efficiency networks (low-power systems)		

Table 1.1: The application areas of RF MEMS switches, varactors, and inductors [7].

<sup>*a*</sup>SPST, SPDT, DPDT, SPNT,  $N \times N$ , absorptive and reflective designs

The opportunities brought by RF MEMS technology today not only increase the component and the system performances, but also offer new aspects to build some new structures that cannot be implemented due to the performance limitations of

the components of the current state of the art semiconductor technology. For example, the idea of cascading many switching elements is not feasible since the insertion loss will increase rapidly and become intolerable. However, RF MEMS is an "enabling" technology, and it is possible to design totally reconfigurable structures and even control the whole physical layout of a distributed structure by means of RF MEMS switches [61], thanks to their very low insertion losses and high isolation. Even though this is limited to some special cases, the idea of changing the electrical length of a transmission line with small steps is achieved easily, which, in fact, is sufficient to implement structures with quasi-continuous control of performance using digital RF MEMS switches.

This thesis presents the theory, design, and implementation of reconfigurable circuits that are based on triple stub topology. The idea is to change the electrical lengths of the stubs precisely by means of RF MEMS switches or RF MEMS bi-state capacitors so that the performance of the circuit is controlled in a quasi-continuous manner. The triple stub topology is mostly used as an impedance matching network in the literature. However, it is theoretically and experimentally shown in this thesis that it is possible to control the phase and amplitude of the input signal simultaneously while the input return loss is exactly zero using the triple stub topology. In this thesis, the RF MEMS implementation of a novel impedance tuner, a novel phase shifter, and a novel vector modulator are presented using the triple stub topology.

The following parts of this chapter give the previous works on RF MEMS impedance tuners, RF MEMS phase shifters, and vector modulators. The chapter concludes with the organization of the thesis and the research objectives.

#### 1.1. RF MEMS Impedance Tuners and Matching Networks

One of the applications that employ the advantages brought by RF MEMS technology is the impedance tuners or impedance matching networks. A low-loss, reconfigurable impedance tuner or impedance matching network with wide impedance coverage can be very useful for tomorrow's reconfigurable and intelligent systems. The possible applications of the RF MEMS impedance matching networks and impedance tuners include the impedance matching between a power amplifier and a radiating element; and noise or load-pull measurements, respectively.

Impedance tuners that are used in the noise or load-pull measurements are mostly mechanical devices with either coaxial or waveguide structures, and they use motors for reconfigurability. RF diodes and transistors are also used for impedance tuners; however, these designs introduce high insertion losses, and/or the impedance coverage is limited [62]-[66].

The first study on RF MEMS impedance matching networks was reported by Chiao et al. [67]. The authors use the idea of single stub matching, and the length of the transmission lines are adjusted by a sliding backshort plate on top of a transmission line. The position of the sliding short is changed by MEMS voltage controlled actuators.

The second RF MEMS impedance tuner was reported by Kim et al [68]. In this work, the authors presented two types of RF MEMS impedance tuners. The first one was an analog impedance tuner, which uses two shunt resonant cells connected in parallel to a transmission line. The resonant cells are operated near resonance frequency which allows the authors to obtain a large impedance variation. With this design, they obtained a voltage standing-wave ratio (VSWR) of 21.2 in Ka-band. The second design was a digital RF-MEMS impedance tuner which employs RF

MEMS switches to change the lengths of the two stubs of a double-stub network. The authors report a maximum VSWR of 32.3 with this second design.

The idea of using the triple stub topology on RF MEMS impedance matching networks is presented in [69] by Unlu et al.; however, the fabrication process was not mature at that time. So, measurement results were not presented.

A hybrid implementation of a reconfigurable double-stub impedance tuner was presented by Papapolymerou et al. [70]. The stubs were reconfigured using a digital capacitor bank in which the capacitors were selected using RF MEMS switches in order to obtain different load impedances. The microstrip parts of the design were implemented using standard lithography techniques on alumina substrate whereas the MEMS parts were implemented using surface micromachining on high resistivity Si substrate. The authors report that the structure can match loads with real parts between 1.5  $\Omega$  and 109  $\Omega$  and imaginary parts between -107  $\Omega$  and 48  $\Omega$  at 20 GHz.

A monolithic design on microstrip lines using ohmic contact MEMS switches was presented in [71]. The design has a double stub structure where the end of each stub is connected to a radial stub via MEMS switches forming a "3-bit, 3-bit" network. The aim of the design was efficiency optimization for a Class-E amplifier in X-band. The reported impedance range at 10 GHz is 0.63  $\Omega$  to 57  $\Omega$  in real part and -7.5  $\Omega$  to 51.8  $\Omega$  in the imaginary part.

Vähä-Heikkilä and Rebeiz presented the most mature reconfigurable impedance tuners, which employ the DMTLs. A 4-18 GHz design was implemented using a DMTL directly as an impedance tuner [72]. The authors also presented single stub topology for 20-50 GHz band [74], and double and triple stub [77] topologies to obtain wider impedance coverage. In all of these studies, the stubs and the interconnection lines are implemented using a few DMTL unit sections.

Qiao et al. [73] presented an intelligently controlled RF power amplifier with a reconfigurable RF MEMS varactor tuner for output matching. A double stub tuner is used for the output impedance tuner which employs MEMS switches and varactors. The gain and efficiency of the power amplifier is dynamically optimized by controlling the MEMS switches and the MEMS varactor control voltage for some known and unknown frequencies in 8-12 GHz band.

Another double stub design using shunt capacitive RF MEMS switches was demonstrated by Lu et al. [75]. Here, RF MEMS switches were used to change capacitance, and hence, the electrical length of the stubs. The two matching networks were used as input and output matching networks of a class AB amplifier to optimize gain and efficiency.

A tunable "double-slug" impedance matching network was demonstrated by Shen and Barker [76] which employs distributed MEMS transmission lines (DMTL). The design is composed of two 90° DMTL lines, which determine the operation frequency, and two other DMTL sections with electrical lengths  $\theta_1$  and  $\theta_2$  which determine the load value to be matched. Each switch of the DMTL structure is actuated separately resulting with adjustable  $\theta_1$  and  $\theta_2$ .

Finally, an RF MEMS triple stub impedance matching network was presented by Unlu et al. [78], where the electrical lengths of the stubs were precisely reconfigured using a high number of RF MEMS switches.

In all of the above mentioned stub based studies except [69] and [78], only a small number of RF MEMS switches are employed on each stub. As a result, the electrical lengths of the stubs are digitally controlled with large steps, and the whole Smith Chart coverage could not be achieved because of the electrical length resolution of the stubs. Moreover, only a limited number of measured impedance points are presented.
The RF MEMS triple stub impedance tuner presented in this thesis employs 10 RF MEMS switches per stub, which is equivalently changing the length of the stub by  $\lambda/20$  steps. This results with the coverage of the almost whole Smith Chart at the center frequency. The impedance coverage is better than all of the works in the literature, and a wide coverage is also obtained for the remaining frequencies of the design band. 729 out of 1331 measured impedance points are presented, which is limited by the packaging. This number is the highest presented number in the literature up to now.

# **1.2. RF MEMS Phase Shifters**

Phase shifters are important crucial components for a number of applications in microwave and millimeterwave electronics. The phase shifters can be categorized in terms of design requirements, which are constant and linear phase versus frequency. The applications that require constant phase design are radar applications, communication systems, and components (SSB mixers, vector modulators, balanced amplifiers, etc.), and high precision instrumentation systems. The linear phase designs are mostly employed in phased array applications as true-time-delay networks.

The phase shifters are basically designed in two types, which are analog and digital controlled versions. The analog phase shifters, as the name refers, are used to control the insertion phase within 0-360° by means of varactors. The digital phase shifters are used to produce discrete phase delays, which are selected by means of switches.

There are three main technologies for the implementation of phase shifters, which are ferrite phase shifters, FET based phase shifters, and PIN based shifters. Ferrite phase shifters have low insertion loss, good phase accuracy, and they can handle high power. However, they are bulky, they require a large amount of DC power, and they are slow compared to their rivals [79]-[81]. FET based [82]-[84], PIN based [85]-[87], and varactor diode based [88] phase shifters are the semiconductor alternatives for phase shifter. They propose low cost, low weight, and planar solutions to phased array systems. PIN based phase shifters provide lower loss compared to the FET based ones; however, they consume more DC power.

The phase shifters are implemented in several different topologies [7]. These include reflection-type [85], switched-line [84], loaded-line [86], varactor/switched-capacitor bank [88], and switched network [82] topologies. In all of these digital topologies (except varactor based one), the switching components are FETs or PIN diodes. Since the insertion losses of these components are high, the overall insertion losses of the phase shifters are also high. The reported insertion losses are about 4-6 dB at 12-18 GHz and 7-10 dB at 30-100 GHz [7].

The switches provided by emerging RF MEMS technology offer very low insertion losses, almost zero DC consumption, and very high linearity compared to these semiconductor components. As a result, RF MEMS phase shifters become strong alternatives for semiconductor based phase shifters, provided that the application area is limited to relatively low scanning arrays. A number of phase shifters are demonstrated that employ RF MEMS switches [7]. These include reflect-line [18], switched-line [19], [21], and 1:N switched-line [20] topologies. The reported average insertion losses of these designs vary between -1 and -2.2 dB, which are much lower than that of the semiconductor based designs.

One other alternative topology for the implementation of phase shifters is the distributed structures, which are used to obtain very wide-band circuits. The idea of distributed structures is to load a transmission line periodically with active or passive elements so that the parasitics brought by these elements are included as a

part of the transmission line model. Distributed amplifiers, oscillators, mixers, multipliers, and pulse shaping circuits were presented using this technique. Distributed phase shifters that employ Schottky-diode varactors were presented in [89] and [90], which are analog phase shifters and show good performance up to 20 GHz. Distributed phase shifters that employ RF MEMS varactors have also been presented in [22] and [23] for applications up to 110 GHz. The distributed RF MEMS phase shifters, or using the authors' labeling distributed MEMS transmission lines (DMTL), offer lower insertion loss than the semiconductor varactor based phase shifters. However, the tuning range of the standard MEMS varactors are about 1.3 practically, so the phase range of the analog DMTL phase shifters are lower than that of the semiconductor varactor based ones. It is possible to obtain higher capacitance ratio using a modified DMTL topology, in which the RF MEMS switch is used to select or unselect a static capacitance that is much larger than the upstate capacitance of the RF MEMS switch. Using this topology, a digital phase shifter design is attained with higher capacitance ratio and more reliable operation. Examples of the phase shifters using this topology are presented in [24]-[28], and the reported insertion loss is about at most -2.5 dB up to 60 GHz [7].

The RF MEMS phase shifter presented in this thesis uses on a novel topology, which is based on a triple stub network. The electrical lengths of the stubs and the interconnection lines are adjusted by means of DMTLs. The presented phase shifter is a linear phase design; however, the center frequency can be adjusted between 15 GHz and 40 GHz. This phase shifter design requires a high number of cascaded DMTL unit sections, which makes it possible to implement only using the "enabling" RF MEMS technology. To the author's knowledge, this phase shifter topology and its RF MEMS implementation are the first demonstrations in the literature.

#### **1.3. Vector Modulators**

The concept of the direct (or homodyne) modulation of the carrier signal has been found to be attractive in the last two decades, which is removing the IF stage from a conventional heterodyne transmitter [91]-[94]. The removal of the IF stage means to eliminate the set of IF modulators, filters, and amplifiers that are unavoidable once a conventional mixer is employed. The resulting system has much lower complexity and cost, which makes the direct modulation concept very attractive for the low-cost transmitters. The application areas include phased arrays, electronic warfare, automotive radars, measurement systems, and digital communications, which is maybe one of the most rapidly growing fields.

Direct modulators or in other words vector modulators play an important role in the implementation of the direct modulation systems. The vector modulators are used in the realization of amplitude-shift keying (ASK), binary phase-shift keying (BPSK), and more likely in quaternary (or quadrature) phase-shift keying (QPSK) and quadrature amplitude modulation (QAM).

The vector modulators are generally designed in two types, which are the cascaded (or  $\alpha$ - $\phi$ ) modulator and the I-Q modulator. The  $\alpha$ - $\phi$  modulator consists of a cascade connection of an attenuator and a phase shifter, which can be seen in Figure 1.3-(a). The I-Q modulator divides the input power into two orthogonal vectors so that any vector can be obtained by applying phase and amplitude control on these vectors and finally by combining them. The block diagram of the I-Q modulator is given in Figure 1.3-(b). This topology is more popular compared to the former one, and it is implemented with many variations. The inphase combiner is common in most of the variations, but the divider at the input can be replaced by couplers, hybrids, splitters, etc. The more critical part is the phase and amplitude control part on each arm. Reflective type phase shifters with variable terminations, mixers, and high

pass/low pass structures are some of the alternatives that are used for the phase and amplitude control in vector modulators.



Figure 1.3: The block diagrams of the vector modulator topologies.

The research on vector modulators were started in 1980s, and mature vector modulators started to show up at the end of this decade. The  $\alpha$ - $\phi$  vector modulators were first presented by Norris et al. [95], and these works were followed by other researches [96]-[98]. Devlin et al. [99] presented the first I-Q type vector modulator. This design employs quadrature power splitters with reflective FET terminations as variable resistances, which is actually a modified bi-phase modulator. An improved approach for reflective terminations was followed by Lo et al. [100], in which switching low noise amplifiers (LNAs) are implemented for BPSK. In this work, balanced reflective terminations are naturally

cancelled since two reflective terminations were used in a balanced manner. Ashtiani et al. [101] presented once of the fundamental publications of this topology following the above mentioned work. Ashtiani and his co-authors presented the successful usage of balanced reflective terminations on multiple-QAM (M-QAM) applications at 38 and 60 GHz using cold-pseudomorphic high electron mobility transistor (cold-pHEMT) technology. Ashtiani's approach was very attractive, and it is followed by many publications that use different technologies and work at different frequencies, only a limited part of which is referred here [102]-[105]. Although this approach was improved to be simpler and occupy less area by the same group [106], the former version is still more popular in the recent years, and it was implemented using different technologies such as HBT ([107],[110]), CMOS ([108], [109], [111]), and pHEMT ([112], [113]).

Another popular approach for the I-Q vector modulators is to use mixers. In this approach, the LO is divided into two orthogonal components. These components are modulated by means of two mixers, and finally, they are combined by means of combiners, amplifiers, couplers, etc. This approach was first presented by Pyndiah et al. ([114], [115]), and this work is followed by Telliez et al. [116]. Pyndiah et al. presented a QPSK modulator with double balanced mixers where Telliez et al. a 64-QAM modulator/demodulator with single balanced mixers. These works are followed by a number of publications [117]-[124], which include the implementations with the new solid-state technologies such as HBT [123], SiGe [124], and CMOS [122]. In these entire works, double balanced mixer topology is employed, which supplies better suppression of even-order distortion compared to a single balanced topology.

Modified versions of the I-Q modulators have also been presented. Primerose et al. [125] used a high pass/low pass topology for the phase (BPSK) modulators in the I-Q modulator topology. Burnsed et al. [126] obtained the four necessary quadrature vectors using high pass/low pass filter instead of couplers or dividers, and obtained

the desired vector by switching and changing the amplitudes of these vectors. Grajal et al. [127], who is followed by Ellinger et al. [128], used three base vectors separated by 120° instead of four quadrature base vectors in order to decrease the complexity. A similar approach is also presented by Altuntaş [129], where a four port 120° phase shifter is implemented. Wu et al. demonstrated two different methods that employ vector summation with the commercial CMOS processes ([130], [131]).

Alternative topologies are also presented for the direct modulators. Goldfarb et al. [132] used the SPDT switching for the implementation of a bi-phase modulator. Philibert et al. [133] employed anti parallel diode pairs as mixers monolithically, which is also applied in [134], where the diode pair is hybrid mounted on a multilayer thin-film multichip module. Another hybrid version that uses diode pair is presented on low-cost RF substrate [135]. In [136], a hybrid vector modulator is presented on ceramic substrate that utilizes FET transistors; and in [137], FET transistors are also utilized for a BPSK modulator on a microstrip based circuit.

Riza et al. proposed the photonic approach for vector modulation [138], and this work is followed by Jemison et al. [139], who used this approach in a vector modulator. The RF MEMS technology is applied for direct modulation circuits only in two examples. The first one is a BPSK modulator reported by Barker and Rebeiz, which is actually a 0/180° phase shifter [140]. The second one is also a BPSK modulator where an RF MEMS SPDT switch is employed [141].

The RF MEMS vector modulator presented in this thesis employ a novel topology, which is based on a triple stub network. This is the same topology that is also used in the RF MEMS phase shifter, where the electrical lengths of the stubs and the interconnection lines are adjusted by means of DMTLs. The vector modulator design only uses low-loss loaded transmission lines, i.e., DMTLs, and the amplitude modulation is obtained by using the resonating structure of open-circuit terminated

stubs. The presented vector modulator is a linear phase design; however, the center frequency can be adjusted between 22.5 GHz and 27.5 GHz. This design requires a high number cascaded DMTL unit sections, which makes it possible to implement only using the "enabling" RF MEMS technology. To the author's knowledge, the idea of amplitude modulation using low-loss transmission lines, the designed vector modulator topology and its RF MEMS implementation is first presented in the frame of this thesis.

# 1.4. Research Objectives and Organization of the Thesis

The goal of this thesis is to investigate the theory of triple stub topology, to design novel impedance tuner, phase shifter, and vector modulator circuits based on this topology, to fabricate the designed circuits using METU RF MEMS Fabrication Process, and to measure the performance of the fabricated circuits. The specific objectives that are used to achieve this goal are given as follows:

- Improvement of the METU RF MEMS Fabrication Process. The current fabrication process should be improved for more reliable and higher performance operation of the fabricated RF MEMS structures.
- Design, modeling, fabrication, and measurement of the RF MEMS triple stub impedance tuner. A reconfigurable, wide impedance coverage impedance tuner should be designed and fabricated, which demonstrates the possibilities that cannot be provided with other technologies.
- Investigation of the theory of triple stub topology. The theory of triple stub topology should be investigated in detail for ideal and non-ideal cases.
  Possible applications of this topology should be demonstrated both theoretically and experimentally.

- Design, modeling, fabrication, and measurement of the novel RF MEMS phase shifter. The novel triple stub topology based phase shifter should be designed, fabricated, and its performance and advantages should be demonstrated.
- Design, modeling, fabrication, and measurement of the novel RF MEMS vector modulator. The novel triple stub topology based vector modulator should be designed, fabricated, and its performance and advantages should be demonstrated.

This thesis is composed of seven chapters. Following this introductory chapter, Chapter 2 describes the METU RF MEMS Fabrication process, and also gives the optimization studies that are made for the improvement of the current fabrication process.

Chapter 3 gives the design, modeling, fabrication, and measurement results of the RF MEMS impedance tuner. This chapter also includes the design, implementation, and the usage of the Multi-pin MEMS Automated Measurement Setup.

Chapter 4 investigates the theory of the triple stub topology. In this chapter, theoretical and numerical analysis of the triple stub topology is presented for both ideal and lossy transmission line cases. Following that, the possible novel configurations such as phase shifter, vector modulator, power divider, etc. are explored, and design methodologies are proposed.

Chapter 5 starts with the application of the triple stub theory on the implementation of the novel RF MEMS phase shifter. It continues with the design, modeling, and the fabrication results. The chapter concludes with the discussion of the measurement results of the RF MEMS phase shifter.

Similar to Chapter 5, Chapter 6 presents the application of the triple stub theory on the implementation of the novel RF MEMS vector modulator. It gives the design, modeling, the fabrication results, and the discussion of the performance of the designed RF MEMS vector modulator.

Finally, Chapter 7 concludes the thesis where it summarizes the achievements of this thesis, and discusses the future research possibilities.

# **CHAPTER 2**

# METU RF MEMS FABRICATION PROCESS AND ITS IMPROVEMENT

# 2.1. Introduction

The fabrication of the RF MEMS structures is one of the most critical parts of this thesis. For this purpose, a surface micromachining fabrication process has been developed at METU-MEMS Center fabrication facilities in the past 9 years with efforts of several researchers<sup>1</sup>. With its current form, the fabrication process developed at METU, namely "METU RF MEMS Fabrication Process", is mature. It is demonstrated in the frame of this thesis and in the former work of METU RF MEMS Group that it is possible to implement complicated structures that employ many RF MEMS switches [142]. Nevertheless, the development phase is still continuing, and the fabrication process is always open for further development. The next section of this chapter gives the details of the METU RF MEMS fabrication process. This is followed by the optimization of some process steps that is developed in the frame of this thesis in order to improve the performance and the reliability of the RF MEMS components.

<sup>&</sup>lt;sup>1</sup> The current METU RF MEMS fabrication process is mainly developed together with Dr. Kağan Topallı. Mr. Hüseyin Sağkol, who is the first researcher of the METU RF MEMS Group, and Mr. Halil İbrahim Atasoy also contributed to the development of former or present versions. Mr. Engin Ufuk Temoçin, Mrs. İpek İstanbulluoğlu Turhan, Mr. Ömer Bayraktar, Mr. İlker Comart, and Mr. Çağrı Çetintepe has also involved and contributed to the application of the fabrication process in the past years. The optimization of the process development that is given at the end of this chapter is developed only by the author, and the other researchers have involved and contributed to the application of the fabrication steps.

# 2.2. METU RF MEMS Fabrication Process

#### 2.2.1. Process Layers and Material Selection

The METU RF MEMS process is mainly composed of two metal, one isolation, and one resistive layer. A sacrificial layer is also used to supply the separation between the two metal layers. The first metal layer is used to define the coplanar waveguides (CPW) for the RF signal propagation as well as the DC lines for the actuation of the RF MEMS switches. The second metal layer is used to form the MEMS bridges, which are the mechanically movable parts of the structure. The isolation layer is used to prevent the DC short between the first and the second metal layers, which is necessary for the operation of the RF MEMS switches. In addition to that, it also forms the dielectric layer of the metal-insulator-metal (MIM) capacitance in both RF MEMS switches and other components. The resistive layer is used to form resistances that are used to isolate the RF Ines from the DC lines. This is also necessary for the proper operation of the RF MEMS switches.

The METU RF MEMS fabrication process, with its current form, is implemented on 500 µm-thick, 4", low-loss quartz substrates. These substrates have relative dielectric constant of  $\varepsilon_r$  = 3.8, and their loss tangent can be better than 0.001 up to 60 GHz [143]. The electric fused quartz substrates [143] are chosen for the METU RF MEMS fabrication process considering antenna and millimeterwave applications. The METU process was formerly done on glass substrates, which have measured loss tangent of  $tan \delta$  = 0.015-0.020. These substrates were left due to increasing loss performance of CPW lines, which will be explained quantitatively in the following parts of this chapter.

The METU RF MEMS fabrication process is composed of 7 layers and 6 lithography steps. The selection of the materials for each layer is quite important both from the

performance and the fabrication point of views. The material selection is a hard task as the deposition and the definition of each layer should not disturb any of the remaining layers. The selection should also be considered to preserve the RF performance since material properties may the limit the performance of the designs. Several materials and approaches have been tried for a properly working, reliable RF MEMS fabrication process in the past experience, which are explained in [144] and [145]. The current version is the 4<sup>th</sup> generation fabrication process that is developed at METU. The seven layers that are used in the fabrication process are summarized in Table 2.1.

Layer Name	Material	Abbreviation	Deposition Met.	
Resistive layer	Silicon chromium	SiCr	RF sputtering	
Adhesion layer	Titanium	Ti	DC sputtering	
First metal layer	Gold	Au	DC sputtering	
Isolation layer	Silicon nitride	SiN	PECVD	
Sacrificial layer	Polyimide	PI	Spin coating	
Strengthening layer	Gold	Au	Electroplating	
Structural layer	Gold	Au	DC sputtering	

Table 2.1: The materials used in the METU RF MEMS fabrication process.

The physical properties such as electrical conductivity, mechanical stress, thermal conductivity, etc. of the selected materials play an important role on the performance of the fabricated structures. The properties of the selected materials are very sensitive to deposition conditions, and they are usually not equal to the bulk material properties. The following properties of the materials are important, and the below mentioned consequences may occur due to the changes in the physical properties of the selected materials:

• *Resistive layer:* The electrical conductivity of this layer is important, and it determines the isolation level between the RF and the DC lines.

- Adhesion layer: This layer is used as an adhesion layer between the substrate and the first metal layer. The electrical conductivity of the adhesion layer affects the insertion loss of the RF MEMS structures.
- First metal layer: The electrical conductivity of the first metal layer is extremely important, and it dominantly determines the insertion loss of the RF MEMS structures. The roughness of this layer also affects the downstate performance of the capacitive RF MEMS switches.
- Isolation layer: The relative dielectric constant and the surface roughness of the isolation layer directly affect the downstate performance of the capacitive RF MEMS switches. These properties cause a strong degradation in the capacitance level if the layer is not deposited properly.
- Sacrificial layer: The mechanical stress of the sacrificial layer is important. This is because the sacrificial layer is a thick layer, and coated on more than 99% of the whole wafer area. If the stress level of this layer is high, it may buckle the wafer; moreover, it may also affect the stress level of the above layers.
- Strengthening layer: The layer is used to strengthen the anchor points of the RF MEMS switches so that they are resistant to the buckling. The mechanical stress of the strengthening layer is important since it affects the stress level of the structural layer. The electrical conductivity of this layer is also important, and it affects the insertion losses of the RF MEMS structures.
- Structural layer: The mechanical stress of this layer is vitally important, and it should have almost zero residual stress and controlled stress gradient. If the stress level of this layer is not very low, the MEMS bridges are buckled upwards or downwards depending on the situation, which directly disturbs

the RF performance, the insertion losses, actuation voltages, and lifetime of the RF MEMS switches.

Table 2.2 gives the thicknesses of the layers and the measured physical parameters of the materials that are used in the fabrication process. The electrical conductivities have been measured with four-point probe measurement method using Magnetron Model M-700 four point probe system. The residual stresses have been measured with the wafer buckling method using the Veeco Dektak 8 surface profiler.

Material	Thickness (Å)	Electrical Cond. (S/m)	Residual Stress (MPa)	Relative permittivity					
SiCr	1000	10000	Not measured	Not measured					
Ti	200	Not measured	-740	NA					
Au	9000	$3 \times 10^7$	Not measured	NA					
SiN	3000	NA	42	7					
PI	15k-20k	NA	18 <sup>1</sup>	2.9 <sup>1</sup>					
Au	20k-30k	$1 \times 10^{7}$	Not measured	NA					
Au	11k	$\sim 2 \times 10^7$	26	NA					
<sup>1</sup> These values are provided by the manufacturer [146].									

Table 2.2: The measured physical properties of the METU RF MEMS fabrication process materials.

# 2.2.2. Process Flow

The METU RF MEMS fabrication process flow can be visualized as in Figure 2.1 and summarized as follows:

i) The process starts with the organic cleaning of the quartz wafers using piranha solution.

- ii) The wafers are dipped into buffered HF for increasing the surface roughness of the wafers, which aims to increase the first metal layer adhesion.
- iii) SiCr resistive layer is deposited using RF sputtering and patterned using wet etching (Figure 2.1-(a), Mask 1).
- iv) Ti/Au first metal layer is deposited using DC sputtering and patterned using wet etching (Figure 2.1-(b), Mask 2). Ti layer is used as adhesion layer between the quartz substrate and Au layer.
- v) SiN isolation layer is deposited using plasma enhanced chemical vapor deposition (PECVD) and patterned using the reactive ion etching (RIE) ((Figure 2.1-(c), Mask 3).
- vi) Polyimide sacrificial layer is deposited using spin coating, patterned using lithography, and cured (Figure 2.1-(d), Mask 4).
- vii) Au structural layer is deposited using DC sputtering (Figure 2.1-(e)).
- viii) Au strengthening layer is selectively deposited on the anchor point of the RF MEMS switches (Figure 2.1-(f), Mask 5).
- ix) Au structural layer is patterned using wet etching (Figure 2.1-(g), Mask 6).
- x) The sacrificial layer is wet etched, and the structures are released using critical point drying system (Figure 2.1-(h)).

The METU RF MEMS fabrication process is optimized for the implementation of the capacitive contact RF MEMS switches. It was also demonstrated that it is possible to implement metal contact RF MEMS switches successfully using this process [147]; however, the lifetime of the switches needs to be improved. This can easily be achieved by some modification without changing the process flow significantly.

The details of the METU RF MEMS fabrication process flow are given in Appendix.

















(e)





(f)



Figure 2.1: (a) The flow of the METU RF MEMS fabrication process.

#### 2.3. The Optimization of METU RF MEMS Fabrication Process

A number of successful studies have been demonstrated that are implemented using the METU RF MEMS fabrication process ([142], [144], [145], [147]-[150]). However, two problems have always been observed in the measurement results of the fabricated RF MEMS structures in all of these studies. The first problem is that the insertion losses of the measured structures are much higher than the simulation results. The measured insertion losses are also higher than the results presented in the literature. The dominant loss mechanism for the RF MEMS structures is usually the loss of the first metal layer, and the conductivity of the gold layer is the primary responsible that determines the amount of the loss. Two examples are demonstrated in the following figures. Figure 2.2 shows the simulated impedance coverage of a triple stub RF MEMS impedance tuner compared with the measurement results. The right hand side figure shows that the increasing loss significantly reduces the impedance coverage. The second example is the degradation of the return loss of an RF MEMS frequency tunable antenna [49], which is given in Figure 2.3. It is clearly observed from the figure that the antenna return loss is also low even at the frequencies that it does not radiate, which means that the antenna dissipates the input power.

The second problem is the height of the MEMS bridges. In all of the above mentioned studies, it is observed that the MEMS bridge heights differ from the expected values due to the buckling of the MEMS bridges. The buckling can be upwards or downwards, and the residual stress and/or stress gradient of the structural gold layer is the primary responsible that determines the amount of the buckling. The examples for both cases are demonstrated in Figure 2.4. The change in the MEMS bridge height seriously affects the performance of the RF MEMS structures. This is because MEMS is generally used as tunable loading capacitances, and the change of these capacitances may distort the design performance, shift the design frequency, increase the losses, etc.



Figure 2.2: The comparison of the simulated and the measured impedance coverage of an RF MEMS impedance tuner for different unloaded line loss values.



Figure 2.3: The comparison of the measured and simulated return loss performance of a tunable RF MEMS antenna [49].



(a)



Figure 2.4: Examples of the RF MEMS structures from previous runs that suffer from buckling due to residual stress (a) Compressive stress level (b) Tensile stress level.

To overcome the problems of the current fabrication process, separate optimization studies have been made to obtain a high-conductivity Au first metal layer and a lowstress Au structural layer. These studies will be explained in the following subsections.

#### 2.3.1. Optimization of the Conductivity of the Ti/Au First Metal Layer

The first metal layer of the METU RF MEMS fabrication process is composed of Ti and Au layers, both of which are deposited using sputtering. Sputtering is a physical vapor deposition (PVD) method, and in this method, the deposition conditions such as type of sputtering setup, ion energy, substrate bias, gas type and pressure, atomic mass of gas and target atom, substrate temperature, angle of deposition and emission, and target are some of the important parameters that control the properties of the deposited films [151]. These properties control a number of film properties namely residual film stress, thickness, density, porosity, film morphology, elastic modulus, hardness, electrical conductivity, TCR, chemical etch rate, optical reflectance, color, etc.

For the first metal layer of the METU RF MEMS fabrication process, the important film parameters are the electrical conductivity, optical reflectivity, which is directly related with surface roughness, and residual stress. The electrical conductivity is the vitally important for the performance of the fabricated RF MEMS structures. For this purpose, an optimization study is made in order to obtain highest possible electrical conductivity with acceptable surface roughness and residual stress.

There are a high number of publications on the characterization of the thin films that are deposited with sputtering. The theoretical and experimental studies present the correlation of the deposition conditions with the film properties. The PVD theory says that the deposited film microstructure is modeled with zones that are determined due to the ion energy and the substrate temperature with respect to the melting temperature of the deposited material [152]. It is stated that the *zone T* is the most suitable region for device fabrication since the films in this zone yields refractory coating at relatively low temperatures. Moreover, the grains are small, dense, and formed in columnar forms [153]-[155], and the resulting film density is high [156]. Since high density films are necessary for obtaining high

conductivity, the films in this zone are feasible for the first metal layer of the METU RF MEMS fabrication process.

At the METU-MEMS Fabrication Facility, the tool used for the first metal layer deposition is Bestec planar magnetron DC sputtering tool [157]. Using this tool, the power (or DC bias), which is actually ion energy, gas pressure, substrate temperature, and the distance are the deposition conditions that can be controlled by the user. Ar is used as the inert gas for sputtering. Considering the above discussion and the controls of the sputtering tool at the facility, a number of deposition trials are made, in which the major target is to maximize the Au conductivity. The process conditions of the trials are given in Table 2.3.

During the trials, the target is to obtain proper grain formation, and hence, increase the conductivity by increasing the film density as explained above. For this purpose, the substrate temperature is elevated, and the deposition rate is lowered by decreasing the applied power and increasing Ar pressure. The surface resistances of the deposited films are measured using the four-point probing technique, the thicknesses are measured using Veeco Dektak 8 surface profiler, and the conductivity values are extracted using the standard formulation. In the final recipe, which is Trial No 6, an electrical conductivity of  $3 \times 10^7$  S/m is achieved. This is almost three times of the former conductivity value, which was not measured directly, but extracted from the RF measurements. This is 73% of the bulk conductivity of gold, which is  $4.1 \times 10^7$  S/m. However, the measured conductivity of the metals that are deposited with sputtering can be different than their bulk conductivities [158], and the obtained value is close to the maximum values reported in the literature to the author's knowledge.

Trial No	Power (W)	Pressure (×10 <sup>-3</sup> mBar)	Temp. (°C)	Thickness (Å)	End Point (Å/s)	Rate (Å/s)	Sheet Resistance $(\Omega/\Box)$		Conductivity (S/m)
							min max		
1	150	2	RT	3100	1500	2.1	0.123	0.154	1.92×10 <sup>7</sup>
2	150	3	150	4600	2000	2.3	0.088	0.091	2.08×10 <sup>7</sup>
3	300	3	150	4800	2000	4.4	0.091	0.096	1.94×10 <sup>7</sup>
4	100	1.5	150	4600	2000	1.5	0.075	0.08	2.27×10 <sup>7</sup>
5	100	13	150	5500	2000	1.4	0.053	0.055	2.77×10 <sup>7</sup>
6	100	20	150	6200	2000	1.2	0.045	0.045	3.03×10 <sup>7</sup>

Table 2.3: The summary of the sputtering process parameters that are used for theelectrical conductivity optimization of the first metal layer.

Another important parameter for the losses of the first metal layer is the substrate loss tangent. The METU RF MEMS fabrication process is used to be implemented on Pyrex glass substrates, which has  $\varepsilon_r = 4.6$  and  $tan \ \delta = 0.005$ . These values are the values reported by the manufacturer at low frequencies [159]. However, at the microwave and millimeterwave frequencies, it is reported that the loss tangent can go up to 0.015-0.020, and it may change from wafer to wafer [28]. In order to reduce also the dielectric losses, quartz substrates are used for the METU RF MEMS fabrication process. The quartz substrates have  $\varepsilon_r = 3.8$  and better than  $tan \ \delta = 0.001$  up to 60 GHz [143].

For the verification of the obtained recipe and the performance of the quartz substrates, a CPW test structure with gap/signal/gap dimensions  $G/W/G = 25 \,\mu\text{m}/170 \,\mu\text{m}/25 \,\mu\text{m}$  are fabricated on glass and quartz substrates with first metal layer that is deposited both the optimized recipe and the old recipe. The measurement results of the fabricated CPW test structures are given in Figure 2.5. Table 2.4 gives the summary of the extracted material properties and the CPW losses. The results presented in the table also verify the DC measurements of the optimized first metal layer process and the expected tangent losses of both substrates.

Wafer Mat.	Metal. Process	Metal. Thickness (μm)	Sim. Loss <sup>1</sup> (dB/m)	Ext. Loss <sup>1</sup> (dB/m)	Extracted σ (S/m)	Extracted tan δ					
Quartz	Optimized	1.2	63.3	66.6	$3 \times 10^7$	0.001					
Glass	Optimized	1.2	103	99.7	$3 \times 10^7$	0.015					
Glass	Old	0.6	188	172	$1 \times 10^7$	0.020					
<sup>1</sup> The loss	<sup>1</sup> The loss values are calculated and extracted at 20 GHz.										

Table 2.4: The extracted material properties and losses for the optimized and old Ausputtering processes on quartz and glass substrates.



Figure 2.5: The measurement results of the CPW test structures that are fabricated on glass and quartz substrates with optimized and old first metal layer recipes (a)  $|S_{11}|$  (b)  $|S_{21}|$ .

An optimization study is also conducted that aims to increase the conductivity of the strengthening layer, which is deposited using electroplating. The Au electroplating solution is a cyanide-based solution. This process was previously used to increase the thickness of the first metal layer. However, the process was quitted since the conductivity of this layer was too low. In the optimization study, the current density and the plating waveform is tuned in order to increase the conductivity of the electroplated Au layer. The conductivity of the deposited films is again measured from using four-point probing technique. Table 2.5 shows the measured material properties, which shows that the conductivity of the electroplated at most up to  $1 \times 10^7$  S/m with this solution. This value is very low, and it is limited with the solution, which was not designed for microelectronic applications.

Table 2.5: The measured material properties for the Au electroplating optimization trial processes.

Trial No	Material	Current (mA)	Area (cm <sup>2</sup> )	Time (min)	Temp. (°C)	Duty Cycle (%)	Thicl (µ	kness m)	Resistance ( $\Omega/\Box$ )		Conductiv	vity (S/m)
							min	max	min	max	min	max
1	Au	228	57	20.00	32	50	1.55	1.73	80x10 <sup>-3</sup>	86x10 <sup>-3</sup>	0.71×10 <sup>7</sup>	0.69×10 <sup>7</sup>
2	Au	114	57	16.00	32	50	0.96	1.14	93x10 <sup>-3</sup>	96x10 <sup>-3</sup>	1.02×10 <sup>7</sup>	0.89×10 <sup>7</sup>
3	Au	114	57	16.00	32	10	1.04	1.32	89x10 <sup>-3</sup>	95x10 <sup>-3</sup>	0.97×10 <sup>7</sup>	0.91×10 <sup>7</sup>
4	Au	340	57	8.00	32	50	1.68	2.3	77x10 <sup>-3</sup>	84x10 <sup>-3</sup>	0.70×10 <sup>7</sup>	$0.58 \times 10^{7}$
5	Au	114	57	16.00	32	50	0.96	1.13	98x10 <sup>-3</sup>	103x10 <sup>-3</sup>	0.96×10 <sup>7</sup>	0.75×10 <sup>7</sup>

Gold salt added to the electroplating solution as refreshment on the 5<sup>th</sup> trial.

# 2.3.2. Optimization of the Stress of the Au Structural Layer

The structural layer of the METU RF MEMS fabrication process is Au, and it is deposited also by sputtering. This layer should have minimum residual stress and stress gradient. As explained in the previous subsection, the sputtered film properties depend upon many deposition conditions. There are several publications in the literature that explain the formation of the stress and investigate

the effects of deposition conditions on the stress behavior. These publications are first demonstrated and leaded by Thornton and Hoffman [160]-[164], and followed by other researchers [165]. It is stated in these publications that the residual stress is predominantly compressive. The stress is formed because of the bombardment of the growing film surface by the energetic atoms arriving to the surface, which is the caused by the basic mechanism of sputtering. This leads to the generation of excess "interstitials" on the surface. The amount of interstitial formation is reported to be dependent upon incident atom energy and the sputtering gas pressure.

It is observed by the authors that the stress is compressive for low gas pressures and low deposition rates, where the incident atom energy is high. There also occurs a compressive-to-tensile transition for the deposited material. This point is mostly dependent upon the deposition conditions such as power and gas pressure that affect the incident atom energy. This behavior is explained by "atomic peening" model, which states that damage is induced at the growing film surface during either the bombardment of the highly energetic atoms or the atomic recoil. The atomic peening is also states that the amount of damage, i.e., the formed stress is correlated with the atomic mass of the deposited material, and it is shown that the materials with larger masses have more compressive stresses.

The effects of many deposition parameters on the film stress are explained in [164]. However, the sputtering tool used in METU-MEMS Fabrication Facility allows only a limited number of process parameters to be changed due to its nature. So, several trials are made in order to obtain low-stress Au films. During the trials, the applied power, the Ar pressure, and the substrate temperature are used as the optimization parameters, which are actually the main determinants of the stress. The stress level is targeted to be slightly tensile due to reasons that will be explained later. The temperature is also elevated in order to increase the adatom mobility, i.e., the mobility of the adsorbed atoms, so that the stress level is reduced [166]. The stress in thin films has been measured experimentally using different methods [166]-[168]. In this thesis work, the stress levels are extracted from the wafer curvature using Stoney formula [166]. The stress levels are measured using Veeco Dektak 8 surface profiler. Table 2.6 summarizes the process conditions and the measured stress levels of the structural layer optimization trials.

Rec.	Wafer	. Act	Power	Pressure	Temp.	Thickness	Rate		Stress (MPa)		
No	No	iviat.	(W)	(×10 <sup>-3</sup> mBar)	(°C)	(Å)	(Å)	ten	sile	compressiv	
								avg	max	avg	max
1	-	Al	172	10	120	7000	-	81	365	-	-
2	-	Al	300	14	120	8200	-	47	70	-	-
3	-	Au	213	10	RT	6300	2.7	27	32	-	-
4	RD#11	Ti	300	2	150	~200	1.2				
		Au	100	6.6	150	5100	1.6	18	1.2	-6	-16
5	RD#12	Ti	300	2	150	~200	1.2				
		Au	100	9.3	150	5400	1.6	72	83	0	0
6	RD#13	Ti	300	2	150	~200	1.2				
		Au	100	12	150	5800	1.5	81	88	0	0
7	RD#14	Ti	300	2	150	~200	1.2				
		Au	100	14.7	150	6300	1.4	99	112	0	0
8	RD#15	Ti	300	2	150	~200	1.2				
		Au	100	4	150	4900	1.7	52	57	0	0
9	RD#17	Ti	100	9.3	150	-	0.2				
		Au	100	12.7	150	5800	1.5	0	0	-22	-25
10	RD#18	Ti	100	9.3	150	-	0.2				
		Au	100	12.7	150	5600	3.2	0	0	-29	-34
11	RD#22	Ti	100	9.3	150	1960	0.2	134	168	0	0
12	RD#23	Ti	300	2	150	2400	1.2	0	0	-726	-740

Table 2.6: The process conditions and the measured stress levels of the recipes that aremeasured on blank wafers for the structural layer stress optimization.

The results given in Table 2.6 include the stress levels of sputtered Au films as well as the stress levels of sputtered Ti and Al films. Sputtered Ti is used as the adhesion layer for sputtered Au, and it is also very effective although it is much thinner when compared to Au. The stress level of the sputtered Al is investigated since it is used to be used as the structural layer of the METU RF MEMS fabrication process.

During the trials, the power level is first fixed to the level of the maximum conductivity process, and the Ar pressure is swept in a large range in order to find the compressive-to-tensile transition point (zero transition). The stress level increase in tensile direction with increasing Ar pressure as stated by the theory. Then, the deposition conditions of the Ti adhesion layer are changed without changing the deposition conditions of the Au layer. The residual stress of the Ti layer is also measured separately as explained above, and it is found out that the stress level is changing in an extremely wide range from -740 to +168 MPa (Recipe 11 and 12). The effect of Ti is easily observed on the results given in Table 2.6 (Recipe 6 and 10). Finally, two solution points are found, which are Recipe 3 and Recipe 4. The main difference between these recipes is the substrate temperature. One other point is that Recipe 3 is slightly tensile, and it is preferred due to the reasons that will be explained in the following parts of this chapter.

In order to verify the optimized recipes, they are applied as the structural layer of METU RF MEMS fabrication process. For the verification, the structural layer is implemented only together with the sacrificial layer on blank or Ti/Au coated glass substrates, and no other layers are deposited. Then, the sacrificial layer is etched, and the buckling amount is measured from the released MEMS test structures. Table 2.7 presents the summary of the optimization trials on MEMS test structures. The buckling amount is measured on a MEMS clamped-clamped beam, which is 300 µm-long and 80 µm-wide, using Veeco Wyko NT 1100 optical profiler.

Trial No	Wafer No	Mat.	Power	Pressure (×10 <sup>-3</sup>	Temp. (°C)	Thick. (Å)	Sacr. layer height (µm)		Sacr. spin spd	Meas. stress	Meas.
			(,	mBar)	( 9)	(,,,	min	max	(rpm)	(Mpa)	Suching
1	R#7	Au	213	10	RT	10000	2.5	2.76	3500	26	-1.5
2	RD#6	Ti	300	2	RT	200	2.24	2.36	4000	-740	-1.4
		Au	213	10	RT	10000				26	
3	RD#7	Ti	300	2	150	200	2.37	2.73	4000	-740	-1.3
		Au	100	20	150	10000				> 100	
4	RD#15	Ti	300	2	150	200	2.5	2.93	3000	-740	-1.2
		Au	100	6.6	150	11000				~0	
5	RD#15-2	Ti	300	2	150	200	2.5	2.93	3000	-740	-1.1
		Au	100	6.6	150	11000				~0	
6	RD#16	Ti	100	9.3	RT	200	2.61	2.75	3000	170	<-2.5
		Au	213	12.7	RT	10000				-	
7	RD#19	Ti	300	2	RT	2000	2.4	2.56	3750	-740	-1.4
		Au	213	10	RT	10000				26	
8	RD#20	Ti	300	2	RT	600	2.42	3.11	3750	-740	-1.4
		Au	213	10	RT	10000				26	
10	R#3	Au	213	10	RT	10000	2.54	2.98	3750	26	-1.4
11	RD#22	Au	213	10	RT	10000				26	+0.9
		Ti	300	2	RT	600	2.35	2.93	3750	-740	
12	RD#23	Au	213	10	RT	10000	2.27	2.9	3750	26	~0

Table 2.7: The process conditions and the measured buckling levels of the recipes that are measured on released MEMS bridge for the structural layer stress optimization.

#### Notes:

1) The measured stress of the Au layer in Trial 1 and Trial 10 is 32 MPa (Table 2.6 Recipe 3).

2) The measured stress of composite Ti/Au layer in Trial 4 is ~0 MPa (Table 2.6 Recipe 4).

3) Trial 5 is the same as Trial 4 except the released structures are annealed 1 hour @ 200 %.

4) The measured stress of composite Ti/Au layer in Trial 6 is -25 MPa (Table 2.6 Recipe 9).

5) Ti is deposited upon Au layer in Trial 11.

6) The strengthening layer is added to the process (anchor filling).

Recipe 3 is employed in Trial 1 where the MEMS clamped-clamped beam is buckled 1.4  $\mu$ m downwards. At this point, this result was not expected, and it is concluded that the residual stress of the structural layer changes due to the thermal processes. Then, different recipes for Au are investigated; moreover, a different method is also applied, which is controlling the beam stress by using a bilayer structure [169]. In this case, the stresses coming from different layers compensate each other so that a buckling-free beam is obtained. For this purpose, Ti adhesion layer is added with different thicknesses (Trial 2, 7, and 8), Ti adhesion layer with tensile stress level is employed (Trial 6), other Au recipes are used (Trial 3 and Trial 4), annealing is used

(Trial 5), and finally Ti layer is deposited on top of Au layer (Trial 10). However, the resultant buckling was always downwards with different amounts. Only in Trial 10, the MEMS bridge is buckled 0.9  $\mu$ m upwards, which opens the way to control buckling amount by means of Ti layer thickness.

The last result proves that this method can be used to obtain buckling free MEMS structures by changing the thickness of Ti layer; however, it is important to understand the reason behind the buckling of the beams in previous trials. To enlighten the situation, the beam theory is investigated using both analytical and numerical methods<sup>1</sup>. For the investigation, only residual stress is assumed, and no stress gradient is applied. The test structure is again the MEMS clamped-clamped beam, which is fabricated and measured for the optimization trials.

The beam theory states that if the residual stress is compressive, the beam can be buckled to upwards, downwards, higher order modes (sinusoidal for example), or no buckling is observed. This is because the anchor points are mechanically grounded, and the analytical solution of the problem results that these points are mathematically instable. As a result, all of the above mentioned cases may occur for a uniform beam. However, if there is an imbalance due to, for example a fabrication defect, it is expected to determine the state of the beam. If the residual stress in tensile, the beam theory states that there should be no buckling.

This phenomena is also verified with Coventorware [170] and SUGAR [171] simulations. Figure 2.6-(a) and (b) show that the beam is not buckled in the longitudinal direction from when it is fixed under -50 MPa compressive stress. However, if an imbalance is placed by thinning the beam at a position, the beam is buckled downwards as in Figure 2.6-(c) and (d). The beam is buckled 5.5  $\mu$ m in this

<sup>&</sup>lt;sup>1</sup> The theoretical analysis and Coventorware simulations are carried out by Korhan Şahin. The SUGAR simulations and other necessary Matlab simulations are carried out by Çağrı Çetintepe.

case, which is a significant result. The beam is not buckled under 50 MPa tensile stress as expected, which is shown in Figure 2.6-(e) and (f).

The fabricated MEMS structures do not have ideal mechanical grounds; instead, they are nominally clamped at the anchor points and connected to the substrate by only short beams of same thickness with the original beam. The anchor points in this case behave as imbalance points, and they change the beam shape under the presence of residual stress. For example, the beam buckles sinusoidally under -100 MPa compressive stress as in Figure 2.6-(g) and (h). The most important result is obtained when 50 MPa tensile stress is applied to the nominally clamped beam. The Coventorware simulations show that the beam is buckled downwards as in Figure 2.6-(i) and (j). This is the result observed also from the measurements. However, the amount of buckling is less than what is expected. SUGAR simulations also show similar response, and they give a better estimation of the buckling amount, which is about 1.4  $\mu$ m.

The above analysis clearly demonstrates that the buckling is not due to the amount of the residual stress, but it is because of the nonideal clamping of the MEMS beams. As a solution to this problem, the anchor points of the MEMS beams are filled in order to strengthen the clamping at these points. For this purpose, the strengthening layer is added to the process, which is used to fill the anchor points of the MEMS beams by Au electroplating. The measurement results of Trial 11 show that MEMS beams without any buckling are achieved. Figure 2.7 shows the measured 3D views and the beam shapes of the MEMS test beams from Trial 11, and comparison with the previous trials results, Trial 2 and Trial 10. The measurement results are obtained using Veeco Wyko NT 1100 optical profiler measurements. Figure 2.7 also includes the SEM picture of the MEMS test beam without any buckling.



(a) Ideal clamping, compressive - No buckling.

(b) Closer view of (a) at the anchor point.



(c) Ideal clamping with imbalance, compressive -Downwards buckling.



(d) Closer view of (c) near the imbalance point.



(e) Ideal clamping, tensile - No buckling.



(f) Ideal clamping with imbalance, tensile -No buckling.



(g) Nominal clamping, compressive -Sinusoidal buckling.

(h) Closer view of (g) at the anchor point.



Downwards buckling.

(j) Closer view of (i) at the anchor point.

Figure 2.6: The snapshots of the MEMS beams simulated using Coventorware.



(a) Measured 3D view optical profiler view of the MEMS test beam from Trial 2 (RD#6).


(b) Measured beam height variation of the MEMS test beam from Trial 2 (RD#6).



(c) Measured 3D view optical profiler view of the MEMS test beam from Trial 10 (RD#22).



(d) Measured beam height variation of the MEMS test beam from Trial 10 (RD#22).



(c) Measured 3D view optical profiler view of the MEMS test beam from Trial 11 (RD#23).



x 250 4.0KV LEI 100µm JEOL 11/9/2007 (f) SEM photograph of the MEMS test beam from Trial 11 (RD#23).

Figure 2.7: The optical profiler measurements and SEM pictures of the MEMS test beams.

Trial 11 also shows good results in other of MEMS structures that have different geometries. Figure 2.8 shows the measured 3D views and the beam shapes of two example structures. The first structure is a 600  $\mu$ m-long MEMS clamped-clamped beam, and it is buckled by only 0.2  $\mu$ m throughout the whole beam. The previous fabrication results showed that this beam buckles by about 2  $\mu$ m downwards. Another example is a beam that is anchored at 4 points on the sides and at 2 points along the beam. This structure is buckled by 0.2  $\mu$ m at maximum. The previous fabrication results showed that this beam buckles by about 1.8  $\mu$ m upwards.

An interesting behavior is observed in the second example structure of Figure 2.8. This structure has 4 small anchors that hold the long beam by means of short cantilevers, and 2 wider anchors along the beam that are intended to prevent the buckling in the previous designs. The shape of this beam given in Figure 2.8-(d) starts with upward buckling, which is the characteristic shape of this structure, and this behavior was observed in the previously fabricated structures. However, the beam changes shape and bends downwards as it approaches the wide anchor. The beam has a similar shape with Figure 2.7-(b) in between the two wide anchors. The two wide anchors are not filled with electroplated Au; they are nominally clamped; so, they are "weak" compared to the ones that are filled with electroplated Au. The center part of the beam tries to get shorter due to tensile residual stress, and it stretches the weak anchors from both sides. This measurement shows that the anchor points or any other discontinuities through the longitudinal direction may totally change the beam shape in equilibrium once these anchors or discontinuities nominally clamped.



(a) Measured 3D view optical profiler view of a 600  $\mu m$ -long MEMS beam.



(b) Measured beam height variation of a 600  $\mu m$  long MEMS beam.



(c) Measured 3D view optical profiler view of the MEMS structure with 4 + 2 anchors.



(e) SEM picture of the MEMS structure with 4 + 2 anchors.

Figure 2.8: The optical profiler measurements and SEM pictures of other MEMS structures.

# 2.4. Conclusion

This chapter presents the details of the fabrication process, namely METU RF MEMS fabrication process, which is used to implement the RF MEMS structures in the frame of this thesis. The materials used and their deposition methods are introduced, and the process flow is presented. Also, the bottlenecks of the process

are identified, and solutions are proposed. In this frame, the conductivity of the first metal layer is increased up to  $3 \times 10^7$  S/m by a detailed investigation of the effects of the process parameters. The obtained result is also verified with DC resistivity measurements, and parameters extracted from RF measurements. The residual stress of the structural layer is also minimized, and the possible solutions that prevent the buckling are investigated both theoretically and experimentally. Two solution methods are proposed for this purpose, and a process recipe is developed that result with MEMS test structures without any buckling.

# **CHAPTER 3**

# A RECONFIGURABLE RF MEMS TRIPLE STUB IMPEDANCE TUNER FOR 6-20 GHZ APPLICATIONS

#### 3.1. Introduction

One of the applications that use the advantage of realizing reconfigurable circuits with the RF MEMS technology is impedance tuners and impedance matching networks. RF MEMS impedance tuners (IT) and impedance matching networks (IMN) propose a new aspect to some important problems in RF engineering such as noise parameter (NP) measurement, load-pull analysis, low-noise amplifier (LNA) and power amplifier (PA) matching, and antenna matching.

ITs and IMNs are generally implemented by transmission line based structures such as single stub, double stub, triple stub [172], and double slug [174] topologies at microwave and millimeter-wave frequencies. The idea of reconfigurable RF MEMS ITs and IMNs is to change the electrical lengths of the transmission lines in these structures by means of RF MEMS switches or RF MEMS varactors so that the tuned or matched impedances can be adjusted [67]-[78]. The use of RF MEMS components in the design of ITs or IMNs is due to the fact that these components have very low-loss with respect to their rivals. In this case, it becomes possible to implement structures by cascading a number of RF MEMS components, which brings an unaffordable amount of loss in case of using other microwave components. As a result, reconfigurable ITs or IMNs can be designed that have a high number of RF MEMS components, which results with a high number of impedance states or wide impedance coverage.

The aim in this part of the thesis is to design and implement a universal tuner that can cover the whole Smith Chart in an ultra-wide band. The following subsections of this chapter explain the details of the design, the fabrication, and the measurement results of the triple stub RF MEMS impedance tuner.

#### **3.2. RF MEMS Impedance Tuner Design**

Figure 3.1 shows the block diagram of the impedance tuner (IT). The tuner is composed of three adjustable stubs, two interconnection lines, and three T-junctions that are used to connect the stubs to the interconnection lines. The electrical lengths of the stubs are adjusted by means of RF MEMS switches.



Figure 3.1: The block diagram of the triple stub impedance tuner.

The triple stub topology has been selected for the impedance tuner since the entire Smith Chart can be covered theoretically as the impedance range using this topology. Coplanar waveguides (CPW) are used as the transmission lines of the IT since both the signal and the ground lines are on the same plane. In this case, it is easy to implement shunt RF MEMS switches, and thru wafer via are not necessary. The CPW interconnection lines between the stubs have an electrical length of approximately  $\lambda_g/8$  which was chosen in order to obtain uniform impedance distribution on the Smith Chart.

The center design frequency is selected as 18 GHz. The aim of the design is to cover the whole Smith Chart by a maximum number of impedance combinations at the center design frequency; in addition to that, the design is also targeted to cover as much area as possible on the Smith Chart in an ultra-wide frequency band. For this purpose, 10 RF MEMS switches have been uniformly placed on each stub so that there are  $11^3 = 1331$  impedance states to cover the whole Smith Chart. All the three stubs are identical, and the electrical lengths of the stubs are greater than  $\lambda/2$ at 18 GHz. Having a stub electrical length greater than  $\lambda/2$  is necessary to be able to provide susceptances theoretically in (- $\infty$ , + $\infty$ ) range, which is required to reach each impedance value on the Smith Chart. However, since the positions of the RF MEMS switches are fixed, the  $\lambda/2$  length is divided into 10 discrete steps, and the design targeted to cover the Smith Chart uniformly with discrete impedance points.

#### 3.2.1. RF MEMS Switch (Unit Section) Design

The working principle of the reconfigurable triple stub IT in this thesis is to change the lengths of the stubs by actuating the RF MEMS switch closest to the required stub lengths. In this way, short or open circuited terminations can be obtained with discrete separations, and for this purpose, capacitive shunt RF MEMS switches are used. Because there are 10 RF MEMS switches on a stub and each of them must be actuated independently, a standard capacitive, shunt RF MEMS switch, where the actuation signal is applied between the CPW signal and the CPW grounds, cannot be used. Hence, a special switch design is necessary, which has capacitive coupling from MEMS bridge to CPW ground. The 3D and the top views of the designed RF MEMS switch can be seen in Figure 3.2.





Figure 3.2: (a) 3D view of the RF MEMS switch of the triple stub IT (z-dimension exaggerated) (b) Top view of the RF MEMS switch of the triple stub IT.

The MEMS switch is designed first in Microwave Office v6.5 (MWO) [175] using the circuit model given in Figure 3.3. In the circuit model, all the capacitances have two states.  $C_{sbu}$  and  $C_{sbd}$  stand for the capacitance between the signal line of the CPW and the MEMS bridge for upstate and downstate, respectively, and  $C_{bgu}$  and  $C_{bgd}$  stand for the capacitance between the MEMS bridge and the CPW for upstate and downstate, respectively. In this step of the design, the design goal is to obtain such a switch design that the 1331 impedance states of the designed IT is distributed uniformly on the Smith Chart.



Figure 3.3: The circuit model of the MEMS switch that is used for the design of the triple stub IT.

The most critical parameters for the design are the characteristic impedance of the unloaded line,  $Z_o$ , the periodicity, s, and the upstate and downstate capacitances,  $C_{sbu}$ ,  $C_{sbd}$ ,  $C_{bgu}$  and  $C_{bgd}$ . The upstate capacitances are calculated from the physical dimensions using [177]:

$$C_{sbu} = \varepsilon_0 \left[ 1.15 \frac{W w_{brd}}{h_{equ}} + 2.8(W + w_{brd}) \left(\frac{t}{h_{equ}}\right)^{0.222} + 4.12 h_{equ} \left(\frac{t}{h_{equ}}\right)^{0.728} \right]$$
(3.1)

$$C_{bgu} = \varepsilon_0 \left[ 1.15 \frac{w_{gnd} w_{brd}}{h_{equ}} + 2.8 \left( w_{gnd} + w_{brd} \right) \left( \frac{t}{h_{equ}} \right)^{0.222} + 4.12 h_{equ} \left( \frac{t}{h_{equ}} \right)^{0.728} \right]$$
(3.2)

$$h_{equ} = g + \frac{t_d}{\varepsilon_r} \tag{3.3}$$

where *t* stands for the first metal layer thickness,  $t_d$  stands for the dielectric layer thickness,  $\varepsilon_0$  is the free-space permittivity, and  $\varepsilon_r$  is the relative dielectric constant. The above formulation includes the effects of the fringe capacitances, and it is successful in guessing the capacitance of the MEMS switch generally with less than 10% error when compared to the capacitance that are extracted from full wave EM simulations. So, this formula is very handy for having a good starting point for the design. The downstate capacitances are calculated using:

$$C_{sbd} = \varepsilon_r \varepsilon_0 \frac{W w_{brd}}{t_d}$$
(3.4)

$$C_{bgd} = \varepsilon_r \varepsilon_0 \frac{w_{gnd} w_{brd}}{t_d}$$
(3.5)

The selection of CPW dimensions is important on the design in several ways. First, the total ground-to-ground spacing should be less than:

$$W + 2G < \frac{c}{8f_0\sqrt{\varepsilon_{r,subs}}} \tag{3.6}$$

where  $\varepsilon_{r,subs}$  is the relative dielectric constant of the substrate and  $f_o$  is the maximum operation frequency, in order to minimize radiation losses [22]. The total ground-to-ground spacing is selected as high as possible in order to reduce the unloaded CPW line losses. Having minimum unloaded CPW line loss is another

criteria, and it occurs when W is selected as equal to G, in which case the unloaded line  $Z_o$  becomes around 95  $\Omega$ . Also, the center conductor dimension, W, is directly effective on the capacitances in both upstate and downstate, and the actuation voltage of the RF MEMS switch. Considering the design requirements and all the criteria pointed out above, a set of circuit model parameters and their corresponding physical dimensions are determined.

Once the circuit model parameters of the switch are found, the switch is simulated and optimized by tuning the physical dimensions in the full-wave EM simulation environment, which is HFSS v9.2 [176]. The final dimensions of the RF MEMS switch design are given in Table 3.1. The comparison of the HFSS simulations of the RF MEMS switch with that of the circuit model simulations are given in Figure 3.4. The figures show that the HFSS simulations can successfully be modeled with the circuit model. The circuit model parameters that are extracted from the HFSS simulations are given in Table 3.2.

Parameter	Value	
Wswt	550 μm	
l <sub>swt</sub>	1446 µm	
W	146 µm	
G	150 μm	
l <sub>brd</sub>	1046 µm	
$l_{brd2}$	1246 µm	
Wbrd	100 µm	
Wanc	100 μm	
Wgnd	300 μm	
WSiCr	20 µm	

Table 3.1: The physical design parameters of the RF MEMS switch.



Figure 3.4: The comparison of HFSS simulations with that of the circuit model of the RF MEMS switch that is used for the design of the triple stub IT.

Circuit Model Parameters Extracted from HFSS				
$Z_o$	97.5 Ω	$C_{sbd}$	1.15 pF	
$\mathcal{E}_{r,eff}$	2.32	$C_{bgd}$	4.73 pF	
α	33 dB/m @20 GHz	$L_b$	18.3 pH	
S	550 μm	$R_{bu}/R_{bd}$	2 / 0.18 Ω	
$C_{sbu}$	91.2 fF	$C_U/C_D$	74.2 fF/ 926.3 fF	
$C_{bgu}$	370 fF	<i>R</i> <sub>bias</sub>	4000 Ω	
Interconnection line parameters				
$Z_o$ = 53.5 Ω, $\varepsilon_{r,eff}$ = 2.375, α = 71 dB/m, $l$ = 994 μm				

Table 3.2: The circuit design parameters of the RF MEMS switch design.

For the HFSS simulations, the relative dielectric constant of the silicon nitride dielectric layer is used as  $0.37 \times 7 = 2.59$ . This change is made in order to include the degradation in the down state capacitance of the RF MEMS switches. From the previous measurement experience, it was observed that the extracted capacitance values are about 37% of the ideal parallel plate capacitance values. The decrease in the downstate capacitance is due to the surface roughness of the isolation layer and first metal layer, and insufficient contact force. This amount of degradation is not only observed in the METU RF MEMS fabrication process, and it is close to the ones that are observed in the literature ([34], [37]).

#### **3.2.2. T-Junction Design**

The T-junctions are quite important for the performance of the triple stub impedance tuner, and they should be designed carefully. Otherwise, the junction point adds additional inductance and capacitance that affect the lengths of the stubs and the CPW lines between the stubs. Suspended air bridges were used on all ports of the junction to prevent unwanted modes. The signal line of the CPW lines were narrowed to cancel the capacitance of the air bridges; and hence, to improve matching at the junction point. The T-junction was designed using Microwave Office v6.5 (MWO) [175] EM simulator and the designs were verified with HFSS v9.2 [176]. A more detailed analysis will be presented in Chapter 5.2.1. The top view of the designed T-junction can be seen in Figure 3.5, and its dimensions are given in Table 3.3.



Figure 3.5: The layout of the T-junction that is used in the design of the triple stub IT.

Parameter	Value	
$W_1$	178 μm	
$G_1$	21 µm	
$W_2$	62 µm	
$W_3$	146 µm	
$G_3$	150 μm	
ljunc	1064 µm	
Wjunc	714 μm	
Wbrd	24 µm	

Table 3.3: The physical design parameters of the T-junction design.

#### 3.2.3. The Simulations of the Triple Stub Impedance Tuner

The simulations of the RF MEMS triple stub IT is made using MWO v6.5. The S-parameters of the RF MEMS switches, T-junctions, and CPW interconnection lines are cascaded, and the overall performance of the IT is investigated. The Smith Chart coverage of the designed IT for 6-20 GHz band is given in Figure 3.6. Each figure includes  $11^3$  impedance states. The center frequency of the design is 18 GHz. The area of the final IT design measures  $8.6 \times 8.6 \text{ mm}^2$ . The interconnection line CPW parameters are  $Z_o = 53.5 \Omega$ ,  $\varepsilon_{r,eff} = 2.375$ ,  $\alpha = 71 \text{ dB/m}$ , and  $l = 994 \mu\text{m}$ .







Figure 3.6: The Smith Chart coverage of the designed IT in 6-20 GHz band for 1331 impedance points.

The above figures show that the design is successful in covering almost entire Smith Chart in 6-20 GHz band. 20 GHz is actually not a limit for the operation of the design; however, due to the limitations in the measurement setup, the design and modeling is limited to this frequency.

#### 3.2.4. Case Studies

To demonstrate the operation of the IT, some sample impedance transformations are presented. Figure 3.7 shows the performance of the IT when it is adjusted to transform 50  $\Omega$  to 250  $\Omega$  at 10 GHz. Here, the reference impedance of the first port is 250  $\Omega$ , and S<sub>11</sub> is defined with respect to this port. The insertion loss of the IT for this transformation case is -1.2 dB. The loss performance of the IT is also given for this impedance state in Figure 3.7-(c), where the loss of the IT is defined as:

$$L = 1 - |S_{11}|^2 - |S_{21}|^2$$
(3.7)



Figure 3.7: The performance of the triple stub IT for  $50 \Omega$ -to- $250 \Omega$  impedance transformation at 10 GHz (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c) The loss of the tuner at this state.



Figure 3.8: The performance of the triple stub IT for  $50 \Omega$ -to- $20 \Omega$  impedance transformation at 20 GHz (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c) The loss of the tuner at this state.

Another example is presented in Figure 3.8, where the IT is adjusted to transform 50  $\Omega$  to 20  $\Omega$  at 20 GHz. The insertion loss in this case is again -1.2 dB. The stub state numbers in the figures of the decimal equivalent of the states of the switches in each stub. For example, Stb1 = 32 means Stb1 = 0000100000, which means that the sixth switch on stub1 is in the downstate.

### 3.3. The Fabrication and the Measurement Results

The triple stub RF MEMS impedance tuner is implemented using the METU RF MEMS fabrication process, which was explained in Chapter 2 in detail. 350 µm-thick, 4" quartz substrates were used in the fabrication. The tangent loss of the substrate is  $tan \delta$  = 0.001 in all the band of interest. The photographs and the SEM pictures of the fabricated RF MEMS IT are given in Figure 3.9 and Figure 3.10, respectively.

#### 3.3.1. The Packaging of the Impedance Tuner

The RF MEMS impedance tuner needs to be packaged before they are measured as there are many DC connections for the biasing of the RF MEMS switches. For this purpose, ceramic IC packages are used. The area of the triple stub IT measures  $8.6 \times 8.6 \text{ mm}^2$ , so it was hard to find a package with this window size. Two different packages are selected for the IT from Spectrum-Semi Company, which are 24-pin, alumina based package (HYB02413) with a window of 10.9 x 26.7 mm<sup>2</sup> and 40-pin, ceramic based package (CSB04046) with a window of  $8.9 \times 11.4 \text{ mm}^2$ . The IT is first fastened on the package by using white epoxy, and then the pads are connected to the package by aluminum wire bonds. Figure 3.11 shows the photographs of the packaged RF MEMS IT, which is placed on the daughter board for the measurements. The bonding diagram of the IT for HYB02413 alumina package is given in Figure 3.12.



Figure 3.9: The photographs of the fabricated IT.



(a) Wide view of two of the stubs and the T-junction.



(b) The closer view of one of the stubs.



(c) The closer view of the MEMS switch.

Figure 3.10: The SEM pictures of the fabricated RF MEMS IT.



(a) The RF MEMS IT packaged in HYB02413 alumina IC package.



(b) The RF MEMS IT packaged in CSB04046 ceramic IC package and placed on the daughter board.





Figure 3.12: The bonding diagram of the RF MEMS IT for HYB02413 package.

# 3.3.2. The Design and the Implementation of the Multi-Pin MEMS Automated Measurement Setup (MAMS)

An automated measurement setup is needed for the measurements of the fabricated impedance tuner since there are 30 control pins, and all of the pins

should be controlled simultaneously. This is necessary for the measurement of all possible states of the impedance tuner, which has more than 1000 possible states. For this purpose, an automated measurement setup is designed and implemented. Figure 3.13 shows the block diagram of the setup. The setup consists of six main parts, which are;

- i) the network analyzer to measure the S-parameters (Agilent E8361A),
- ii) the probe station to contact the MEMS device (Summit 9000),
- iii) the MEMS biasing waveform generator card to produce the biasing waveform,
- iv) the biasing waveform distribution card to direct the biasing waveform to the MEMS switches on the correct time and order,
- v) the computer controlled DC supply to bias the distribution card (HP 3631A),
- vi) and the computer to control the network analyzer, DC supply, and save the data.

The most important parts of the setup are the "MEMS Biasing Waveform Generator Card<sup>1</sup>" and the "Biasing Distribution System<sup>2</sup>". The generator card is used to produce the waveform that is applied to the MEMS switches as the actuation voltage. With this generator card, it is possible to produce a number of different waveforms, which can be seen in Figure 3.14. For the all of the waveforms, the important parameter is not the voltage levels or the frequency of the waveform, but it is the transition time between the states. The transition time between any states of the waveform must be short enough that the RF MEMS switches are around a few microseconds, a transition time that is less than 1  $\mu$ s should be enough.

<sup>&</sup>lt;sup>1</sup> The MEMS Biasing Waveform Generator Card is designed and the implemented by Halil İbrahim Atasoy.

<sup>&</sup>lt;sup>2</sup> The Biasing Distribution System is designed and the implemented by Halil İbrahim Atasoy, the author has accompanied him during most of the steps.



Figure 3.13: The block diagram of the multi-pin MEMS automated measurement setup.

The generator card is designed to produce different types of waveforms, which are unipolar, bilevel unipolar, bipolar, and bilevel bipolar waveforms. There are two types of bipolar waveforms for the actuation of the RF MEMS switches. In the first one, the actuation voltage is applied in different polarities in two consecutive actuations. In the second one, the actuation voltage is applied in positive and negative polarities and switching between them very quickly so that the switch cannot respond in a single actuation. By this method, the dielectric charging can be reduced. The bilevel bipolar waveform has two levels, which are the actuation and hold down voltages. In this waveform, first, the actuation voltage is applied in both polarities and the switch is pulled down, and then, the hold down voltage is applied in both polarities continuously to hold the switch in downstate. Since the hold down voltage is less than the actuation voltage, the dielectric charging can be reduced further.



Figure 3.14: Some of the possible waveforms that can be produced by the "Waveform Generator Card".

Figure 3.15 shows the generator card photograph and the card block diagram. The card consists of 8 high voltage and high speed transistors (IRF 630), standard (IR 2113) and optocoupled (not shown in the figure) gate drivers for biasing the transistors, and a microcontroller (PIC 16F877) that organizes the time and duration of on/off states of the transistors. Figure 3.16 demonstrates the schematic of the generator card. There are 8 high voltage transistors in the circuit, which are grouped in 4, and they are driven by 4 gate driver ICs. The color coding of the transistors in Figure 3.16 shows which gate driver are connected to which transistor. The red transistors in the figure are driven by optocoupled gate drivers because of practical reasons.



Figure 3.15: The photograph of the waveform generator card and its block diagram.



Figure 3.16: The basic schematic of the waveform generator card.

The generator card works differentially by between the outputs of the two arms, which are composed of 4 transistors. The card operates as follows:

- State 1: Tr1 and Tr2 are open on V<sub>1</sub> side so that V<sub>1</sub> = V<sub>H</sub> + V<sub>L</sub>. Tr7 and Tr8 are open on V<sub>2</sub> side so that V<sub>2</sub> = 0. Since the output is taken differentially, V<sub>out</sub> = V<sub>H</sub> + V<sub>L</sub>.
- State 2: Tr5 and Tr6 are open on V<sub>2</sub> side so that V<sub>2</sub> = V<sub>H</sub> + V<sub>L</sub>. Tr3 and Tr4 are open on V<sub>1</sub> side so that V<sub>1</sub> = 0. Since the output is taken differentially, V<sub>out</sub> = -(V<sub>H</sub> + V<sub>L</sub>).
- State 3: Tr2 and Tr3 are open on V<sub>1</sub> side so that V<sub>1</sub> = V<sub>L</sub>. Tr7 and Tr8 are open on V<sub>2</sub> side so that V<sub>2</sub> = 0. Since the output is taken differentially, V<sub>out</sub> = V<sub>L</sub>.
- State 3: Tr6 and Tr7 are open on V<sub>2</sub> side so that V<sub>2</sub> = V<sub>L</sub>. Tr3 and Tr4 are open on V<sub>2</sub> side so that V<sub>1</sub> = 0. Since the output is taken differentially, V<sub>out</sub> = -V<sub>L</sub>.

The biasing waveform distribution system is designed to distribute the waveform produced by the generator card to the MEMS switches. The design and implementation of the distribution system has been accomplished by Halil Ibrahim Atasoy and the author has accompanied him during most of the steps. The distribution system is designed as two types of cards, which are the microcontroller and the relay driver cards, so that they can be used modularly. Figure 3.17 illustrates the two cards. The modular approach is feasible for this case since the number of relay driver cards depends upon how many MEMS switches we will drive. The microcontroller card can control 8 relay driver cards where each driver card can control 8 MEMS switches simultaneously. During the measurement of the RF MEMS IT, three or four distribution cards are used since there are a total of 30 MEMS switches to be controlled.





Figure 3.17: The waveform distribution system cards photograph and the block diagram. (a) Microcontroller card. (b) The relay driver card.

The microcontroller card is composed of a microcontroller (PIC16F877) and two buffers (74HC367). The microcontroller is triggered by the Agilent E3631A DC supply, which is controlled by the HPVEE program on the computer using GPIB interface. The microcontroller produces the 8-bit control data to control the relays and 3-bit address data to select the relay control cards. Each relay control card is composed of a latch (74HC574) to hold the control data coming from the microcontroller, a demultiplexer (74HC138) to convert the address data, a dipswitch to set the card address, and 8 relays (V23026) to switch the biasing waveform. The relays are used as single pole double throw (SPDT) switches, which select either one

or the other terminal of the generator card output. But since one of the terminals is connected to the ground of the relay control card, it behaves as the ground. The photographs of the all cards attached and the whole measurement setup can be seen in Figure 3.18.



Figure 3.18: The photograph of the multi-pin MEMS automated measurement setup.

#### 3.3.3. The Measurement Results of the RF MEMS Impedance Tuner

The MAMS is used for the measurements of the RF MEMS IT. The IT is measured in 1-20 GHz band using SOLT calibration technique, and Cascade ACP-GSG-150 coplanar probes are used during the measurements.

The IT is mounted on 24-pin, alumina based HYB02413 package instead of 40-pin, ceramic based CSB04046 package although the former has not enough number of pins. This is because some problems occurred during the measurements of the IT sample mounted on 40-pin, ceramic based CSB04046 package. Since HYB02413 package has 24 pins and there are no alumina based packages with a higher number of pins, 30 control pins of the IT are connected to the 24 pins of this package. This is done by shorting the fourth and fifth, and eighth and ninth pins of each stub, which reduces the total number of control pins of the IT to 24. The pins to be shorted are selected to obtain minimum reduction of the impedance coverage of the IT on the Smith Chart. As a result, a total of  $(8 + 1)^3 = 729$  impedance states of the IT are measured. 1-20 GHz band is measured in 401-points using SOLT calibration since increasing the number of points also increase the data transfer time to the PC, and the total measurement time increase significantly. This situation affects the measurement reliability since the RF MEMS switches start to be affected from the dielectric charging as the total downstate time of the switches increase.

Figure 3.19 shows the measured upstate performance of the fabricated RF MEMS IT compared with the simulations. Figure 3.20 and Figure 3.21 show the measured performance of the RF MEMS IT for same sample impedance states compared with the simulations. The figures show that the measurements are in agreement with the simulations. The values of the circuit model parameters that are extracted from the measurements are given in Table 3.4. Most of these parameters simulations are the close to the ones given in Table 3.2. However, the upstate capacitances are increased significantly. This is because of the decrease in the bridge heights of the MEMS switches. Although the bridge is almost flat in between the anchors, they do not follow the changes on the substrate, and the bridge heights do not increase at the areas where the first metal layer is present. As a results, the height between the first metal layer and the MEMS bridge decrease significantly, which results the increase in the upstate capacitances. This behavior was explained in more detail in Chapter 5.





Figure 3.19: The measured upstate performance of the RF MEMS IT compared with the simulations (Stb1 = 0, Stb2 = 0, Stb3 =0) (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c)  $\angle S_{21}$  (d)  $\angle S_{11}$ .

Table 3.4: The circuit model parameters that are extracted from the measu	irements of
the RF MEMS IT.	

Circuit Model Parameters Extracted from Measurements				
$Z_o$	97.5 Ω	$C_{sbd}$	1.085 pF	
$\mathcal{E}_{r,eff}$	2.45	$C_{bgd}$	4.46 pF	
α	33 dB/m @20 GHz	$L_b$	10 pH	
S	550 μm	$R_{bu}/R_{bd}$	0.5 / 0.5 Ω	
$C_{sbu}$	193.9 fF	$C_U/C_D$	155.7 fF/ 872.9 fF	
$C_{bgu}$	791.1 fF	<i>R</i> <sub>bias</sub>	4000 Ω	
Interconnection line parameters				
$Z_0 = 51.2 \Omega$ , $\mathcal{E}_{r,eff} = 2.44$ , $\alpha = 71 \text{ dB/m}$ , $l = 994 \text{ µm}$				




Figure 3.20: The measured performance of the RF MEMS IT compared with the simulations (Stb1 = 4, Stb2 = 2, Stb3 = 32) (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c)  $\angle S_{21}$  (d)  $\angle S_{11}$ .





Figure 3.21: The measured performance of the RF MEMS IT compared with the simulations (Stb1 = 64, Stb2 = 384, Stb3 = 24) (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c)  $\angle S_{21}$  (d)  $\angle S_{11}$ .

Figure 3.22 shows the impedance coverage of the fabricated RF MEMS IT. The left-hand side figures show the measurement results for  $9^3 = 729$  impedance states, and the right-hand side ones show the corresponding simulations of those states. These figures show that the simulated Smith Chart coverage is almost achieved from the measurements. It should be mentioned here that the circuit model parameters given in Table 3.4 are used for the simulations, which account the changes in the upstate capacitances.  $R_{bu}$ ,  $R_{bd}$ , and  $L_b$  are also differed from their values given in Table 3.2. These modified values are used in the simulations since

the simulations using these modified values give a better Smith Chart coverage when compared with the measurements. The change in  $L_b$  is because of the modified performance of the T-junctions. Since the MEMS bridge heights are changed, the performance of the T-junction, which is directly effective and optimized for the best the Smith Chart coverage, are also affected. The change in  $L_b$  compensates the consequences of the change in the performance of the T-junctions. The discussion about the changes in  $R_{bu}$  and  $R_{bd}$ , will be given in detail in Chapter 5.3.







Figure 3.22: The measured and simulated Smith Chart coverage of the RF MEMS IT in 6-20 GHz band.

Figure 3.22 shows the "exact" points that can be covered with the fabricated IT. However, -10 dB return loss is acceptable for many applications. So, considering a impedance mismatch level of 10 dB, the one-to-one correspondence between each measured-simulated impedance point pair is investigated. Table 3.5 summarizes the criteria for the comparison and the number of points that satisfy each criterion.

# of points	10 GHz	20 GHz	Amp. Margin	Phase Margin
$n_{10dB}$	267	182	r <sub>10dB</sub>	$\phi_{10dB}$
$n_{\phi 1}$	286	329	r <sub>10dB</sub>	20°
n <sub>ø2</sub>	345	363	$r_{10dB}$	30°
n <sub>ø3</sub>	402	412	r <sub>10dB</sub>	50°
n <sub>amp</sub>	553	472	r <sub>10dB</sub>	NA

Table 3.5: The summary one-to-one correspondence between the measured and the simulated impedance point of the IT.

Radius of 10 dB constant impedance mismatch circle (variable for each impedance point)

 $\begin{array}{c} \hline \phi_{10dB} \\ \hline \phi_{10dB} \\ \hline mpedance mismatch circle (variable for each impedance point) \\ \hline n_{10dB} \\ \hline m_{10dB}$ 

# of simulation points that has  $|\Gamma_{sim}| = |\Gamma_{meas} \pm r_{10dB}|$  and maximum phase deviation of  $\phi_i$  degrees

 $n_{amp}$  # of simulation points that has  $|\Gamma_{sim}| = |\Gamma_{meas} \pm r_{10dB}|$ 

For the comparison of the measured and simulated point pairs, different criteria are used. The first one is the "constant impedance mismatch circle (CIMC)", which is used in the design of two-port network that is not matched at both of its ports. The CIMC determines the points that will have 10 dB return loss when connected to the original points. These points fall onto a circle for each impedance point on the Smith Chart, as its name already implies. The center,  $\Gamma_{SM}$ , and the radius,  $r_{SM}$ , of the CIMC can be calculated as in [172] using (3.8) and (3.9), respectively, where (3.10)-(3.11) give the impedance mismatch factor,  $M_1$ . (3.10) is the general form of  $M_1$ , and (3.11) is the reduced form that is used to calculate  $\Gamma_{SM}$  and  $r_{SM}$ .

$$\Gamma_{SM} = \frac{M_1(\Gamma_{in})^*}{1 - (1 - M_1)|\Gamma_{in}|^2}$$
(3.8)

$$\mathbf{r}_{SM} = \frac{\sqrt{1 - M_1} (1 - |\Gamma_{in}|^2)}{1 - (1 - M_1) |\Gamma_{in}|^2}$$
(3.9)

$$M_{1} = \frac{(1 - |\Gamma_{in}|^{2})(1 - |\Gamma_{s}|^{2})}{|1 - \Gamma_{in}\Gamma_{s}|^{2}}$$
(3.10)

$$M_{1} = 1 - \rho^{2} \text{ and } \rho = \frac{VSWR_{req} - 1}{VSWR_{req} + 1}$$
(3.11)

Table 3.5 shows that only 267 and 182 out of 729 simulation points fall into CIMC of their corresponding measured impedance points. However, this is mostly due to the phase errors occur in the phase of  $S_{11}$ . The phase of  $S_{11}$  generally changes rapidly in the triple stub IT topology, which cause the target impedance point to shift. This behavior can also be observed from Figure 3.19 to Figure 3.21. To prove this, the phase margin is loosed to three constant levels, which are namely 20°, 30°, and 50°, while keeping the amplitude margin as  $r_{10dB}$ . In this case, it is observed that the number of points that fall into to the margin of its corresponding point increases. Finally, if only the amplitude margin is used, which is again  $r_{10dB}$ , and the phase margin is removed, 553 and 472 out of 729 simulation points fall into the margin of its corresponding measured impedance point.

One point should also be mentioned before concluding the impedance coverage discussion, which is related with the limitations of the measurement setup. Since 729 impedance points have been measured continuously, some of the RF MEMS switches stay in downstate for a "long time". Long time here refers the time that the switch performance starts degrading due to the dielectric charging. The phase response of the tuner shifts some degrees even the performance of a single RF MEMS switch drifts from its original upstate response. One proof of this statement is that the IT upstate response is "shifted" when it is measured after the measurement of 729 states. The "new" upstate response can be modeled by increasing the upstate capacitance, which proves the statement. As a result, it is expected that the IT may have better performance in terms of number of matching impedance pairs once it is measured considering this situation.

Finally, if the 10 dB impedance coverage of the all 729 measured impedance points is investigated, it is seen that this range covers almost 100% of the simulated impedance points. This means that the targeted impedance range that is defined by the simulations is covered by the measured impedance points with a worst case impedance mismatch of 10 dB. Figure 3.23 shows the mentioned comparison at two sample frequencies, which are namely 10 GHz and 20 GHz. The red regions show the area that is covered by the 10 dB circles of the measured 729 impedance points, and the blue points show the 729 simulated impedance points. In this figure, the conjugates of the simulated and measured points are used due to the definition of impedance mismatch.



Figure 3.23: The 10 dB impedance coverage of the measured impedance points that can be matched by the fabricated RF MEMS IT (a) 10 GHz (b) 20 GHz. The impedance points in this figure are the conjugates of the ones given in Figure 3.22.

Finally, some example impedance transformations using the fabricated IT are presented. Figure 3.24 the measured performance of the IT when it is adjusted to transform 50  $\Omega$  to 5  $\Omega$  at 10 GHz compared with the simulations. Another example is presented in Figure 3.25 where the measured performance of the IT is given

compared with the simulations. In the figure, the IT is adjusted to transform 50  $\Omega$  to 150  $\Omega$  at 20 GHz, and the losses are calculated using (3.7).

The measured insertion losses are about -3.3 dB for both cases. Although the loss behavior is similar to the simulations that use the extracted circuit model parameters, the measured losses are higher than that of the original design values. The reason for this case is the reduced MEMS bridge height. The reduced height increases the upstate capacitances, and the increase in the loss is actually expected as can be seen from the formula of the calculation of the loss of a DMTL. The formula is given in (3.12), and it is expected from the formula that the losses will increase as the upstate capacitance increases. Since the IT structure is composed of 30 MEMS switches that are as used as unit sections, and only 3 of them are hold in downstate at each impedance state, the amount of the loss increase is reasonable.

$$\alpha_{ld} = \frac{R_t}{2Z_{ld}} + \frac{R_b Z_{ld} \omega^2 C_U^2}{2} dB / unit section$$
(3.12)

If the loss is calculated using the extracted circuit model parameters, the loss per unit section increases from 0.044 dB/unit section to 0.078 dB/unit section. This is obtained by only increasing the upstate capacitance from the original simulation values to the extracted values. So, this much change per unit section (switch) in the upstate also supports the above reasoning.



Figure 3.24: The measured performance of the triple stub IT for 50  $\Omega$ -to-5  $\Omega$  impedance transformation at 10 GHz compared with the simulations (a)  $|S_{11}|$  (b)  $|S_{21}|$ (c) The loss of the tuner at this state.



Figure 3.25: The measured performance of the triple stub IT for 50  $\Omega$ -to-150  $\Omega$  impedance transformation at 20 GHz compared with the simulations (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c) The loss of the tuner at this state.

## 3.4. Conclusion

In this chapter, the design and the implementation of a reconfigurable RF MEMS triple stub impedance tuner is presented for 6-20 GHz applications. The aim is to design and implement a reconfigurable impedance tuner that covers as much area as possible on the Smith Chart in this frequency band. A multi-pin MEMS automated measurement setup is also designed and implemented for the measurements of the designed RF MEMS IT. The measurement results of the IT show good agreement with the simulations. Although the fabricated device suffers from the reduced MEMS bridge height, it still shows a good Smith Chart coverage within the band of interest. One-to-one correspondence of the measured and the simulated impedance points is also investigated, and it is observed that the phase error of S<sub>11</sub> mainly determines the deviations in the one-to-one correspondence. In any case, it is shown that the measured impedance points almost totally cover the area that includes all of the targeted impedance points with 10 dB impedance mismatch.

# **CHAPTER 4**

# THE THEORY OF PHASE AND AMPLITUDE CONTROL USING TRIPLE STUB TOPOLOGY

## 4.1. Introduction

The impedance matching using the stub based networks is a very well known and frequently applied way of solving impedance mismatch problems [172], [173]. Single, double, and triple stub impedance matching networks are the most commonly used topologies, and the easiest way to implement these networks is to use transmission lines. The triple stub topology is actually an extension of the double stub topology. The third stub is generally used to "move" the load impedance point out of the "forbidden region" of the double stub topology on the Smith Chart [172]. The triple stub topology comes up with infinitely many solutions for the lengths of the three stubs. Some of these solutions can be selected so that some other parameters other than the reflection coefficient can be tuned while keeping the input impedance unchanged.

This chapter shows that the triple stub (TST) topology can be used to control the insertion phase and the amplitude of the input signal simultaneously while the input impedance matching is preserved, using low-loss transmission lines. The TST is arranged to make a 50  $\Omega$ -to-50  $\Omega$  impedance transformation, which is actually an all pass network, and the lengths of the three stubs are selected so that the insertion phase and the insertion loss can be tuned from 0° to 360° and -0.8 dB to more than -20 dB, respectively. The control mechanisms for the insertion phase

and the insertion loss are investigated and identified. The idea is also verified experimentally using coplanar waveguides (CPW) on quartz substrates.

## 4.2. The Theory of Phase Control Using Triple Stub Topology

## 4.2.1. Lossless Case

The theory of phase control using triple stub IMN is based on the infinitely many solutions for  $Z_o$ -to- $Z_o$  transformation. This all-pass network may be used to control the phase of output voltage signal as an all-matched phase shifter.

In an ideal phase shifter, the phase of  $S_{21}$  is controlled while its amplitude is kept at 1. This means that all the power is transferred to the output and its  $S_{11}$  is equal to 0. So, there are two complex equations and three real unknowns, which are the susceptances of the three stubs. The  $S_{21}$  equation can be separated into its real and imaginary parts, which actually means that there are four real equations and three real unknowns, and the system of equations has infinitely many solutions. The  $S_{21}$ and  $S_{11}$  of a two-port network can be represented in terms of its ABCD parameters as in (4.1) and (4.2). In (4.1), *a* and *b* are used to represent the real and imaginary parts of a unity magnitude complex number with arbitrary phase.

$$S_{21} = \frac{2}{A+B+C+D} = \frac{a+jb}{\sqrt{a^2+b^2}}$$
(4.1)

$$S_{11} = \frac{A+B-C-D}{A+B+C+D} = 0$$
(4.2)

(4.1) and (4.2) can be reduced to

$$A+B = \frac{a-jb}{\sqrt{a^2+b^2}} \tag{4.3}$$

$$A + B = C + D \tag{4.4}$$

which are the final form of the two complex equations.

The schematic of the TST is given in Figure 4.1. The network consists of lossless transmission lines with characteristic impedance  $Z_o = 50 \Omega$ , relative dielectric constant  $\varepsilon_{r,eff} = 1$ , and lengths x, y, z, and t. X, Y, and Z stand for the normalized susceptances of the three stubs. During the calculation normalized values will be used; therefore,  $Z_o$  is equal to 1. The ABCD parameters of the TST are found using (4.5) and (4.6).

$$\begin{bmatrix} A & B \\ C & D \end{bmatrix}_{TST, lossless} = \begin{bmatrix} 1 & 0 \\ jX & 1 \end{bmatrix} \begin{bmatrix} U & jT \\ jT & U \end{bmatrix} \begin{bmatrix} 1 & 0 \\ jY & 1 \end{bmatrix} \begin{bmatrix} U & jT \\ jT & U \end{bmatrix} \begin{bmatrix} 1 & 0 \\ jZ & 1 \end{bmatrix}$$
(4.5)

$$X = \tan(\beta x), Y = \tan(\beta y), Z = \tan(\beta z), T = \sin(\beta t), U = \cos(\beta t)$$
(4.6)



Figure 4.1: The schematic of the triple stub topology that is used for phase and amplitude control.

For the solution of the stub susceptances, (4.3) is separated into its real and imaginary parts, and solved for *Y* and *Z*, which results with:

$$Y = \frac{1}{T^2} \frac{b + 2TU\sqrt{a^2 + b^2}}{\sqrt{a^2 + b^2}}$$
(4.7)

$$Z = \frac{1}{bT} \left( T \sqrt{a^2 + b^2} + aT + bU \right)$$
(4.8)

(4.4) can also be separated into its real and imaginary parts. Inserting (4.7) and (4.8) into the real part of (4.4), the conservation of power is verified, which is expected. Inserting (4.7) and (4.8) into the imaginary part of (4.4), X is found as:

$$X = \frac{1}{abT} \Big( T^3 (2 - T^2) (a^2 + b^2) - b^2 T + aT \sqrt{a^2 + b^2} + abU + TU^4 (a^2 + b^2) \Big)$$
(4.9)

The equations (4.7)-(4.9) are the solutions for arbitrary interconnection line lengths and they are functions of only a, b, T and U. The lengths of the stubs are found using the inverse of the tangent function as:

$$x = \frac{\lambda}{2\pi} \arctan(X), y = \frac{\lambda}{2\pi} \arctan(Y), z = \frac{\lambda}{2\pi} \arctan(Z)$$
(4.10)

The analytical solution presented for the lossless case is for the phase shifter solution with  $Z_o$ -to- $Z_o$  transformation, which was intended to preserve the input impedance matching. However, there might be some cases where the input impedance is not required as to be  $Z_o$ , but rather,  $kZ_o$ . The analytical solution for this case is also investigated. By changing the conditions case given in (4.1) and (4.2) as

$$S_{21} = \frac{2}{A+B+C+D} = \frac{a+jb}{\sqrt{k}\sqrt{a^2+b^2}}, k \in \mathbb{R} \text{ and } k > 1$$
(4.11)

$$S_{11} = \frac{A+B-C-D}{A+B+C+D} = 0$$
(4.12)

and following the same solution method, the solution is found for the  $Z_o$ -to- $kZ_o$  transformation case where k is a real number. The solution for the susceptances in this case is given in (4.13)-(4.15).

$$Y = \frac{1}{\sqrt{k}T^2} \frac{b + 2\sqrt{k}TU\sqrt{a^2 + b^2}}{\sqrt{a^2 + b^2}}$$
(4.13)

$$Z = \frac{1}{bT} \left( \sqrt{kT} \sqrt{a^2 + b^2} + aT + bU \right)$$
(4.14)

$$X = \frac{1}{abT} \left( kT^{3}(2 - T^{2})(a^{2} + b^{2}) - b^{2}T + \sqrt{k}aT\sqrt{a^{2} + b^{2}} + abU \right) + \frac{1}{abT} \left( kaTU^{4}(a^{2} + b^{2}) \right)$$
(4.15)

# 4.2.2. Lossy Case: Example Case Study: $t = \lambda/8$

To verify the formulation, a special case for an interconnection line length of  $t = \lambda/8$ , which is the most commonly used interconnection line length for TSTs, is studied. This TST is simulated in MWO for the whole phase range of 0-360° with 1° steps, where the lengths of the stubs of the IMN are calculated using (4.7)-(4.10), and the interconnection line lengths are fixed as  $\lambda/8$ . The simulation frequency is selected as 15 GHz, and transmission lines with  $\varepsilon_{r,eff} = 1$  are used in the simulations. Figure 4.2-(a) shows that the formulation works well for 0-360° for the ideal (lossless) case (black line with circles in Figure 4.2-(a)). Figure 4.2-(b) verifies that the input matching is preserved since all the  $S_{11}$  point lie at the origin of the Smith Chart.



Figure 4.2: The insertion phase, the insertion loss, and the return loss performances of the TST based phase shifter for different transmission line losses (a)  $S_{21}$  (b)  $S_{11}$ .

After verifying that the formulation is valid for the lossless case, the TST is simulated by adding a non-zero loss factor to the transmission lines and without changing any other parameters, i.e., keeping the stub and the interconnection line lengths the same. The simulation results for  $\alpha = 20$  dB/m and 50 dB/m are also given in Figure 4.2. Examining the simulation results in the figure, it is seen that the formulation starts to deviate from the expected insertion phase and input return loss performance. Considering a worst case -3 dB range in the S<sub>21</sub> magnitude, the phase shifter works for the 42-337° range for  $\alpha = 20$  dB/m, and for the 86-326° range for  $\alpha = 50$  dB/m. Out of these ranges, the phase shifter suffers from the increasing insertion loss, and the corresponding return losses also become worse than about -12 dB.

The above analysis show that the TST based phase shifter has a limited insertion phase range for a given value of the interconnection line, *t*. This is true for any value of *t*; but, the insertion phase range shifts as *t* changes. In other words, phase

shifter has limited insertion phase range for a specific value of t; however, whole 0-360° range can be covered by changing this interconnection line length. Since the solution exists for the stub lengths for each value of t, we can find one t value, for which the required insertion phase can be obtained. Figure 4.3 shows the performance of the phase shifter for different interconnection line lengths. For the cases in the graphs, S<sub>11</sub> characteristics are also as good as  $t = \lambda/8$  case for the possible phase range of each case. The reason of the discontinuities in the traces in Figure 4.3 is that the lengths of the stubs were increased by  $\lambda/2$  in order to get rid of the negative line lengths coming from the mathematical formulation for some insertion phase values.



Figure 4.3: The performance of the TST based phase shifter for different interconnection line lengths (a)  $t = 0.0375\lambda$ ,  $t = \lambda/8$ ,  $t = \lambda/4$  (b)  $t = 3\lambda/8$ ,  $t = 0.4625\lambda$ .

The reason of the distortion in the insertion phase characteristics is understood better by examining the real and imaginary parts of the impedances of the stubs in detail. Figure 4.4 shows the real, R, and the imaginary, X, parts of the impedances

of the first and the third stubs, and their ratio, R/X for two sample interconnection line lengths. The first and the third stubs have the same length for all cases, which is a result of the analytical formulation. The imaginary parts of the impedance of the stubs get smaller as the desired insertion phase approaches to either 0° or 360°. Once the ratio of the real part to the imaginary part of the first and the third stub impedances increase significantly (0.05 as the insertion phase approaches 0° and 0.1 as the insertion phase approaches 360°; the difference between the ratios is the added  $\lambda/2$  line because of the negative line lengths as explained above), the required reactance to be realized by the stub is suppressed by its resistance. As a result, the desired impedance transformations are not achieved, and the insertion and the return losses increase. The insertion phase also starts to deviate from the desired values. The real part of the input impedance of the second stub also becomes effective at some range; however, the real parts of the impedances of the first and the third stubs are the ones that distort the performance first.





Figure 4.4: The real and imaginary parts of the input impedance of the first stub and their ratio

#### 4.3. The Phase / Amplitude Control Using the Triple Stub Topology

The results of the previous section show that 0-360° insertion phase range can be covered using the formulation for the lossless case and changing the length, t, of the interconnection line. It is observed on Figure 4.3 that the graphs are actually the shifted versions of each other, and there is always a low-loss region which can be tuned to the desired insertion phase value by changing t. At the other regions where the insertion loss increases significantly, the insertion phase and the input return loss is preserved up to some point. This result came up with the fact that the amplitude of the output signal can also be controlled.

It should be pointed out here that the aim is to achieve the above mentioned goal by only using low-loss transmission lines. The transmission lines used in the frame of this work have 50 dB/m loss at 20 GHz with characteristic impedance,  $Z_o = 50 \Omega$ , and relative dielectric constant,  $\varepsilon_{r,eff} = 1$ . The  $R/\omega L$  is equal to 0.033 for these transmission lines at 15 GHz, where R and L are the per unit length parameters of the transmission line. This means that the transmission lines that are used in this work are low-loss transmission lines. Several analytical solution methods are utilized to find the solution with lossy transmission lines; but, the problem results with four complex, transcendental equations that include complex hyperbolic functions. Instead, it is shown that the problem has solutions using two methods, (i) optimization and (ii) direct search of the whole solution space. In the first method, the Simplex optimization tool of the MWO is used, in which x, y, z, and t, are used as optimization variables. The optimization goals are setting:

- i)  $\angle S_{21}$  to any desired insertion phase value,
- ii) |S<sub>21</sub>| = 1, and
- iii)  $|S_{11}| = 0$ .

Using this method, the solutions are found for any magnitude value of the  $S_{21}$  from -0.8 dB to -20 dB and for any phase value of the  $S_{21}$  from 0° to 360° where  $|S_{11}|$  is less than -250 dB for all cases. The solution took at most 30 s, and the cost of the optimization is always less than  $10^{-26}$ . -20 dB for the  $S_{21}$  magnitude is not actually a limit; however, the insertion phase and the input return loss deviates from the desired values as the  $S_{21}$  magnitude becomes less than this value.

The second method is to sweep all the combinations of the line lengths from 0 to  $\lambda/2$  with the smallest step size possible. A Matlab code has been developed for this purpose which can sweep the line length variables with  $\lambda/80$  step size. The code can calculate  $40^4 = 2.56M$  combinations by searching the solution in 0 to  $\lambda/2$  range for each variable in 300 seconds using multidimensional matrix operations supported by Matlab. Please note that a direct calculation of  $12^4 = 20736$  combinations take about the same time, and the total simulation time grows exponentially as the step size decreases linearly. Using the developed code, it is observed that the same solutions can be found with the optimization method for any desired value of the phase and magnitude of the S<sub>21</sub> where  $|S_{11}|$  is less

than -30 dB for all cases. The solutions found using this method, of course, has some errors because of the fact that the lengths are discreet variables.

The results of the two methods show that the TST can be used to control the insertion phase and the amplitude of the input signal at once while the input kept as exactly matched. This means that the TST can be used as a "vector modulator" circuit.

The frequency response of the TST is presented in Figure 4.5, where the vector modulator is set to different insertion phase/amplitude states. The circuit is naturally a narrowband circuit as it is based on transmission lines. However, the instantaneous bandwidth of the TST is large enough when the TST is used as a phase shifter. The bandwidth (BW) for -1 dB/45°, as an example, is about 3 GHz considering flat insertion loss (±0.2 dB for -1 dB) and acceptable return loss characteristics (better than -15 dB), and linear phase response. 2.6 GHz BW is equivalent to 17.3% bandwidth at 15 GHz, and it is limited by the return loss performance. The BW of the TST gets narrower as the required amplitude decreases. The BW for -7 dB/45° is 260 MHz (1.73%), and it is limited by return loss response (better than -15 dB). The amplitude remains within ±1 dB range in the BW determined by the return loss RW, and gives flat response when taken differentially with respect to the 0° state.





Figure 4.5: The performance of the TST when it is set to three sample insertion phase/amplitude states at 15 GHz (a)  $|S_{21}|$  (b)  $\angle S_{21}$  (c)  $|S_{11}|$  (d) The loss of the TST.

#### 4.3.1. Investigation of the Loss Behavior

It is shown in the previous subsections that the TST can be used to control the amplitude of the input signal. It is possible to obtain amplitude levels as low as 0.1, which is equivalent to -20 dB insertion loss. Since the return loss is approximately 0 for all cases, having such small amplitudes means that most of the input power is attenuated in the circuit. It is interesting to have such a case since the circuit only consists of low-loss transmission lines.

The physical reasoning behind the losses, which can be defined as in (3.7) for a two-port network, is the resonating behavior of the TST. If the circuit is examined in detail, it is seen that there are three open-circuited terminations at the ends of the three stubs. The paths between any two of these open circuited ends behave as a  $\lambda/2$  resonator, and the insertion loss of the TST increases significantly at any of the frequencies where the total length between the open-circuited terminations, i.e., x + y + t, y + z + t, and x + z + 2t, is equal to  $n\lambda/2$ . The resonators can be visualized in Figure 4.6.

One example of the above mentioned loss behavior can be seen Figure 4.5-(d). The loss of the TST that was set to (-7 dB, 45°) has peaks at 8.06 GHz and 15.08 GHz. The x + y + t, y + z + t, and x + z + 2t totals are equal to 20042 µm, 20646 µm, 31415 µm, which are  $\lambda/2$  at 7.48 GHz, 7.26 GHz, and 4.77 GHz, respectively. So, the peaks are due to the first resonator, and the loss peaks occur at frequencies that are very close to the first (7.48 GHz) and second (14.96 GHz) multiple of the  $\lambda/2$  resonances. The difference between the simulated frequencies of the resonance points and the natural resonance frequencies of the  $\lambda/2$  resonators is due to the fact that the real parts of the lossy transmission line distort the impedance transformation in and between the stubs.





Figure 4.6: The open-circuited resonators of the TST.

Controlling the amplitude of the input signal using the  $\lambda_o/2$  resonators is explained by approaching the problem from the quality factor point of view. Let us consider the circuits given in Figure 4.7. The circuit (1) is a  $\lambda_o/2$  open-circuited transmission line resonator, which behaves like a shunt resonant circuit around  $f = f_o$ . The unloaded quality factor of circuit (1), Q, is given in (4.16) [173] as:

$$Q = \frac{\beta}{2\alpha} = \frac{R}{\omega_o L} \tag{4.16}$$

where  $\alpha$  and  $\beta$  are the real and imaginary parts of the propagation constant of the transmission line, and R and L are the parameters of the equivalent parallel resonant circuit. R and L are given in (4.17).

$$R = \frac{Z_o}{\alpha l}, L = \frac{2Z_o}{\pi \omega_o}, \text{ and } l = \frac{\lambda_o}{2}$$
(4.17)

Once this resonator is excited from any point between 0 and  $\lambda_o/2$  as in circuit (2), it is loaded by the  $Z_o$  of the port 1, which is 50  $\Omega$  for this case. As a result, Q remains the same where the loaded quality factor,  $Q_L$ , decreases significantly since the external quality factor,  $Q_e$ , is much lower than Q.  $Q_e$  and  $Q_L$  are given in (4.18) and (4.19), respectively.



Figure 4.7: The schematic of  $\lambda/2$  resonator with 0, 1, and 2 input/output ports.

$$Q_e = \frac{Z_o}{\omega_o L} \tag{4.18}$$

$$\frac{1}{Q_L} = \frac{1}{Q} + \frac{1}{Q_e}$$
(4.19)

The input admittance at port 1 is purely real because the imaginary parts of the admittances looking towards each direction cancel each other since  $t_1 + t_2 = \lambda_0/2$ . So, the input admittance is purely real, and by changing the position of the port within the resonator, the real part of the input admittance at port 1 can be set to  $Y_o = 1/Z_o$ . This situation is true for any value of transmission line attenuation constant,  $\alpha$ , which is greater than 0. This can be visualized in Figure 4.8. If  $t_1$  and  $t_2$  are selected around  $\lambda_0/4$ , there always exists a point where the sum of the real parts of the admittances looking towards each direction is equal to  $Y_o$ . The values of  $t_1$  and  $t_2$  that satisfy the condition for different values of  $\alpha$  are given in Table 4.1. This, as a result, means that the maximum power transfer to the circuit (2) is obtained.





Figure 4.8: The  $|S_{11}|$  and the input admittance of the resonator circuit (2) where the input admittance is equal to  $Y_o$ .

α	$t_1$	$t_2$
(dB/m)	(µm)	(µm)
0.01	4986	5007
0.1	4965	5028
1	4896	5097
10	4679	5314
50	4285	5708

Table 4.1: The lengths used in the circuit (2) where the input admittance is equal to  $Y_o$  for different values of transmission line attenuation constants.

If one more port is added to the circuit as in circuit (3), the power that will be output power from the circuit can be adjusted. This can be achieved by arranging the positions of the two ports within the  $\lambda_0/2$  resonator, and the S<sub>11</sub> magnitude at port 1 can still be kept approximately equal to zero. This means that the maximum power transfer is still possible provided that the position of the port 1 in circuit (3) will be different than that of in circuit (2) because of the loading of  $Z_0$  of the port 2. An example case for circuit (3) is presented in Figure 4.9, where  $t_3$ ,  $t_4$  and  $t_5$  are adjusted to have  $|S_{21}|$  to be -9 dB for different values of transmission line attenuation constant,  $\alpha$ . The values of  $t_3$ ,  $t_4$  and  $t_5$  that are used in the simulations given in Figure 4.9 are presented in Table 4.2.

The idea behind controlling the output power is changing the value of  $Q_e$ . Arranging the positions of the ports is actually changing  $Q_e$  of the circuit by changing the ratio of  $Y_o$  to the real part of the input admittances at the port points. As a result,  $Q_L$  can be adjusted, which means that the amount of the power that will be dissipated in the circuit can also be adjusted. The same behavior, that is controlling the amount of the output power while keeping S<sub>11</sub> magnitude equal to zero, can be obtained by changing the port impedances for given fixed positions of port 1 and port 2. In this case, it is clear that the  $Q_e$  of the circuit (3) is adjusted, and the output power control is obtained by changing  $Q_e$  of the circuit (3).

_	α	$t_3$	$t_4$	$t_5$
_	(dB/m)	(µm)	(µm)	(µm)
-	0.01	4986	4993	14.6
_	0.1	4963	4984	46
_	1	4889	4958	146
	10	4658	4876	459
_	50	4249	4730	1015

Table 4.2: The lengths of the parts of the circuit (3) for different values of transmission line attenuation constants.



Figure 4.9: The  $|S_{21}|$  and  $|S_{11}|$  of the resonator circuit (3).

#### 4.3.2. Design Methodology in the Presence of the Losses

The design of the phase/amplitude control using the TST requires the analytical solution of the previously solved problem with lossy transmission lines. But, the analytical solution of the problem does not give explicit equations as explained in the previous subsections. Therefore, a design procedure that is based on the formulation for the lossless case is developed.

Observations on the solutions of the problem with lossy transmission lines using optimization and search of the whole solution space can be summarized as follows:

- There is always an interconnection line length, t, that the analytical solution for the lossless case gives approximate results.
- ii) The lengths of the first and the third stubs were equal in the analytical solution of the lossless case. The lengths of these stubs are not equal but close to each other for the lossy case.
- iii) The second and the third stubs were enough to obtain the insertion phase/amplitude control, and the third stub was used to obtain the input matching in the lossless case. This is still true for the lossy case.
- iv) The loss performance of the TST shows peaks in the vicinity of the frequencies where the lengths of the open-circuited resonators of the TST are equal to  $n\lambda/2$ .

Based on the observations above, the following method is developed for the design of the insertion phase/amplitude controlling using the TST:

i) Find the initial points for the stub lengths for the desired insertion phase value using the formulation for the lossless case given in (4.7) to (4.10).

- ii) Tune the interconnection line length, *t*, so that x + y + t, i.e., total length of the one of the shorter open-circuited resonators, is equal to  $\lambda/2$  at the design frequency. y + z + t or x + z + 2t may also be used in some cases.
- iii) Tune the lengths of the first and the third stubs in opposite directions; more clearly, tune z as  $z \pm \Delta l$ , meanwhile, tune x as  $x \mp \Delta l$  as the S<sub>11</sub> magnitude gets as small as possible.

For the demonstration of the proposed method, a sample problem is presented. In the sample problem, S<sub>21</sub> is targeted as (-6 dB, -120°). The problem is solved first with the optimization method, and then, with the proposed method. The performance comparison of the optimization solution with the proposed solution method is given in Figure 4.10. The line lengths obtained for both solutions are given in Table 4.3. The figures show that the proposed method can approximately guess the solution. The insertion phase is very close to the target value, and the return loss is acceptable. However, the amplitude response has some error. There is a trade-off between the insertion phase and the amplitude errors that come up with this method, and the designer can minimize the error in either the insertion phase or the amplitude of the input signal.

	Opt. Soln.	Prop. Soln. Met.
x (μm)	4241	4305
y (μm)	5169	5227
z (μm)	4267	4329
t (μm)	674.4	605.7
Δl (μm)	-	12.1
Loss peak (GHz)	14.83	14.79

Table 4.3: The line lengths used in the TST circuit model that are obtained from the optimization solution and the proposed solution method.





Figure 4.10: The performance of the proposed solution method for the insertion phase/amplitude control using the TST for (-6 dB/120°) state in the presence of transmission line losses (a)  $|S_{21}|$  (b)  $\angle S_{21}$  (c)  $|S_{11}|$  (d) The loss of the TST.

## 4.4. Experimental Verification

## 4.4.1. The Design of the Single Metal Layer Triple Stub Topology Circuits

The experimental verification of the insertion phase/amplitude control using the TST concept is done by the implementation and the measurements of some sample TST based insertion phase/amplitude control circuits. The circuits include 14 sample insertion phase/amplitude states that are reconfigured as phase shifter, vector modulator, attenuator, and wider band design versions of these circuits.

For the design of the circuits, 50  $\Omega$  CPW lines have been used as the transmission lines. The dimensions of the CPW lines are  $G/W/G = 36 \,\mu\text{m}/378 \,\mu\text{m}/36 \,\mu\text{m}$ . The parameters of the CPWs are calculated using the formulation given in [172]. The CPW parameters that are used in the designs are  $Z_o = 50 \,\Omega$ ,  $\varepsilon_{r,eff} = 2.36$ ,  $\alpha = 60 \,\text{dB/m} @ 20 \,\text{GHz}$ . The formulation given in [172] calculated the attenuation constant as  $\alpha = 42 \,\text{dB/m} @ 20 \,\text{GHz}$ ; however,  $\alpha = 60 \,\text{dB/m} @ 20 \,\text{GHz}$  used in the designs depending on the previous measurement experience. The circuits are implemented using CPW lines on quartz substrate with 0.8  $\mu$ m-thick single metal layer, which is nothing but the first metal layer of the METU RF MEMS process that is detailed in Chapter 2.

T-junctions are used in each design at three points where the stubs are connected to the interconnection lines and the input/output ports. A special CPW T-junction is designed for this application. This special T-junction design is very important and definitely necessary for the proper operation of the designed circuits. If the T-junction design is not proper, the connection points would not behave as sections of transmission lines, which results with distortion from the expected performance of the circuit.

Figure 4.11 shows the layout of the T-junction, which is designed and simulated in HFSS v9.2. In the design, the grounds of the CPW lines are connected by wirebonds because the circuits are implemented using a single metal layer process, and it is not possible to use air-bridges in this process. The wirebond are modeled with a metallic strip that has 25  $\mu$ m × 25  $\mu$ m cross section area, which are used to simulate aluminum wirebonds with 25  $\mu$ m diameter. The details about the design and the modeling of CPW T-junctions will be given in Chapter 5.2.1. The lengths of the transmission lines that are used to model the effects introduced by the T-junction are  $x = 498.8 \mu$ m and  $y = 229.8 \mu$ m, and the maximum error between the S-parameters of the circuits with and without T-junctions is 0.0003 in 1-20 GHz band.


Figure 4.11: The layout of the T-junction that is designed for the single metal layer TST based insertion phase/amplitude control circuit. The dimensions are given in  $\mu$ m.

The circuits have been designed using the optimization method where the four optimization variables are the lengths of the three stubs and the interconnection line. The design procedure was explained previously in this chapter. There are 14 designs that are used to demonstrate some sample states of phase shifter, vector modulator, attenuator, and wider band versions of some of these designs. Table 4.4 shows the target insertion phase/amplitude values, the simulation results, the lengths of the stubs and the interconnection line, and the measurement results for each design. Figure 4.12 shows the layout of (-1 dB, 90°) state as an example.



Figure 4.12: The layout of the single metal layer TST based insertion phase/amplitude control circuit, which is set for (-1 dB, 90°) state.

# 4.4.2. The Measurements of the Single Metal Layer Triple Stub Topology Circuits

The designed TSTs are fabricated on 4" quartz substrate with single metal layer process. The metallization used is 0.8  $\mu$ m-thick gold layer. The ground connections of the T-junctions of the TSTs are the aluminum wire bonds that are made with wedge bonding. The photographs of the fabricated single metal layer TST based phase/amplitude control circuits are given in Figure 4.13 for some example states.

Target		Simulation					Measurement			_			
Mag (dB)	Ang (°)	S11	S21		Stb1	Stb2 Stb3 Int. Line		Int. Line	S11	S21		Freq.	Design No
		Mag (dB)	Mag (dB)	Ang (°)	LA2 (um)	LB2 (um)	LC2 (um)	mL (um)	Mag (dB)	Mag (dB)	Ang (°)	(GHz)	
PHASE SHIFTER													
-1	0	-287	-1	0	398.4	5170	636.7	4928	-15.3	-1.6	0	14.44	1
-1	45	-281	-1	45	5738	281.3	5666	4348	-28.2	-1.2	45	14.79	2
-1	90	-280	-1	90	7192	944.1	7173	2520	-28.4	-1.1	90	14.6	3
-1	180	-281	-1	180	7413	1476	7836	602.7	-24.8	-0.9	-179.9	14.55	4
VECTOR MODULATOR													
-6	90	-275	-6	90	8431	2618	8553	6906	-20.9	-5.6	90	14.64	5
-4	-135	-284	-4	-135	4169	3495	3939	3906	-24	-4.3	-135	14.43	6
-3	-90	-291	-3	-90	8101	3793	8064	6210	-12.7	-3.7	-90	14.69	7
-7	135	-275	-7	135	8468	9184	8356	6659	-29	-5.2	128.2	14.7	8
ATTENUATOR													
-9	0	-272	-9	0	1691	3963	2730	4487	-18.8	-10.5	0	14.46	9
-20	0	-278	-20	0	3742	9367	2641	4940	-12.7	-16	-7.9	14.43	10
WIDER BAND PS and VM (14.75-15.25 GHz)													
-1	90	-22	-1	90	0	237	0	3120	-29	-0.7	90	14.44	11
-1	0	-26.6	-1	0	6227	6341	6141	4981	-19.7	-1.3	1.8	14.75	12
-3	90	-15	-3	90	4350	7669	3963	4615	-25.3	-2.8	90.1	14.47	13
WIDER BAND ATTENUATOR (14.5-15.5 GHz)													
NA	-3	-16	-3	-	19110	5845	6177	15970	-33.9	-3.3	NA	14.84	14

Table 4.4: The summary of the simulation and measurement results of the TST insertion phase/amplitude control circuit designs that are implemented using single metal layer process.



(a) (-1 dB, 0°) state (Area:  $15.4 \times 8 \text{ mm}^2$ ).



(b) (-4 dB, -135°) state (Area:  $13.4 \times 7 \text{ mm}^2$ ).



(c) The closer view of the T-junction and the wire bonded CPW ground connections.

Figure 4.13: The photographs of examples of the fabricated single metal layer TST based insertion phase/amplitude control circuits.

The fabricated samples are measured using Agilent 8361A network analyzer and Cascade Summit 9000 manual probe station. Cascade ACP-GSG-150 coplanar probes are used during the measurements. TRL calibration technique is applied for the measurements, and the measurements are made in 5-20 GHz frequency band.

The parameters of the 50  $\Omega$  CPW lines are extracted from the measurements of the CPWs that are fabricated on the same wafer with the TSTs. The extracted parameters for the line are  $Z_o = 50.4 \Omega$ ,  $\varepsilon_{r,eff} = 2.43$ , and  $\alpha = 58 \text{ dB/m} @ 20 \text{ GHz}$ . The characteristic impedance and the attenuation constant of the CPW lines are almost same as the calculated values, which are  $Z_o = 50 \Omega$  and  $\alpha = 60 \text{ dB/m} @ 20 \text{ GHz}$ ; however, the effective permittivity is slightly higher than the design value, which was calculated as  $\varepsilon_{r,eff} = 2.36$ .

Figure 4.14, Figure 4.15, Figure 4.16, and Figure 4.17 show the measurement results of the fabricated TSTs. The figures prove that the desired performances are achieved from the circuits. Table 4.4 also shows the performance of all of the designed TST circuits.



Figure 4.14: The measurement results of the TST based single metal layer PS for (-1 dB, 180°) state compared with the simulations (a)  $|S_{21}|$  (b)  $\angle S_{21}$  (c)  $|S_{11}|$ .



Figure 4.15: The measurement results of the TST based single metal layer VM for (-6 dB, 90°) state compared with the simulations (a)  $|S_{21}|$  (b)  $\angle S_{21}$  (c)  $|S_{11}|$ .



Figure 4.16: The measurement results of the TST based single metal layer attenuator without phase offset for (-9 dB, 0°) state compared with the simulations (a)  $|S_{21}|$  (b)  $\angle S_{21}$  (c)  $|S_{11}|$ .



Figure 4.17: The measurement results of the TST based single metal layer wider band attenuator for (-3 dB) state compared with the simulations (a)  $|S_{21}|$  (b)  $\angle S_{21}$  (c)  $|S_{11}|$ .

Considering the performances that are given in the figures, some shift is observed in the frequency response of all of the circuits, and the desired performance is achieved at different frequencies for all circuits. These frequencies are slightly less than the center design frequency, which is 15 GHz. This situation has two reasons. The first one is that the effective relative dielectric constant used in the designs, 2.36, is lower than the measured one, 2.43. Therefore, all the CPW lines are electrically longer than the simulated ones, and the desired performance shifts to the lower frequencies. The second reason is the repeatability of the wirebonds. There are 3 wirebonds in each T-junction, which makes a total of  $3 \times 3 = 9$  wire bonds for each circuit. These wirebonds are placed in a narrow area at the edge points of the CPW ground, and it is hard to repeat the positions and the heights of these bonds change the parasitic capacitance in each junction point; hence, the center frequency of each circuit slightly shifts in different amounts with respect to the design frequency.

## 4.5. Other Possible Application Areas

It is shown in the previous subsections that insertion phase and amplitude control is possible using TST while keeping the input impedance of the circuit equal to exactly  $Z_o$ . This allows us to use the idea to implement fixed phase shifter or vector modulators using one TST. However, the usage of the idea is not limited to this. It is shown in the second subsection of this chapter that it is also theoretically possible to control the insertion phase of the input signal while making  $Z_o$ -to- $kZ_o$  impedance transformation, i.e., having different real input impedance. This is useful for designing some other useful circuits using the TST.

The concepts that are presented from this point on will depend only on the results of the lossless formulation for insertion phase control with  $Z_o$ -to- $kZ_o$ 

transformation. It was shown previously in this chapter that a solution can be found in the presence transmission line loss for all times even though the lossless formulation starts to fail at some insertion phase ranges. The formulation of the insertion phase control with  $Z_o$ -to- $kZ_o$  transformation is very similar to that of with  $Z_o$ -to- $Z_o$  transformation. As a result, it is believed that solutions can also be found for the insertion phase control with  $Z_o$ -to- $kZ_o$  transformation in the presence of the transmission line losses, and it is unnecessary to show it one more time.

The circuits that are given next are for presentation of further applications of the proposed idea, and the details are not included here.

### 4.5.1. IQ Divider

Figure 4.18 shows the block diagram of the proposed "IQ divider" topology that uses TSTs. Two TSTs are used in the proposed topology, which are used to transform  $Z_0$ -to- $2Z_0$  while the insertion phases of the two arms are set as 0° and 90°. In this case, the impedance at the input port is equal to  $Z_0$ , and the power is divided equally as the input impedances at the two arms are real and equal to each other. Since the insertion phases of the two arms are set as 0° and 90°, one of the arms will be the inphase signal where the other will be the quadrature signal.



Figure 4.18: The block diagram of the proposed "IQ Divider" topology that uses TSTs.

#### 4.5.2. 1:k Power Divider

Figure 4.19 shows the block diagram of the proposed "1:*k* Power Divider" topology that uses TSTs. Two TSTs are used in the proposed topology, which are used to transform the input admittance of the two arms as  $Y_o$ -to- $G_1$  and  $Y_o$ -to- $G_2$ . The sum of  $G_1$  and  $G_2$  should be equal to  $Y_o$  for the input matching as given in (4.20), and the power will be divided proportionally with the ratio of the real parts of the input admittances of the two arms as given in (4.21). Solving these two equations, it is found that  $G_1$  and  $G_2$  should be selected as given in (4.22). The insertion phases are not important for this case; they can either be set both to 0° for inphase operation at the two arms, or they can be set free to any insertion phase values,  $\phi_1^{\circ}$  and  $\phi_2^{\circ}$ .



Figure 4.19: The block diagram of the proposed "1:k Power Divider" topology that uses TSTs.

$$Y_{in} = G_1 + G_2 = Y_0 \tag{4.20}$$

$$Power \, ratio = \frac{G_1}{G_2} = k \tag{4.21}$$

$$G_1 = \frac{k}{k+1} Y_0, G_2 = \frac{1}{k+1} Y_0 \tag{4.22}$$

## 4.5.3. Vector Modulator Type I

Figure 4.20 shows the block diagram of the proposed "Vector Modulator Type I" topology that uses TSTs. Two TSTs are also used in the proposed topology, which are used to transform the input admittance of the two arms as  $Y_{o}$ -to- $G_1$  and  $Y_{o}$ -to- $G_2$ . The  $G_1$  and  $G_2$  should be selected as in the 1:k power divider topology. The insertion phases should be set as either 0° or 180° and either 90° or 270° for the upper and the lower arms, respectively. This is necessary for the vector modulator to cover all four quadrants of the polar space.

Setting the arms as explained above, the input signal will be divided into two arms orthogonally with the desired ratio while the impedance at the input port will be equal to  $Z_o$ . And finally, combining these two signals with an inphase combiner, the vector modulator operation can be achieved.



Figure 4.20: The block diagram of the proposed "Vector Modulator Topology I" that uses TSTs.

## 4.5.4. Vector Modulator Type II

Figure 4.21 shows the block diagram of the proposed "Vector Modulator Type II" topology that uses TSTs. Two TSTs are again used in the proposed topology, which are used to transform the input admittance of the two arms as  $Y_{o}$ -to- $G_1$  and  $Y_o$ -to- $G_2$ . The  $G_1$  and  $G_2$  should be also selected as in the 1:k power divider topology in this case. The insertion phase at the output arm should be set to the required insertion phase,  $\phi_{req}$ , and the insertion phase at the termination arm should be set to any values,  $\phi_3^{\circ}$ .

Setting the arms as explained above, the input signal will be divided into two arms with the desired ratio. This will result with transferring the necessary amount of

power to the upper arm. The upper arm will be the output arm, and it will be used to control the insertion phase of the output signal. The lower arm is used to terminate the amount of power that is left from the upper arm.



Figure 4.21: The block diagram of the proposed "Vector Modulator Topology II" that uses TSTs.

## 4.6. Conclusion

In this chapter, it is shown that it is possible to control the insertion phase and the amplitude of an input signal while keeping the input impedance equal to any desired real impedance using the TST. The analytical solutions for both phase and amplitude control are examined, and the solution for the insertion phase control for the lossless case is presented. The solution for the lossy case, which is actually the amplitude control case, does not give explicit equations; however, a design methodology is developed in the presence of losses. The reasons of the losses are discussed in detail. For the experimental verification, phase shifters, vector modulators, and attenuators are designed for a single metal layer process. The

measurement results of the fabricated structures verified the proposed idea. Finally, other possible application areas of the proposed idea are presented.

# **CHAPTER 5**

# RF MEMS PHASE SHIFTER USING TRIPLE STUB TOPOLOGY

## 5.1. Introduction

The idea of using triple stub topology as insertion phase and amplitude control circuit brings the opportunity to implement novel reconfigurable phase shifters that have a large insertion phase range with fine phase resolution. Achieving these specifications requires precise control of electrical lengths of the transmissions lines, which necessitates many switching components. At this point, MEMS technology is the best alternative since the MEMS technology offers low-loss, high-performance capacitive switches ([5], [8]-[15]), and therefore, using a high number of switches do not increase the insertion loss considerably.

In this chapter, a novel RF MEMS phase shifter that uses the TST is presented. In this topology, the electrical lengths of the stubs of the TST and the interconnection lines between the stubs are adjusted by means of distributed MEMS transmission lines (DMTL) [22]-[28]. By this way, insertion phase can be adjusted as explained in the previous chapter. In the following parts of this chapter the details of the design are explained and the details of the fabrication are given together with the measurement results, and finally, the measurement results are examined.

#### **5.2.** The Design of the RF MEMS Phase Shifter

Figure 5.1 gives the block diagram of the RF MEMS phase shifter (PS). This PS is composed of three main building blocks; (i) DMTL interconnection lines, (ii) DMTL stubs, and (iii) the connecting T-junctions. All of these components are very critical and directly affects the performance of the PS; so, each component is designed separately. Then, the component performance is verified as a part of the PS by cascading the S-parameters of all of the components in a circuit simulator, which is, in this case, Microwave Office (MWO) [175].



Figure 5.1: The block diagram of the RF MEMS phase shifter.

The PS is controlled digitally because all of the unit sections of the DMTLs are either in *on* or in *off* states. In the DMTL stubs, all of the unit sections are controlled independently, and these unit sections are used to obtain the required reflection coefficient at input of the corresponding stub. On the other hand, the unit sections of DMTL interconnection lines are controlled in groups because the overall impedance of these lines should not be altered significantly while adjusting the electrical lengths, i.e., the insertion phase, precisely. The importance of keeping the characteristic impedance in the same level was explained in the previous chapter. It should also be emphasized at this point that, the DMTL interconnection in the PS design is *used to change the electrical length, i.e., the insertion phase,* and hence, it is not used as a *differential phase shifter*.

Transmission lines are implemented as coplanar waveguides (CPW) because it is easier to implement DMTL using CPWs. 15 GHz is chosen as the center frequency, and 0-360° phase range coverage with 10° resolution is aimed, which requires at least 5-bit digital control. The resolution can be increased without altering the structure. The limiting point in resolution is the phase error which is an absolute value. It should be mentioned here that the phase shifter operation is not limited to center design frequency, 15 GHz. Since the electrical lengths of the DMTL interconnection lines and DMTL stubs are reconfigurable, it is also showed in this thesis that the phase shifter can work up to 40 GHz. In the following subsections, the design of each part is explained in detail.

## 5.2.1. The T-Junction Design

The discontinuity at the connection points of planar transmission lines are subject to mode conversion. The mode conversion may result with unexpected increase in the insertion and return losses at the junction point ([178], [179]). Moreover, T-junctions may affect the electrical lengths of the stubs and the interconnection lines between the stubs, which should be precisely adjusted to achieve the desired insertion phase from the PS. As a result, the T-junction design is a critical part of the PS design.

A number of methods have been used in order to reduce the mode conversion in CPW T-junctions [179], and one of the most commonly used methods is using air bridges. Connecting the grounds of the CPW in the vicinity of the junction by air

bridges decreases the unwanted parasitic modes. The air bridges can easily be implemented using the RF MEMS process, and hence, is chosen for the T-junctions of the PS design.

A special CPW T-junction is designed for the connection points between the DMTL stubs and the DMTL interconnection lines, the layout of which can be seen in Figure 5.2. MEMS air bridges are used to connect the grounds of the CPW and the CPW center conductor is narrowed. Moreover, a notch is opened at the connection point on the thru line direction. This layout minimizes the effects of the parasitics that occur at the junction point.



Figure 5.2: The layout of the designed T-junction. The dimensions are given in  $\mu$ m.

The reason of using air bridges was explained above. The reason of using narrower lines and the notch can be understood better by examining the electrical model of the T-junction, which is given in Figure 5.3 [179]. In the model,  $C_{air-bridge}$  is used to model the parasitic capacitances brought by the air bridges,  $C_{junction}$  is used to model

the fringe capacitance occurring at the junction point, and the  $L_1$  and  $L_3$  are used to model the change in the current densities at the junction point. The T-junction is a strong discontinuity and it affects the PS performance significantly, therefore a fundamental solution is to design the T-junction such that it only behaves as connection of three 50  $\Omega$  transmission lines. With this solution, the equivalent electrical lengths of the T-junction arms are subtracted from the required electrical lengths of the interconnection lines and the stubs. For this purpose, a notch is used at the junction point in order to reduce  $C_{junction}$  as much as possible, and the center conductors of the CPWs are narrowed to change  $L_1$  and  $L_3$  so that these inductances compensates the  $C_{air-bridge}$  and the  $C_{junction}$ .



Figure 5.3: The electrical model of the T-junction discontinuity.

The T-junction is designed using HFSS v9.2 [176]. The design parameters are the width and the position of the MEMS air bridge, the width and the depth of the notch, and the width of the narrowed CPW center conductor. An iterative design method is followed. The T-junction simulation results are substituted into the PS design, and the simulation results with and without T-junctions are compared. The

circuits that are used to evaluate the performance of the T-junctions are given in Figure 5.4.



Figure 5.4: The PS models that are used to compare the effects of the T-junction: (a) The PS without T-junction (b) The PS with T-junction.

The T-junction design method is as follows: the PS is simulated using the circuit model given in Figure 5.4-(a) in MWO, where  $l_1$ ,  $l_2$ ,  $d_1$ ,  $d_2$ , and  $d_3$  are fixed to some predefined lengths. Then, the EM, HFSS v9.2, simulation results are inserted in the PS circuit model as shown in Figure 5.4-(b). Following this step, the error between the S-parameters of the two circuits, which is defined as the "average squared magnitude of the difference between each element of the S-parameter matrix" in this case, is minimized by optimization. The optimization is made using the Simplex Optimization tool of the MWO [175], and the optimization parameters are x and y

in Figure 5.4-(b), i.e., the lengths of the extra 50  $\Omega$  transmission lines that are subtracted from the original line lengths in order to compensate the presence of the T-junction.

In order to demonstrate the necessity of such a complicated design for the T-junction, the simulation results of the T-junction with MEMS air bridges only, which is the initial design, is compared with that of the final design, the layout of which was given in Figure 5.2. Figure 5.5 shows the S-parameters and the error between S-parameters of the PS for the cases without any T-junction, with the initial T-junction design, and the final T-junction design. The line lengths are given in Table 5.1. As can be seen in the figure, the error between the S-parameters for the initial design case, which is seen on the right axis, is much bigger than that of the final design case. The difference can also be easily observed on the S-parameters. It should be pointed out here that  $l_1$ ,  $l_2$ ,  $d_1$ ,  $d_2$ , and  $d_3$  set was selected randomly; in spite of this, the T-junction still works for *any* set of these variables, and the error does not change considerably.

Considering the performance of the final design, this T-junction is selected to be used in the RF MEMS PS design.



Figure 5.5: The comparison of the S-parameters of the PS for with and without T-junction cases: (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c) The error between the S-parameter of the two cases.

	Initial Design	Final Design
$d_1$	14500	14500
$d_2$	5700	5700
<i>d</i> <sub>3</sub>	3700	3700
$l_1$	2500	2500
$l_2$	1000	1000
X	731.7	694.9
у	367.6	356.1

Table 5.1: The line lengths used in the PS model for the comparison of with and without T-junction cases. All the parameters are given in  $\mu$ m.

## 5.2.2. The DMTL Interconnection Line Design

DMTLs are one of the most widely used MEMS components. Although the most popular application area is the phase shifters, DMTLs are also demonstrated in resonators, filters, and impedance matching networks, the examples of which were given in Chapter 1. The main application areas of the DMTLs are phased arrays, radars, wireless communication systems, and measurement instruments in a very large frequency band starting from X-band up to W-band.

Operation principle of the DMTL is based on the theory of periodic structures ([172], [180]), which is actually means loading a transmission line periodically with reactive elements so that by changing the reactance of the loading elements, the phase velocity of the wave travelling inside the structure can be adjusted. The reactive element in the DMTL is the MEMS switch. Using MEMS switches as tunable capacitors, in which case the capacitance of the switch is tuned by changing the applied voltage, the phase velocity of the travelling wave can be controlled in an *analog* manner [22]-[23]. However, the maximum phase shift that can be obtained from this structure is limited by the capacitance tuning range of the MEMS switches, which is theoretically limited to 1.5 for standard switches caused by the mechanical instability of the MEMS switches [23]. On the other hand, *digital* type DMTL structure have also been presented in the literature ([24], [25], [27], [28]), in

which the MEMS switch is serially connected to another static MEMS capacitor and is used to select one of the two equivalent capacitance levels by holding the switch either in on or off states. In this case, the capacitance tuning range can be increased to 2.5 [27], which means that more phase shift can be obtained from one DMTL unit section. This also means that the degree per dB, shortly °/dB, performance of the structure is better in the digital case ([181], [142]). Moreover, the digital type structure shows stronger resistance to electrical and mechanical noise coming from the analog control of the MEMS structure [182]. Considering the performance of the digital type MEMS phase shifters, it is also selected as to change the electrical length, i.e., the insertion phase, of the interconnection line of the RF MEMS PS designed in this thesis.

Figure 5.6 shows the basic circuit model of a unit section of a digital type DMTL. In the model,  $sR_t$ ,  $sL_b$  and  $sC_t$  are the per unit length parameters of the unloaded transmission line, which is the CPW in this case. The unloaded transmission line is loaded by two MEMS capacitors, which are  $C_b$  and  $C_s$ .  $C_b$  represents the capacitance of the MEMS switch whereas  $C_s$  represents the capacitance of the static MEMS capacitance, which is, in this case metal-air-metal (MAM) capacitors. Since the unloaded CPW is *loaded* by two capacitances that are connected in series, the parameters of the equivalent structure are modified forming a new distributed transmission line structure. The loaded impedance and the phase velocity of a loaded transmission line are given in (5.1) and (5.2), respectively.



Figure 5.6: The circuit model of a unit section of a digital type DMTL.

$$Z_l = \sqrt{\frac{sL_t}{sC_t + \frac{C_bC_s}{C_b + C_s}}}$$
(5.1)

$$v_p = \frac{s}{\sqrt{sL_t\left(sC_t + \frac{C_bC_s}{C_b + C_s}\right)}}$$
(5.2)

The standard design methodology of the digital type DMTL starts with the selection of the Bragg frequency, which is given in (5.3), since it is the upper limit for the periodic structure [180]. The safe way is to select Bragg frequency as more than twice the center frequency of the DMTL. The upstate and the downstate characteristic impedances,  $Z_{lu}$  and  $Z_{ld}$ , are also selected as to stay in an acceptable return loss level, which is usually selected as -15 dB. The calculated impedance levels for upstate and downstate are 60  $\Omega$  and 42  $\Omega$ , respectively [27]. Then, the phase shift per unit section of the DMTL can be found using (5.4).

$$f_B = \frac{1}{\pi \sqrt{sL_t \left(sC_t + \frac{C_b C_s}{C_b + C_s}\right)}}$$
(5.3)

$$\Delta \phi = \frac{s2\pi f_0 Z_0 \sqrt{\varepsilon_{eff}}}{c} \left(\frac{1}{Z_{lu}} - \frac{1}{Z_{ld}}\right) rad/section$$
(5.4)

The design parameters such as the periodicity, the static MAM capacitance,  $C_s$ , and the MEMS switch capacitance,  $C_{bu}$ , can be found using (5.5)-(5.7). The reader is encouraged to see [142] for an extensive analysis and a good example of the DMTL design.

$$s = \frac{Z_{ld}c}{\pi f_B Z_0 \sqrt{\varepsilon_{eff}}}$$
(5.5)

$$C_s = \frac{Z_0^2 - Z_{ld}^2}{\pi f_B Z_0^2 Z_{ld}}$$
(5.6)

$$C_{bu} = C_s \frac{Z_{ld}^2 \left( Z_0^2 - Z_{lu}^2 \right)}{Z_0^2 \left( Z_{lu}^2 - Z_{ld}^2 \right)}$$
(5.7)

Although the standard design methodology explained above is well-suited and proven for the DMTL design, a different method has been followed during the design of the DMTL section that is used in the RF MEMS PS design. This is because the RF MEMS PS DMTL interconnection line sections have different design requirements and limitations.

First of all, the RF MEMS PS design requires a specific *insertion phase* between the stub connection points for a predefined phase shift value of the whole structure. Hence, the important parameter for the DMTL interconnection line is *not only the differential phase shift, but also the upstate insertion phase* of the DMTL interconnection line.

Secondly, a high phase resolution is required for the DMTL interconnection line since the phase performance of the PS is very sensitive to small changes in this parameter. It is required by (4.7) to (4.10) that in order to obtain the targeted insertion phase range of the designed PS, i.e., 0-360° range with 10° resolution, the electrical length of the interconnection line should be changed between 0 and  $\lambda/2$ 

with minimum  $\lambda/100$  steps. This is requirement is valid when the minimum insertion loss point is targeted for each phase state of the PS. This means that the DMTL interconnection line should have  $(\lambda/2) / (\lambda/100) = 50$  unit sections, which has  $180^{\circ}/50 = 3.6^{\circ}$  phase resolution. Let us assume that  $6.125^{\circ}$  resolution, which is equivalent to 6-bit controlled phase shifter, is enough instead of  $3.6^{\circ}$ . Applying the above DMTL formulation for a  $60 \Omega$ - $40 \Omega$  design at 15 GHz with a Bragg frequency of 35 GHz results with periodicity of  $s = 740 \,\mu$ m. This results with a total DMTL interconnection line length of  $32 \times 740 = 23690 \,\mu$ m. Taking into account that two of these DMTLs will be used in the design, this approach is inapplicable in the sense that the design becomes infeasible to fabricate; moreover, the insertion loss will increase significantly.

Other than the above mentioned facts, there are also other limitations that are related with the measurement and packaging of the PS, which should be taken into account during the design. The PS has many DC control connections that are used to actuate the MEMS switches, and these connections should be somehow connected to the outer world for the measurements. This can be done in two ways, (i) using a probe card in the probe station and (ii) connecting the fabricated design into an integrated circuit (IC) package that allows the connection of many DC controls. Using a probe card is not feasible since each design necessitates a different probe card. The multi-pin MEMS automated measurement system (MAMS), which was explained in Chapter 3.3.2, will be used for the measurement of the designed PS. The MAMS allows 32 DC control pins simultaneously; so, this is also a design constraint for the PS. Additionally, the packages that the PS will be mounted have standard window sizes excluding the custom packages, and the maximum available window sizes limit the area of the PS design.

The design constraints explained above makes the usage of a standard digital type phase shifter impossible for the DMTL interconnection line, which means that an alternative approach is needed. For this purpose, a DMTL that has a lower insertion phase range and a smaller area is designed. The design has a total of 8 unit sections, which are controlled by 5 pins. It uses two types of DMTL unit sections that are designed to give 5° and 10° differential phase shifts per unit section, respectively. The designed DMTL interconnection line gives an insertion phase of -115° and -178° when all the MEMS switches are in upstate and downstate, respectively.

The reason behind selecting the aforementioned insertion phase range and the number of unit sections is understood better if the phase shifter theory, which was given in the previous chapter, is investigated in more detail. It is theoretically known that the interconnection line length should be changed between 0 to  $\lambda/2$  in order to obtain 0-360° insertion phase range. However, the interconnection line length that is necessary in order to obtain one specific insertion phase value with minimum insertion loss also works for the neighboring insertion phase values. In other words, once the electrical length of the interconnection line length is fixed to some value, a range of insertion phase can be realized with the PS. But, this range is not constant and depends on the interconnection line length. The range is maximum when the interconnection line length is about  $\lambda/4$ , and it decreases as it goes towards either 0 or  $\lambda/2$ , which can be seen in Figure 5.7. The range becomes so small near 0 or  $\lambda/2$  that the electrical length of the interconnection line should be exactly what is required by the formulation in order to obtain the desired insertion phase. So, during the interconnection line design, the electrical length is targeted to stay in 0-to-0.1 $\lambda$  and 0.4 $\lambda$ -to-0.5 $\lambda$ , which is equivalently about from -145° to -215°, and this range is covered by 5° steps to be able to cover the range accurately.

The electrical length of the interconnection line is determined by the length of the DMTL interconnection line and the extensions of the T-junctions at each stub connection point. The additional transmission lines that come from both sides of

the T-junction add  $2 \times -20 = -40^{\circ}$ . Adding the value to the insertion phase coming from the extensions, the interconnection line has a total insertion phase between  $-115 - 40 = -155^{\circ}$  and  $-178 - 40 = -218^{\circ}$ .





(c)  $0.415\lambda = 8300 \ \mu m$ .

Figure 5.7: The  $S_{21}$  of the ideal TST based PS with different interconnection lines lengths. The figures show two insertion phase values when the insertion loss is minimum and when the insertion loss is equal to -3 dB.

Having identified the design requirements and targets, the next step is to design the DMTL unit sections. Two types of unit sections, which are named as *DMTL Unit Section Type I* and *Type II*, are designed for the DMTL interconnection line that give 5° and 10° differential phase shift per unit section, respectively. DMTL Unit Section Type I (UST-I) is used to decrease the size of the phase steps, and two unit sections of UST-I are used. All of the other remaining six unit sections are DMTL Unit Sections of Type II (UST-II).

The design of the DMTL unit sections begins using the formulation (5.1)-(5.7), which is followed by verifying the results using the circuit model that is given in Figure 5.8. Some key points should be considered here. First, the Bragg frequency should be adjusted so that the upstate insertion phase is not lower than -14° in order not to exceed -115°. Secondly, the differential phase shift should be 5° and 10° for UST-I and UST-II, respectively. And finally, the MEMS bridge widths should be less than 80 μm if possible for proper operation of the CLR model.



Figure 5.8: The circuit model of the DMTL unit section that is used for the design of the DMTL interconnection line.

Using (5.1)-(5.7), the parameters that are given in the first and the third columns of Table 5.2 are calculated as a starting point for the design for UST-I and UST-II, respectively. The impedance, effective relative dielectric constant, and the total loss of the CPW are calculated using the formulation in [172], and verified with the measurement results of the previously fabricated single metal layer TST based phase shifters, which were explained in Chapter 4. The impedance of the unloaded transmission line is selected as 96  $\Omega$ , and the CPW unloaded line loss is minimum around this impedance level. The CPW losses become minimum when the center conductor width is selected the same as the spacing between the signal and the ground. After determining the capacitance values and the periodicity, the physical dimensions of the unit section are extracted using capacitance formulation (3.1) to (3.3) given in Chapter 3. Here,  $C_{bu}$  and  $C_s$  values had to be modified as the parallel plate down state capacitance, C<sub>bd</sub>, should be multiplied with 0.37 as explained in Chapter 3, which, indeed, causes the MEMS bridge to affect not only the upstate capacitance, but also the down state capacitance. Then, the unit sections are simulated using the model given in Figure 5.8 in MWO to adjust the capacitances, where the effects of the  $L_b$ ,  $R_{bu}$ ,  $R_{bd}$ , and  $R_{bias}$  are not included.

	DMTL	UST-I	DMTL UST-II			
	DMTL	Circuit Model	DMTL	Circuit Model		
	Formulation	HFSS Sim.	Formulation	HFSS Sim.		
$Z_o$	96 Ω	96 Ω	96 Ω	96 Ω		
$\mathcal{E}_{r,eff}$	2.36	2.36	2.36	2.36		
α	40 dB/m	40 dB/m	40 dB/m	40 dB/m		
	@20 GHz	@20 GHz	@20 GHz	@20 GHz		
$f_{Bragg}$	90 GHz	73 GHz	69 GHz	59 GHz		
S	302 μm	300 μm	300 µm	300 µm		
$Z_{lu}$	58 Ω	55.9 Ω	54 Ω	52.9 Ω		
$Z_{ld}$	42 Ω	40.9 Ω	32 Ω	34.5 Ω		
$\phi_{\!up}$	-13.8°	-14.1°	-14.8°	-15°		
$\phi_{down}$	-19.1°	-20°	-24.9°	-24°		
$\Delta \phi$	5.3°	5.9°	10.1°	9.1°		
$C_{bu}$	47.7 fF	46.4 fF	47.4 fF	46.7 fF		
$C_{bd}$	458.4 fF	469.2 fF	458.4 fF	469.2 fF		
$C_s$	68.1 fF	89.5 fF	128.1 fF	147.6 fF		
$L_b$	-	30 pH	-	30 pH		
$R_{bu}/R_{bd}$	-	$1.6/0.15~\Omega$	-	$1.6/0.15~\Omega$		
$C_U/C_D$	28.1 fF/59.3 fF	30.6 fF/ 75.2 fF	34.6 fF/100.1 fF	35.5 fF/112.3 fF		
$Q_U/Q_D$	-	27/12	-	60/27		
R <sub>bias</sub>	-	1400 Ω	-	1542 Ω		
R <sub>bias,DC</sub>	-	20.25 kΩ	-	18.375 kΩ		

Table 5.2: The circuit design parameters of the DMTL UST-I and DMTL UST-II.

Table 5.3: The physical design parameters of the final designs of the DMTL UST-I and UST-II.

Parameter	UST-I	UST-II		
W	150 μm	150 μm		
G	150 μm	150 μm		
l <sub>brd</sub>	300 µm	300 µm		
Wbrd	38 µm	38 µm		
<b>W</b> gnd	40 µm	40 µm		
l <sub>brd-g</sub>	138 µm	138 µm		
W <sub>brd-g</sub>	49 µm	2 x 44 µm		
Wgo	30 µm	30 µm		
$l_{go}$	40 µm	40 µm		
WSiCr	20 µm	20 µm		

The unit sections designed in MWO are also simulated and optimized in HFSS v9.2 in order to obtain the required capacitance values. The 3D and the top views of the UST-I are given in Figure 5.9. The dimensions used for the final design for DMTL UST-I and UST-II are given in Table 5.3. During the simulations and optimization using HFSS v9.2, the following points have been taken into account:

- The quartz substrate with parameters  $\varepsilon_r = 3.8$  and  $tan \delta = 0.001$  are used. The conductivity of gold is used as  $\sigma_{Au} = 3 \times 10^7$  S/m.
- During the downstate simulations, the relative permittivity of silicon nitride is used as  $\varepsilon_r = 2.53$  instead of 7 in order to include the 0.37 factor.
- The conductivity of the SiCr bias lines is taken as  $\sigma_{SiCr} = 12500$  S/m, which is extracted from the measurement results of the previous runs.
- The dimensions of the CPW was selected as  $G/W/G = 150/150/150 \mu m$ . Although selecting the CPW dimensions as high as possible reduces the losses, there is an upper limit to provide higher order mode-free operation [183]. Considering the center design frequency, which is 15 GHz in this case, this selection is a good compromise considering both the line losses, and the actuation voltage of the MEMS bridge as it directly depends on the CPW signal width.
- MEMS bridge was fixed to  $w_{brd} = 38 \ \mu\text{m}$  in width and  $l_{brd} = 300 \ \mu\text{m}$  for both unit section types. The reason behind this selection is to prevent different mechanical buckling behavior of the MEMS bridges for different types of unit sections, and hence, to prevent different MEMS bridge capacitances. The MEMS bridges with wider and longer dimension are observed to buckle more easily in case of any residual stress. Moreover,  $w_{brd}$  was tried to be fixed around 50  $\mu$ m in order the CLR model to work properly [184].
- The MAM capacitances in the DMTL structures have shown to be "buckled" more than the standard fixed-fixed MEMS bridge because of the residual stress since the area of these capacitors become large so as to achieve large capacitance value. To prevent this behavior, a modified MAM capacitance is

used in the DMTL unit sections in this thesis. First, the MAM is fixed from two sides instead of three to attain a mechanically balanced structure. Secondly, an additional anchor is used in the longitudinal dimension to avoid buckling. And finally, for the MAMs with more than 50  $\mu$ m width, the MAM is divided into strips, whose width is about 50  $\mu$ m and gap is 6  $\mu$ m, as to prevent the buckling in the lateral dimension.

• The SiCr bias lines were shorted to the CPW grounds on both sides with the aim of modeling the connection to biasing network.





(a)


Figure 5.9: (a) 3D view of the DMTL UST-I (z-dimension exaggerated) (b) Top view of the DMTL UST-I (c) 3D view of the DMTL UST-II (z-dimension exaggerated).

During the design in HFSS, all the parameters are extracted by dedicated and separate simulations. In order to extract the MEMS bridge parameters in the upstate, C<sub>bu</sub> and R<sub>bu</sub>, the unit sections are simulated using an HFSS model in which only the MEMS bridge is present, and it is connected to the CPW grounds without any MAM capacitances. Then, these parameters are extracted by cascading the S-parameters of 16 unit sections that are simulated in HFSS and fitting the overall S-parameters with that of the circuit model. The MEMS bridge parameters in the downstate, C<sub>bd</sub>, R<sub>bu</sub>, and L<sub>b</sub>, are extracted in a similar manner, where the same HFSS model is used except the MEMS bridge is simulated in downstate. The MAM capacitance parameters, C<sub>s</sub> and R<sub>bias</sub>, are extracted with an HFSS model where only the MAM capacitance exists, and the center conductor of the CPW is directly connected the lower plate of the MAM capacitance. In the model, the parasitic capacitances, and the loss factors due to the SiCr biasing resistors and the CPW Then, all the components, i.e., the ground overpasses are also included. capacitances, bias lines, and overpasses, are combined, and the HFSS simulations are repeated for both upstate and downstate to verify and optimize the model. To get the desired capacitance values, the dimensions of the MAM capacitance are tuned, and the MEMS bridge dimensions are not changed.

The results of the HFSS simulations yielded the parameters presented in the second and the fourth columns of Table 5.2. Here,  $L_b$  represents the inductance of the MEMS bridge. The change in the Bragg frequency is because of the effects of the added inductance to the model [7].  $R_{bu}$  and  $R_{bd}$  model the resistance of the MEMS bridge in the upstate and downstate, respectively.  $R_{bias}$  models the losses coming from Q-factor of the MAM capacitor ([181], [142]). These losses include the ohmic losses of the capacitor itself and the additional losses coming from the SiCr bias lines. The SiCr bias line passes through the CPW ground planes, and the ground planes are patched up by small air bridges due to the discontinuities caused by the bias lines. Due to the distributed nature of the resistor and the coupling between this resistor, the CPW ground, and the air bridges, the equivalent value of  $R_{bias}$  is about a few k $\Omega$ , although it has a DC value of about 20 k $\Omega$ . Because of this fact, the latter factor is the dominant loss contributor.

A lossy capacitor can be modeled as in Figure 5.10, and the quality factor, Q, of the capacitor can be calculated as in (5.8) where  $R_s$  can be written as in (5.9). Here,  $R_p$ is the total resistance representing both the SiCr bias lines and the ohmic losses of the MAM capacitor. It was stated in [181] and [142] that the resistance of the MEMS bridge,  $R_b$ , does not contribute significantly to the losses. However,  $R_{bias}$  is quite effective on the losses since it causes a considerable decrease in the Q-factor of the MAM capacitance as explained above. Since the phase shifter design requires a resistive bias line in almost all unit sections of the designed DMTLs, it was decided to calculate the overall Q-factor,  $Q_D$ , of the equivalent shunt capacitor,  $C_D$ , and to see the total loss contribution of the overall Q-factor. Here, the subscript "D" states that the calculations were made for the downstate, where the losses are higher. The calculations can be repeated for the upstate in a similar manner. To find the overall Q-factor, the total series resistance is calculated using (5.10), where the equivalent series resistance of the MAM capacitor is given in (5.9). The overall Q can be found using (5.11), and the total loss contribution per unit section is given in (5.12).



Figure 5.10: The circuit model of the DMTL unit section that is used for the design of the DMTL interconnection line.

$$Q = \omega C_s R_p = \frac{1}{\omega C_s R_s}$$
(5.8)

$$R_s = \frac{1}{\omega^2 C_s^2 R_p} \tag{5.9}$$

$$R_q = R_{bd} + \frac{1}{\omega QC} \tag{5.10}$$

$$Q_D = \frac{1}{\omega C_D R_q} \tag{5.11}$$

$$\alpha_{ld} = \frac{R_t}{2Z_{ld}} + \frac{R_q Z_{ld} \omega^2 C_D^2}{2} dB / unit section$$
(5.12)

Figure 5.11 gives the comparison of the HFSS simulations with the proposed circuit model simulations. In the simulations, S-parameters of a unit section that is simulated in HFSS is cascaded in MWO, and the result is compared with the results of the circuit model, which is also obtained by cascading 16 unit sections of the circuit model. The circuit model parameters that are used in the simulation are the ones in the second and the fourth column of Table 5.2. Considering the simulation results, it is seen that the circuit model is successful in predicting the EM full-wave simulations. The model can predict the phase response very accurately, and the insertion loss is also well predicted. The absolute difference between the S<sub>11</sub> responses in the upstate is small. The error, which is defined previously in

Section 5.2.1 between the S-parameters for all cases, which is given in Figure 5.11-(e), verifies that the modeling is successful.



(a)  $|S_{11}|$  and  $|S_{21}|$  of DMTL UST-I in upstate (16 unit sections cascaded).



(b)  $|S_{11}|$  and  $|S_{21}|$  of DMTL UST-I in downstate (16 unit sections cascaded).



(c)  $\angle S_{21}$  of DMTL UST-I in both states (16 unit sections cascaded).



(d)  $|S_{11}|$  and  $|S_{21}|$  of DMTL UST-II in upstate (16 unit sections cascaded).



(e)  $|S_{11}|$  and  $|S_{21}|$  of DMTL UST-II in downstate (16 unit sections cascaded).



(f)  $\angle S_{21}$  of DMTL UST-II in both states (16 unit sections cascaded).



(g) The error between the S-parameters of the circuit model and the HFSS simulations of DMTL UST-I and DMTL UST-II for all cases.

Figure 5.11: The comparison of the S-parameters of the circuit model and HFSS simulations for the DMTL UST-I and DMTL UST-II. The graphs are the results of cascaded 16 unit sections both for the circuit model and the HFSS simulation results (a)-(c) DMTL UST-I (d)-(f) DMTL UST-II (g) The error between the S-parameters of the circuit model and the HFSS simulations of both unit sections for all cases.

The loss of the DMTL unit sections is represented by two resistive components in the circuit, which are the resistance of the MEMS bridge,  $R_b$ , and the *effective* resistance of the MAM capacitor and the SiCr bias line,  $R_{bias}$ . The values presented in Table 5.2, especially  $R_{bu}$ , might be seen unphysical regarding the reported resistor values, which vary between 0.1-0.6  $\Omega$ . ([183], [185]). In these works, however, the value of  $R_b$  is extracted from the downstate measurements, and the same value is

also used in the upstate simulations, where the loss discrepancies in the upstate are explained by the changes in the unloaded line losses. In the design of the DMTL unit sections in this thesis, similar values are also obtained from the downstate simulations that are used to extract the circuit parameters. For example, from the HFSS simulations with only the MEMS bridge at downstate, i.e., without any MAM capacitors and biasing resistors,  $R_{bd}$  is extracted as 0.15  $\Omega$ . However,  $R_{bu}$  is extracted as 1.6  $\Omega$  from the HFSS simulations with only the MEMS bridge at upstate. This value results with a lower error between the S-parameters of the HFSS simulation and the circuit model. It should be also noted here that similar values are also reported previously for  $R_{bu}$ .

It is interesting to note here that the error between the S-parameters of the circuit model and the HFSS simulation can be reduced further for a different set of  $R_{bu}$ ,  $R_{bd}$ , and  $R_{bias}$  values. Figure 5.12 shows the errors between the S-parameters for all cases for the values of  $R_{bu}$ ,  $R_{bd}$ , and  $R_{bias}$  that are given in Table 5.4. It is clear that the error in Figure 5.12 is better that what is given Figure 5.11-(e). In the former case,  $R_{bu}$  and  $R_{bd}$  are fixed during the optimization, and  $R_{bias}$  is used as an optimization variable. In the latter case, all three parameters are used as optimization variables during the optimization. The optimization is made in 5-20 GHz band for both cases.

	US	iT-I	UST-II		
	1 <sup>st</sup> method	2 <sup>nd</sup> method	1 <sup>st</sup> method	2 <sup>nd</sup> method	
R <sub>bu</sub>	1.6 Ω	2.1 Ω	1.6 Ω	0.97 Ω	
$R_{bd}$	0.15 Ω	3.5 Ω	$0.15\Omega$	1.33 Ω	
R <sub>bias</sub>	1400 Ω	2411 Ω	1542 Ω	2462 Ω	

Table 5.4: The values of the resistances that are used for the first and the second modeling method for DMTL UST-I and UST-II.



Figure 5.12: The error between the S-parameters of the circuit model and the HFSS simulation of both unit sections for the latter case, where the MEMS bridge resistance is extracted from "MEMS bridge only" HFSS simulations.

The change in  $R_{bu}$  and  $R_{bd}$  in the latter case is due to the way of using the SiCr bias lines. In standard DMTL phase shifter designs, the SiCr bias lines that penetrate through the CPW grounds are used only one for each bit, and these lines are modeled with additional shunt resistors. The unit sections with such type of biasing resistors are known as having the highest loss per unit section ([181], [142]). The designs in the frame of this thesis have such biasing lines in each unit section, and these lines are included in the HFSS simulations at all times. The change in  $R_{bu}$ ,  $R_{bd}$ , and  $R_{bias}$  values are related according to using small  $C_s$  values in the design, which are comparable with  $C_b$ . In this case, the current distribution on the EM structure is different than that of a standard DMTL unit section where  $C_s$  is much bigger than  $C_b$ . Moreover, as the  $C_s$  is small, the fringe capacitances from to the bias lines to the ground overpasses become rationally larger, and these low-Q capacitances become more effective on the overall loss performance.

The quality factors in the upstate and downstate of UST-II is higher than that of UST-I. This is due to the increasing value of  $C_s$ . Moreover,  $R_{bias}$  also increase, which means the losses due to the biasing resistor has decreased, which is the dominant

loss mechanism of the DMTL unit sections of this type. This idea is also supported by the circuit model parameters of UST-III, which is the unit section type that is used in the DMTL stub design.

Figure 5.13 shows the layout of the final design of the DMTL interconnection line, which is composed by cascading two UST-Is and six UST-IIs. The DMTL interconnection line has five control pins, which results with 32 insertion phase states. The simulation results of the final design are given Figure 5.14, which conclude the design of the DMTL interconnection line.



Figure 5.13: The layout of the DMTL interconnection line.



(a)  $|S_{11}|$  and  $|S_{21}|$  of DMTL interconnection line for states 0 and 31.



Figure 5.14: The simulation results of the DMTL interconnection line (a)  $|S_{11}|$  and  $|S_{21}|$  for all the switches are in upstate (State 0) and downstate (State 31) (b)  $\angle S_{21}$  for all states (c) The change  $\angle S_{21}$  with respect to the changes in the states.

#### 5.2.3. The DMTL Stub Design

The electrical lengths of the stubs of the RF MEMS PS are also controlled by using DMTL unit sections. However, the design requirements of the DMTL stubs are yet different than that of the DMTL interconnection line. This is because the stubs are one port networks and the only design requirement is the phase of the input reflection coefficient. As a result, the loaded impedance of the DMTL is not required to be 50  $\Omega$ . Nevertheless, the DMTL should have minimal losses.

The design requirements of the DMTL stubs are determined by the theory of the TST based phase shifter, which was given in Chapter 4. It is required by (4.7) to (4.10) that in order to obtain the targeted insertion phase coverage of the designed PS, i.e., 0-360° range with 10° resolution, the phase of the reflection coefficients of the stubs should be changed between 0-360°. This is requirement is valid when the interconnection line lengths is not limited, and minimum insertion loss point is targeted for each phase state of the PS. But, once the interconnection line length is limited to the design constraints of the DMTL interconnection line, which is 0 to 0.1 $\lambda$  and 0.4 $\lambda$  to 0.5 $\lambda$ , the required values of the DMTL stubs are also changed. Using (4.7) to (4.10) one more time, it is shown that the phase of the reflection coefficients of stub 1 and Stub 3 should be changed between -118° and 145°, the phase of the reflection coefficients of Stub 2 should be changed between -152° and 164° when the interconnection line length is in 0 to 0.1 $\lambda$  and 0.4 $\lambda$  to 0.5 $\lambda$  range.

The design requirements determined by the theory should also be verified by the circuit simulations once the interconnection line is replaced by the final DMTL interconnection line design. For this purpose, the circuit, whose block diagram was given in Figure 5.1, is used to find the necessary phase of the reflection coefficients for the stubs after the DMTL interconnection line is inserted to the circuit model. During the simulations, the stubs are modeled by lossy transmission lines, and the DMTL interconnection line and the T-junctions are used in their final design forms.

Using the Simplex Optimization tool of the MWO, the phase of the reflection coefficient should be changed between -110° and 119° for Stub 1 and Stub 3, and the phase of the reflection coefficient of Stub 2 should be changed between -143° and 120°. In order not to complicate the design, the maximum and minimum values of all the necessary phase of the reflection coefficient are chosen consequently, which are -143° and 120°, and a single DMTL stub design is made.

The DMTL stubs have 9 DMTL Unit Sections of Type III (UST-III). The number of unit sections is determined by the limitations related with the packaging and the measurement setup, which were explained previously in this chapter. The design of the UST-III again starts with using equations (5.1)-(5.11). Here, the unit section is designed using the equations in Matlab, and to check whether the electrical length of the stub is enough to cover the specified range, 9 UST-IIIs are cascaded with the T-junction extension, and the phase of S<sub>11</sub> is checked. The parameters that are obtained using the equations are given in the first column of Table 5.5.

Following to same method that is used during the design of the UST-I and UST-II, the dimensions that are necessary for the physical implementation are extracted using the above mentioned capacitance formulation (3.1) to (3.3) given in Chapter 3. The structure is then simulated and optimized in HFSS v9.2. The 3D view and the physical dimensions of the final design of the DMTL UST-III are given in Figure 5.15 and Table 5.6, respectively.



Figure 5.15: 3D view of the DMTL UST-III (z-dimension exaggerated).

	DMTL	Circuit Model		
	Formulation	HFSS SIM.		
$Z_o$	96 Ω	96 Ω		
$\mathcal{E}_{r,eff}$	2.36	2.36		
01	40 dB/m	40 dB/m		
u	@20 GHz	@20 GHz		
$f_{Bragg}$	66 GHz	50 GHz		
S	216 µm	210 µm		
$Z_{lu}$	46 Ω	44.1 Ω		
$Z_{ld}$	22 Ω	23.7 Ω		
$\phi_{\!up}$	-12.5°	-12.6°		
$\phi_{down}$	-26°	-25.4°		
$\Delta \phi$	13.6°	12.8°		
$C_{bu}$	47.5 fF	46.3 fF		
$C_{bd}$	458.4 fF	469.2 fF		
$C_s$	207 fF	296.7 fF		
$L_b$	-	30 pH		
$R_{bu}/R_{bd}$	-	0.46 Ω/0.45 Ω		
$C_U/C_D$	38.6 fF/142 fF	40.1 fF/181.8 fF		
$Q_U/Q_D$	-	114/67		
<i>R</i> <sub>bias</sub>	-	1784 Ω		
R <sub>bias,DC</sub>	-	15 kΩ		

Table 5.5: The circuit design parameters of the DMTL UST-III.

Parameter	UST-III		
W	150 μm		
G	150 μm		
$l_{brd}$	300 µm		
Wbrd	38 µm		
<b>W</b> gnd	46 µm		
$l_{brd-g}$	138 µm		
W <sub>brd-g</sub>	4 x 43 μm		
Wgo	30 µm		
$l_{go}$	40 µm		
WSiCr	20 µm		

Table 5.6: The physical design parameters of the final design of the DMTL UST-III.

Figure 5.16 gives the comparison of the HFSS and the proposed circuit model simulations. The simulations are compared as in the design of UST-I and UST-II. The circuit model parameters that are extracted from the HFSS simulations are given in the second column of Table 5.5. The simulation results again show that the model is successful in predicting the EM full-wave simulations, which is also quantitatively given in Figure 5.16-(d). Lastly, Figure 5.17 shows the S<sub>11</sub> characteristics of the final DMTL stub design for all possible states,  $2^9 = 512$ , in which 9 UST-IIIs are cascaded and terminated with an open circuit. The simulation circuit also includes the transmission line that models the T-junction extension. The simulation results show that the circuit satisfies the design requirements, which concludes the DMTL stub design.



(a)  $|\,S_{11}|$  and  $|\,S_{21}|$  of DMTL UST-III in upstate (16 unit sections cascaded).



(b)  $|S_{11}|$  and  $|S_{21}|$  of DMTL UST-III in downstate (16 unit sections cascaded).



(c)  $\angle S_{21}$  of DMTL UST-III in both states (16 unit sections cascaded).



(d) The error between the S-parameters of the circuit model and the HFSS simulations of DMTL UST-III for all cases.

Figure 5.16: The comparison of the S-parameters of the circuit model and HFSS simulations for the DMTL UST-III. The graphs are the results of cascaded 16 unit sections both for the circuit model and the HFSS simulation results (a)-(c) Magnitudes and phases of the S-parameters (d) The error between the S-parameters of the circuit model and the HFSS simulations for all cases.



Figure 5.17: The  $S_{11}$  characteristics of the DMTL stub design for all possible 512 states.

# 5.2.4. The Overall RF MEMS Phase Shifter Design

The final step in the design of the RF MEMS PS is the connection of all the separately designed components, and to make final simulations in order to verify its operation. Figure 5.18 shows the layout of the final design of the RF MEMS PS. The design has  $3 \times 9$  control pins the stubs and 5 control pins that control both DMTL interconnection lines, which make a total of 32 control pins. Each DMTL stub has  $2^9 = 512$  different states and DMTL interconnection lines have  $2^5 = 32$  states, which are controlled by the same 5 control pins. The layout of the design measures 10750 µm × 5915 µm, which includes the TRL extensions.



Figure 5.18: The layout of the final RF MEMS PS design.

For the final simulations of the RF MEMS PS, the ABCD parameters of the separately designed blocks, which are the T-junction, the DMTL interconnection line, and the DMTL stubs, are cascaded in MWO. For each insertion phase state, the required electrical lengths of the DMTL inter connection lines and the DMTL stubs are found

using the method explained in the previous section, and the states that give the closest electrical lengths to the required value is selected for that insertion phase state. Alternatively, the states for the DMTL stubs and the DMTL interconnection lines are found using the optimization tools of MWO. In the optimization, three state variables for the DMTL stubs and one state variable for the DMTL interconnection lines are used, which make a total of four variables. Optimization gives a state set among the 2<sup>32</sup> state sets where the insertion loss and the input return loss performance are better for most of the insertion phase states.

The simulations results of the RF MEMS PS are shown in Figure 5.19, and the corresponding data is given in Table 5.7. In the final simulations, 0-360° insertion phase range is obtained with 10° steps. The average phase error is 0.64°, the average insertion loss is -2 dB, and the average return loss is better than -24 dB among the 36 insertion phase states. The maximum phase error is 4.8°, and there are a total of only 4 out of 36 that have phase errors worse than 1°. The worst case insertion loss is -3.9 dB, and there are a total of only 4 out of 36 states that have insertion losses worse than -3 dB. The worst case return loss is -10.9 dB, and there are a total of only 7 out of 36 states that have return losses worse than -20 dB. The performance parameters, i.e., the phase error, the insertion loss, and the return loss, degrade around some specific phase states, namely 110° and -40°. The reason of having low performance peaks at those specific points is due to the selection of the electrical length of the interconnection lines. These phase states are actually the transition regions, i.e., the maximum and minimum acceptable phase states allowed with the maximum and minimum states of the DMTL interconnection line. There are some differences between the theoretically required electrical length of the interconnection line and the electrical length of the final DMTL interconnection line design. Since the PS performance is very sensitive to the changes in the electrical length of the interconnection line, the performance of the PS degrade in a few insertion phase states around these transition regions.



Figure 5.19: The  $S_{21}$  performance of the designed RF MEMS PS for all 36 phase states.

It should be also pointed out here that  $10^{\circ}$  phase resolution is not the lower limit for the design. The designed RF MEMS PS can be used with a resolution of  $1^{\circ}$  with  $\pm 1^{\circ}$  maximum phase error at more than 90% of the 0-360 ° range.

The frequency response is very important, and it deserves detailed investigation. Since the designed PS is a distributed structure, it has a comparably narrow instantaneous bandwidth; however, the design is completely reconfigurable, and the center frequency can be changed while preserving the instantaneous bandwidth. For 0-360° insertion phase coverage, center frequency should be higher than the initial design frequency. Figure 5.20 shows that the performance of the designed PS is acceptable in 1 GHz bandwidth around the center frequency, which is 15 GHz. The performance in higher frequencies is investigated in the following subsections where the measurements results are given.

Target	S <sub>11</sub>	S <sub>21</sub>		Phase	Stub1	Stub2	Stub3	Int. Line
Phase	Mag (dB)	Mag (dB)	Ang (°)	Error (°)	State	State	State	State
0	-24.08	-1.29	0.05	0.05	352	121	250	1
10	-26.76	-1.04	9.97	0.03	354	224	452	0
20	-27.20	-0.91	20.03	0.03	360	484	224	0
30	-21.19	-0.98	29.94	0.06	304	450	80	0
40	-24.16	-1.10	39.82	0.18	60	489	40	0
50	-23.32	-1.18	49.17	0.83	40	257	20	0
60	-25.97	-1.33	59.62	0.38	20	227	8	0
70	-24.02	-1.45	70.39	0.39	14	341	4	0
80	-17.94	-1.43	79.10	0.90	8	496	0	0
90	-14.21	-1.60	88.99	1.01	0	25	0	0
100	-12.26	-1.76	95.20	4.80	0	0	0	0
110	-15.47	-3.93	110.40	0.40	419	503	491	31
120	-21.90	-3.46	120.90	0.90	419	503	483	31
130	-24.22	-3.27	130.00	0.00	501	487	419	31
140	-22.81	-3.10	140.40	0.40	505	507	499	31
150	-31.55	-2.97	150.40	0.40	425	375	499	31
160	-26.71	-2.70	160.50	0.50	373	499	501	31
170	-34.29	-2.53	170.20	0.20	481	501	501	31
-180	-24.90	-2.44	-179.30	0.70	393	485	355	30
-170	-39.09	-2.26	-169.60	0.40	417	357	457	31
-160	-23.58	-2.18	-160.10	0.10	417	357	457	29
-150	-41.84	-2.07	-150.20	0.20	353	357	481	30
-140	-37.75	-2.01	-139.40	0.60	322	481	417	29
-130	-28.67	-2.04	-129.50	0.50	504	481	193	30
-120	-37.03	-2.05	-120.30	0.30	496	481	456	29
-110	-23.62	-2.17	-109.60	0.40	352	433	480	29
-100	-35.43	-1.97	-99.80	0.20	185	448	505	22
-90	-30.88	-1.98	-89.80	0.20	320	492	371	20
-80	-24.80	-1.96	-80.75	0.75	96	505	481	20
-70	-21.90	-2.24	-70.58	0.58	48	417	481	19
-60	-13.80	-2.46	-61.70	1.70	16	481	417	19
-50	-10.89	-2.49	-52.67	2.67	240	499	80	8
-40	-13.63	-1.72	-38.70	1.30	256	481	393	4
-30	-14.45	-1.48	-30.10	0.10	112	482	481	4
-20	-20.00	-1.71	-19.18	0.82	496	192	485	2
-10	-33.24	-1.53	-10.08	0.08	73	204	497	2

Table 5.7: The final simulation results of the RF MEMS PS design for 0-360° range with 10° steps.

Avr.	-24.82	-2.02	0.64
Std. dev.	7.95	0.72	0.89



Figure 5.20: The performance of the designed RF MEMS PS for all 36 phase states (a)  $\angle S_{21}$  (unwrapped) (b)  $|S_{21}|$  (c)  $|S_{11}|$ .

# 5.3. The Fabrication and the Measurement Results

The designed RF MEMS phase shifter is fabricated using the METU RF MEMS Fabrication Process, the details of which were given in Chapter 2. The photographs and the SEM pictures of the fabricated PS are given in Figure 5.21 and Figure 5.22, respectively. The area of the fabricated PS measured  $10.8 \times 5.9 \text{ mm}^2$ .

The fabricated phase shifters packaged using ceramic IC package with the same method given in Chapter 3. The package size of CSB03228 is  $1.6 \times 0.595$  in<sup>2</sup>. Figure 5.23 shows the photograph of the packaged RF MEMS PS. The packaged phase shifters are measured using Agilent 8361A network analyzer and Cascade Summit 9000 manual probe station. Cascade ACP-GSG-150 coplanar probes are used. The PS is measured using both TRL and SOLT calibration techniques. The TRL calibration kit is designed for the 5-20 GHz frequency range, and it is also fabricated on the same wafer with the PS. For the SOLT calibration, the impedance standard substrate of Cascade probes (101-190) is used, and the calibration frequency range is 10-40 GHz. Special PS samples are fabricated for SOLT and TRL calibrations separately. For these measurements,  $50 \Omega$ - $50 \Omega$  transitions are needed on both sides of the PS since the pitch spacing of the CPW probes are not compatible with that of the PS design. For this purpose,  $50 \mu$ m-long  $50 \Omega$ - $50 \Omega$  transition is designed and added to the layout of the PS.

The same MEMS switch is used in all of the unit sections of the PS, and it is characterized by measurements. The actuation voltage of that switch is measured to be 8 V; nevertheless, all the MEMS structures are measured with 20 V-peak bipolar waveform as explained in Chapter 3. The reason for using a higher peak waveform is to guarantee the contact between the MEMS bridge and the dielectric layer. Each DMTL type is also fabricated separately to extract each parameter in different DMTL circuit models. The next subsection gives the details of the DMTL parameter extraction.



Figure 5.21: The photographs of the fabricated RF MEMS PS.



(a) Wide view of the DMTL interconnection line and the DMTL stub.



(b) The view of the of the DMTL interconnection line (UST I and II).







(e) The closer view of the of the MEMS bridge

Figure 5.22: The SEM pictures of the fabricated RF MEMS PS.



Figure 5.23: The photograph of the packaged RF MEMS PS (Package size:  $1.6 \times 0.595 \text{ in}^2$ ).

### 5.3.1. The DMTL Parameter Extraction

There are several parameters in the DMTL circuit model and specific structures are designed to extract each parameter of the fabricated RF MEMS PS. For the transmission line parameters, an unloaded CPW line is fabricated, and from the measurements of this line, the characteristic impedance, the relative dielectric constant,  $Z_o$ ,  $\mathcal{E}_{r,eff}$ , and lpha are extracted. For the parameters of the MEMS bridge, a DMTL which only includes the MEMS bridge without any MAM capacitors is fabricated and measured. From the upstate measurements,  $C_{bu}$  and  $R_{bu}$ , and from the downstate measurements,  $C_{bd}$ ,  $L_b$ , and  $R_{bd}$  are extracted. For the extraction of C<sub>s</sub>, DMTLs that only include the MAM capacitors without any MEMS bridges are fabricated for each unit section type. From the measurements of these DMTLs, which only have the upstate,  $C_s$  is extracted for each unit section type. Finally, DMTLs that only include a single unit section type are fabricated and measured. From the upstate and downstate measurements of these structures,  $C_{bu}$  and  $C_s$  are tuned to fit the measurement results of the final DMTL of each type (UST-Type I, II, and III); in addition to that, *R*<sub>bias</sub> is extracted. The extracted parameters for all types of unit sections are given in Table 5.8.

Considering the values given in the table above, most of the extracted values are close to the simulated values. The most important change is in the upstate capacitance of the MEMS bridge. This change is due to the change in the bridge height of the MEMS bridge on top of the CPW center conductor. The MEMS bridge is buckled downwards only on the center conductor of the CPW, and the remaining part of the MEMS bridge is totally flat. This can be observed from optical profiler measurements that are given in Figure 5.24-(a). The graphs and the views in Figure 5.24 are obtained from the measurements using Wyko NT 1100 optical profiler.

	DMTL UST-I		DMTL UST-II		DMTL UST-III	
	Sim.	Meas.	Sim.	Meas.	Sim.	Meas.
$Z_o(\Omega)$	96	96	96	96	96	96
$\mathcal{E}_{r,eff}$	2.36	2.43	2.36	2.43	2.36	2.43
$\alpha(dD/m)$	40	38	40	38	40	38
и (иБ/Ш)	@20 GHz	@20 GHz	@20 GHz	@20 GHz	@20 GHz	@20 GHz
$f_{Bragg}(GHz)$	73	77	59	62	50	54
s (µm)	300	300	300	300	210	210
$Z_{lu}(\Omega)$	55.9	54.2	52.9	51.6	44.1	39.7
$Z_{ld}(\Omega)$	40.9	42.3	34.5	35.2	23.7	22.3
Øup ( %)	-14.1	-14.8	-15	-16	-12.6	-13.9
Ødown()	-20	-18.7	-24	-22.9	-25.4	-23.5
$\Delta \phi$	5.9	3.9	9.1	6.9	12.8	9.6
C <sub>bu</sub> (fF)	46.4	61.5	46.7	62.3	46.3	67.7
$C_{bd}(fF)$	469.2	439.8	469.2	439.8	469.2	439.8
$C_s(fF)$	89.5	74.8	147.6	128.6	296.7	232.2
$L_b(pH)$	30	29.79	30	29.79	30	29.79
$R_{bu}/R_{bd}(\Omega)$	1.6/0.15	2 / 0.38	1.6/0.15	2/0.38	1.6/0.15	2/0.38
$C_U/C_D(fF)$	30.6 / 75.2	33.8 / 63.9	35.5 / 112.3	42 / 99.5	40.1 / 181.8	52.4 / 152
$Q_U/Q_D$	27 / 12	33 / 21	60 / 27	49 / 30	114 / 67	56 / 35
$R_{bias}(\Omega)$	1400	2718	1542	2120	1784	1300

Table 5.8: The circuit model parameters of the DMTL UST-I, II, and III that are extracted from the measurements.



(a) Measured 3D view optical profiler view of the center MEMS bridge of the RF MEMS PS.



(b) Measured beam height variation of the center MEMS bridge of the RF MEMS PS.



(c) Measured 3D view optical profiler view of the MAM capacitor of the RF MEMS PS.



(d) Measured beam height variation of the MAM capacitor of the RF MEMS PS.

Figure 5.24: The optical profiler measurements of the fabricated RF MEMS phase shifter.

The measured thickness is of the first metal layer that forms the CPW lines is 0.9  $\mu$ m, and the measured thickness of the structural layer is 1.1  $\mu$ m. Once the MEMS bridge passes over the CPW center conductor, the MEMS bridge height is also expected to increase by 0.9  $\mu$ m, and its top point is expected to be at 0.9 (first metal layer) + 1.1 (structural layer) + 1.8 (sacrificial layer) = 3.8  $\mu$ m. However, the top point is measured to be at 2.9  $\mu$ m, which means that the MEMS bridge is buckled downwards by 0.9  $\mu$ m (Figure 5.24-(b)).

The reason of the buckling is the weak anchoring at the CPW metal step height point as explained in Chapter 2 in detail. Since the surface topology sharply increases due to the 0.9  $\mu$ m-thick first metal layer, a secondary MEMS bridge occurs on top of the CPW center conductor, which is higher than the original bridge. However, this secondary bridge buckles downwards due to the tensile residual stress since the anchor points are not filled by strengthening layer, and they are nominally clamped. This is the behavior that was also observed in Chapter 2. The final MEMS bridge height is 2.9 - (0.9 + 1.1) = 0.9  $\mu$ m, where the 0.9  $\mu$ m and 1.1  $\mu$ m are the thicknesses of the first metal layer and the structural layer, respectively. This result is also verified with the simulation results where the center MEMS bridge is modeled at 0.9  $\mu$ m.

No buckling is observed on the MAM capacitances since there are no first metal layer step changes under these capacitances (Figure 5.24-(c)). The measured bridge height is exactly as expected, which is  $3.8 - (0.9 + 1.1) = 1.8 \mu m$  as presented in Figure 5.24-(d). This result is also verified with the simulation results where the MAM capacitances are modeled at  $1.8 \mu m$ .

The change in the bridge height of the MEMS bridges increases  $C_{bu}$  significantly, which causes the up-down capacitance ratio to reduce significantly. This also results with a lower phase difference between upstate and downstate. The consequences of these changes are explained in the next subsection.

# 5.3.2. The RF MEMS Phase Shifter Measurements

Having extracted the values of the circuit model parameters for each unit section type, the whole RF MEMS phase shifter is measured in 5-20 GHz and 10-30 or 10-40 GHz frequency bands with on-wafer TRL and SOLT calibration techniques, respectively. Since the circuit measured with SOLT calibration has additional 50  $\Omega$ -50  $\Omega$  CPW transitions on both sides, each measurement result is compared with its corresponding circuit.

For the measurement of the whole phase shifter, the multi-pin MEMS automated measurement setup (MAMS), which was explained in Chapter 3, is used since the RF MEMS PS has 32 control pins that should be controlled simultaneously. The applied actuation voltage is 20 V-peak bipolar waveform, which is supplied by the waveform generator card.

Figure 5.25 and Figure 5.26 show the upstate measurements of the PS with TRL and SOLT calibrations, respectively. The measurements with TRL calibration show good agreement with the simulations, whereas the measurements with SOLT calibration show good agreement with the simulations up to 30 GHz. After 30 GHz, the measurements still follow the simulations with some acceptable shift in both amplitude and phase responses. The circuit parameters used in the simulations are extracted from the measurements of corresponding type of DMTL structures as explained in the previous subsection.

184



(a)  $|S_{11}|$  of measured RF MEMS PS compared with the simulations.



(b)  $|S_{21}|$  of measured RF MEMS PS compared with the simulations.



(c)  $\angle S_{21}$  of measured RF MEMS PS compared with the simulations.

Figure 5.25: The measured upstate performance of the RF MEMS PS (TRL calibration used) compared with the simulations (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c)  $\angle S_{21}$ .



(a)  $|S_{11}|$  of measured RF MEMS PS compared with the simulations.



(b)  $|S_{21}|$  of measured RF MEMS PS compared with the simulations.



(c)  $\angle S_{21}$  of measured RF MEMS PS compared with the simulations.

Figure 5.26: The measured upstate performance of the RF MEMS PS (SOLT calibration used) compared with the simulations (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c)  $\angle S_{21}$ .

Figure 5.27 shows the performance of the PS configured for a sample phase state at the design center frequency, which is 15 GHz. The state numbers of the three stubs and the interconnection line are given in decimal form, and the ones in the corresponding binary equivalents indicate the unit sections in downstate. It is observed from the figure that the measurements are in good agreement with the simulations. The phase response is linear around the targeted frequency for at least  $\pm 1$  GHz band. At 15 GHz, return loss is better than -15 dB, the insertion loss is about -3 dB, and both of them stay in acceptable levels in  $\pm 0.5$  GHz. Figure 5.29, Figure 5.30, and Figure 5.31 show that the performance is similar for some other phase states even though they are not optimized specifically.

Figure 5.28 shows the performance of the PS for all possible phase states. From the figure, it is observed that the phase shift performance of the PS is limited to 160-360° range. The average phase error among these 21 states (160-360°) is 1.7°. The average insertion loss and the return loss of the measured 21 states are -3.1 dB and -19.3 dB with standard deviations of 0.6 dB and 6 dB, respectively.

It should be emphasized once more here that the 10° phase resolution is only a demonstrator, and the performance is not limited to this value. The figure clearly demonstrates that 5° resolution is possible, and special effort is not spent to obtain measured the phase values for the gaps in the figure. The author truly believes that the whole insertion range of the PS can be covered with about 2° or 3° steps with a careful search among all possible states using optimization tools.

The reason of having a reduced insertion phase range is the change in the MEMS bridge heights, which resulted in a significant increase in  $C_{bu}$ , and hence significant changes in  $\phi_{up}$  and  $\Delta \phi$ . The change in  $\phi_{up}$  increases the minimum insertion phase of the DMTL interconnection lines and the DMTL stubs where the insertion phase is a vital parameter for the phase shift performance as explained previously in this chapter. The reduction in  $\Delta \phi$  reduces the maximum insertion phase that can be

provided by the DMTL interconnection lines and the DMTL stubs, which is the second important parameter for the insertion phase performance of the PS. As a result of these changes, the electrical lengths that are necessary to obtain some insertion phase states of the phase shifter cannot be provided from the interconnection line and stubs DMTLs, which results with a reduced insertion phase range.



(a)  $|S_{11}|$  of measured RF MEMS PS compared with the simulations.



(b)  $|S_{21}|$  of measured RF MEMS PS compared with the simulations.



(c)  $\angle S_{21}$  of measured RF MEMS PS compared with the simulations.

Figure 5.27: The measured performance of the RF MEMS PS (TRL calibration used) that is reconfigured to -90° @ 15 GHz compared with the simulations (Stb1 = 5, Stb2 = 235, Stb3 = 501, IntLine = 19) (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c)  $\angle S_{21}$ .



Figure 5.28: The  $S_{21}$  performance of the designed RF MEMS PS for all 36 phase states.
Although the insertion loss performance of the PS can be estimated once the circuit model parameters are extracted from the dedicated measurements, there occurred some difference between the initial design simulations and the measurements. The insertion loss of the PS was calculated as -2 dB in average among all 36 states during the design; however, the insertion loss is measured as -3.1 dB in average among the measured 21 states. This increase in the insertion loss is mainly due to the differences in the overall quality factor of MEMS and MAM capacitors between the simulations and measurements. These differences depend upon three main reasons. The first one is the EM simulation environment. It was observed in several different cases that HFSS underestimates the conductor losses in microstrip or CPW based structures, which results in a lower simulated insertion loss. This can easily be observed from the  $R_{bu}$  and  $R_{bd}$  values given in Table 5.8 that are extracted from the HFSS simulations and the measurements. During the extraction of  $R_{bu}$  and  $R_{bd}$ , a DMTL structure that is composed of only MEMS bridges at the center without any MAM capacitances as explained above; hence, the following effects, which will be explained next, are excluded. The second reason is due to the changes in the bridge heights of the MEMS switches and MAM capacitors. The variations in the bridge heights totally change the RF current distributions. These changes directly affect the coupling to the bias line, and hence, the values of  $R_{bias}$  resistors, which is the dominant mechanism of losses in the DMTL unit sections in this thesis. The final reason is the increase in  $R_{bu}$  and  $R_{bd}$  because of the material properties in the fabrication process. The conductivity of the second metal layer which is used to implement MEMS bridges has a lower conductivity than the first (CPW) metal layer. This is because the second metal layer has to have very low residual stress, and the deposition conditions were optimized to reduce the residual stress as much as possible. The conductivity of this layer decreases as a consequence, which, in return, results with increase in  $R_{bu}$  and  $R_{bd}$ . Moreover, the conductivity of the electroplated gold layer that is used to fill the anchors of the MEMS and MAM capacitances is much lower than that of the sputtered gold layer that is used for the first metal layer, which is another reason for the increase in  $R_{bu}$  and  $R_{bd}$ .

One point that should be added to the loss is discussion is that the loss contributions of the T-junctions and the 50  $\Omega$ -to-50  $\Omega$  CPW transitions, which are only composed of gold metal layers. These losses are also underestimated in the EM simulations. Considering that there are three T-junctions and two transitions (SOLT version) in the PS, small differences between the measured and the simulated values are multiplied by a factor of 3 and 2, respectively.

Although the phase shifter is designed at a center frequency of 15 GHz and modeled for 5-20 GHz band, it is also possible to operate the PS at higher frequencies. It was already shown in Figure 5.26 that the upstate performance of the PS was in good agreement up to 30 GHz and acceptable up to 40 GHz. Figure 5.29 and Figure 5.30 also show that the performance is quite good and in agreement with the simulations at 22.5 GHz and 30 GHz, respectively. These two frequencies are selected for demonstrative purpose, and the PS can work *at any center frequency* between 15 GHz and 30 GHz; and, the phase ranges and the insertion losses increase gradually from their corresponding levels at 15 GHz to their corresponding levels at 30 GHz.



(a)  $|S_{11}|$  of measured RF MEMS PS compared with the simulations.



(b)  $|S_{21}|$  of measured RF MEMS PS compared with the simulations.



(c)  $\angle S_{21}$  of measured RF MEMS PS compared with the simulations.

Figure 5.29: The measured performance of the RF MEMS PS (SOLT calibration used) that is reconfigured to 90° @ 22.5 GHz compared with the simulations (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c)  $\angle S_{21}$ .



(a)  $|S_{11}|$  of measured RF MEMS PS compared with the simulations.



(b)  $|S_{21}|$  of measured RF MEMS PS compared with the simulations.



(c)  $\angle S_{21}$  of measured RF MEMS PS compared with the simulations.

Figure 5.30: The measured performance of the RF MEMS PS (SOLT calibration used) that is reconfigured to 180° @ 30 GHz compared with the simulations (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c)  $\angle S_{21}$ .

Figure 5.31 also shows the PS performance that is reconfigured for a specific phase state at 40 GHz. Here in this range, the return loss levels stay in an acceptable range around -10 dB; however, the insertion loss starts to deviate from the extracted simulations whereas the insertion phase performance shows an almost constant offset at each frequency. Nonetheless, it is possible to use the PS between 30 GHz up to 40 GHz using the reconfigurability of the design in spite of the shifted performance. Here, as the offset for each frequency is almost constant, the desired insertion phase value is obtained by reconfiguring the design to the desired insertion phase value plus the phase offset at that frequency. Figure 5.32 shows the measured S<sub>21</sub> performance of the RF MEMS PS at 22.5 GHz, 25 GHz, 27.5 GHz, 30 GHz, 35 GHz, and 40 GHz frequencies. Again, these frequencies are selected as some sample points for demonstrative purposes to show that the PS can work on each frequency in the band of interest. As can be seen from the graphs in Figure 5.32, the insertion phase range gradually increases up to 25 GHz, which is expected since the electrical lengths of the DMTL stubs increase as the frequency increases. The insertion phase range is covered with maximum 5° steps, which also means that the phase resolution is still not lost. However, the insertion phase range ceases to increase and stays about the same level up to 30 GHz while some gaps within some parts of the insertion phase range starts to occur. Finally, the insertion phase range starts to decrease after 30 GHz, and the gaps become more evident. The average insertion losses increase gradually up to 30 GHz; moreover, they increase more rapidly after 30 GHz as expected. The performance of the RF MEMS PS for the higher frequency operation is summarized in Table 5.9.



(a)  $|S_{11}|$  of measured RF MEMS PS compared with the simulations.



(b)  $|S_{21}|$  of measured RF MEMS PS compared with the simulations.



(c)  $\angle S_{21}$  of measured RF MEMS PS compared with the simulations.

Figure 5.31: The measured performance of the RF MEMS PS (SOLT calibration used) that is reconfigured to 0° @ 40 GHz compared with the simulations (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c)  $\angle S_{21}$ .









Figure 5.32: The measured  $S_{21}$  performance of the RF MEMS PS (SOLT calibration used) (a) 22.5 GHz (b) 25 GHz (c) 27.5 GHz (d) 30 GHz (e) 35 GHz (f) 40 GHz.

	Return	Loss (dB)	Insertion Loss (dB)		Insertion Phase	Max.
	Mean	Std. Dev.	Mean	Std. Dev.	Range	Phase Shift
15 GHz	-19.3	6.20	-3.1	0.65	160-360°	200°
22.5 GHz	-18.7	6.26	-4.1	0.38	0-215° ∪ 305-360°	270°
25 GHz	-18.2	4.00	-4.8	0.72	0-155° ∪ 185-360°	330°
27.5 GHz	-16.9	4.63	-4.7	0.90	0-090° ∪ 145-360°	305°
30 GHz	-15.3	3.58	-5.0	0.70	5-305° ∪ 355-360°	305°
35 GHz	-14.6	3.72	-7.8	0.95	0-165° ∪ 225-360°	300°
40 GHz	-13.9	3.47	-8.0	0.76	0-80° ∪ 150-360°	290°

Table 5.9: The summary of measured performance of the RF MEMS phase shifter.

The reduction in the phase shifter performance is understood better if the performance of the DMTL stubs is investigated in more detail in the band of interest. Figure 5.33 shows the simulated  $S_{11}$  performance of the DMTL stub using the circuit parameters that are extracted from the dedicated measurements as explained previously. It is observed in Figure 5.33-(a) that the  $S_{11}$  of the DMTL stub goes very close to the Smith Chart periphery, which is required for the proper phase shifter operation. However, as the frequency increases, the S<sub>11</sub> points commence not only to move away from the periphery, which, in case, means that a considerable real part is added to the DMTL stub reflection coefficient, but also the phase difference between the points commence to increase. This situation becomes even more dramatic as the frequency is greater than 30 GHz, which can be seen in Figure 5.33-(c) and (d). As a result, the DMTL stubs cannot provide the necessary reflection coefficient; and hence, the insertion losses increase significantly, and some insertion phase values cannot be obtained. The distortion in the S<sub>11</sub> performance of the DMTL stub because the losses increase notably as the operation frequency increases, and more importantly, the operation frequency gets closer to the Bragg frequency of DMTL UST-III, which was extracted as 54 GHz. It should be pointed out here that for a proper DMTL operation, the Bragg frequency is usually selected at least more than two times of the design frequency. In this case, the operation frequency becomes only 1.8 and 1.35 times of the Bragg frequency for 30 GHz and 40 GHz, respectively.



Figure 5.33: The simulated  $S_{11}$  characteristics of the DMTL stub design for all possible 512 states with the extracted circuit model parameters at some measurement frequencies (a) 15 GHz (b) 25 GHz (c) 30 GHz (d) 40 GHz.

# 5.4. Conclusion

In this chapter, the design, the fabrication, and the measurement results of the RF MEMS phase shifter that is based on the TST. The measurement results show good agreement with the simulations. The PS has good phase error, insertion loss, and return loss performance at the center design frequency, 15 GHz; however, the insertion phase range decreased due to reduced MEMS bridge height. The PS can also work frequencies up to 40 GHz with gradually increasing insertion phase ranges and insertion losses using the reconfigurability of the design. The phase errors that occur at any insertion phase state can be easily overcome using again the reconfigurability of the design for the frequencies up to 40 GHz.

# **CHAPTER 6**

# RF MEMS VECTOR MODULATOR USING TRIPLE STUB TOPOLOGY

# 6.1. Introduction

This chapter presents a novel RF MEMS vector modulator that is based on triple stub topology (TST). The vector modulator (VM) is uses the same idea with the RF MEMS phase shifter that was presented in the previous chapter; moreover, the amplitude of the input signal is also controlled with the VM. The theory of simultaneous phase and amplitude control was explained in Chapter 4. As explained previously, the theoretical approach results with a range of interconnection line lengths instead of a unique solution for a given insertion phase value. In addition to the phase control, the amplitude of the signal is controlled by selecting the interconnection line length such that the structure shows a resonating behavior between its open-circuited terminations at the end of the stubs. By tuning the interconnection line length and changing the "distance" to the resonance point, the amplitude of the input signal can be controlled simultaneously with the insertion phase while the input return loss is theoretically kept as 0.

In this chapter, first, the design of the RF MEMS vector modulator is presented that explains the design of each part of the circuit. Following the design, the results of the fabrication process of the designed VM are presented. Then, the measurement results of the VM are given. Finally, the measurement results and the performance of the VM are discussed, which is followed by the chapter conclusion.

#### 6.2. The Design of the RF MEMS Vector Modulator

The VM design basically uses the same topology that was presented in Figure 5.2. The electrical lengths of the stubs and the interconnection lines are adjusted by means of the DMTLs. The methodology followed for the design of the VM is the same of that of the PS.

The VM design targets to cover 0-360 ° phase range with 45 ° resolution and to have three amplitude states, which are -2 dB, -5 dB, and -8 dB states; this means that the VM design targets a total of 24 insertion phase/amplitude states. The -2 dB is the average insertion loss of the VM at 15 GHz; so, the minimum amplitude states are set to this value in order to minimize error between the design and the measurements.

The VM design has again three types of parts, which are the T-junction, the DMTL interconnection lines, and the DMTL stubs. The T-junctions in the VM are the ones that are used in the PS design since the T-junction that is used in the PS design is a general purpose design, and it also satisfies the requirements of the VM design.

For the DMTL interconnection line, the same DMTL design that is used in the PS is used. The reason for the design constraints was explained in Chapter 5.2.2. It should be kept in mind that the VM is also a 0-360° phase shifter in the minimum insertion loss states, and the current DMTL interconnection line satisfies the requirement of the phase shifter design. This DMTL interconnection line can also be used for the VM design. The reasoning is explained as follows: The current DMTL interconnection line is designed to cover 0 to  $0.1\lambda$  and  $0.4\lambda$  to  $0.5\lambda$  electrical length range, which was necessary for the phase shifter to be able to cover  $\pm 40^{\circ}$  insertion phase range. For the remaining phase range, either  $0.1\lambda$  or  $0.4\lambda$  electrical length states of the DMTL interconnection line is used (please see Figure 5.7), where the insertion loss for the remaining states are more than their minimum values in this

case. Once the DMTL interconnection line length goes from  $0.1\lambda$  to 0 or  $0.4\lambda$  to  $0.5\lambda$ , the insertion losses in the remaining insertion phase states increase, which means that the amplitude can also be controlled for these states. As the electrical length of the DMTL interconnection line changes in small steps in 0 to  $0.1\lambda$  and  $0.4\lambda$  to  $0.5\lambda$  range, the amplitude of the remaining insertion phase states can also be controlled precisely.

The requirements of the DMTL stubs are, however, totally different from that of the PS, and the DMTL stub is designed considering the specific requirements of the VM design as explained in the following subsection.

#### 6.2.1. The DMTL stub Design

The design of the DMTL stub of the VM is very similar to that of the PS, which is targeted to obtain a one port circuit with constant amplitude and varying phase of its reflection coefficient. The reader is referred to Chapter 5.2.3 for the details of the design constraints and the methodology followed during the design.

The design targets of the DMTL stub of the VM are found by optimization since the analytical solution does not give explicit equations for the lossy case as explained in Chapter 4. However, the initial points for the optimization are found using (4.7) to (4.10), which is critical to find the solution. According to the results of the optimization, the phases of reflection coefficients should be in ranges -120° to 150°, -173° to 180°, and -137° to 139° for Stub1, Stub2, and Stub3, respectively. After this step, the DMTL interconnection and the T-junctions are inserted to the circuit, and the optimization is repeated to see whether the requirements are still the same. It was found out that some ranges are not necessary for all three stubs. The only common unnecessary range is 0° to 70° for all of the stubs. In order to simplify the VM structure, a single DMTL stub has been designed that uses only one type of DMTL unit section, which is DMTL UST-IV. So, the design target for the phase range of the reflection coefficient is selected as 70° to 360° for the DMTL stub.

To obtain the required phase range, DMTL Unit Section Type IV is designed using (5.1)-(5.11) in Matlab, and to check if the electrical length of the stub is enough to cover the specified range, 9 UST-IVs are cascaded with the T-junction extension, and the phase of the reflection coefficient is checked. The maximum number of DMTL unit sections that can be controlled independently is 9 as explained in the previous section. Since the phase range cannot be covered easily with 9 DMTL unit sections, 4 additional DMTL unit sections are added in order to shift the points with minimum and maximum reflection coefficient phases. The final DMTL stub has 13 unit sections of UST-IV, the first four of which is held only in the upstate.

The DMTL UST-IV designed in Matlab is also simulated in HFSS v9.2 to verify and optimize the design. During the simulations, the same physical dimensions are used for the CPWs, and the physical parameters of the MEMS center capacitance,  $C_b$ , and the MAM capacitances,  $C_s$ , are extracted using capacitance formulation (3.1) to (3.3) given in Chapter 3 as explained previously. The values of the parameters used in the initial design with DMTL formulation are given in the first column of Table 6.1. The 3D view of the final DMTL UST-III design and its physical dimensions are given in Figure 6.1 and Table 6.2, respectively.

	DMTL	Circuit Model
	Formulation	HFSS Sim.
$Z_o$	96 Ω	96 Ω
$\mathcal{E}_{r,eff}$	2.36	2.36
~	40 dB/m	40 dB/m
α	@20 GHz	@20 GHz
$f_{Bragg}$	60 GHz	44 GHz
S	216 µm	210 µm
$Z_{lu}$	44.5 Ω	43.1 Ω
$Z_{ld}$	20 Ω	20.3 Ω
$\phi_{up}$	-12.9°	-12.9°
$\phi_{down}$	-28.6°	-28.3°
$\Delta \phi$	15.7°	15.4°
$C_{bu}$	50.4 fF	46.9 fF
$C_{bd}$	458.4 fF	469.2 fF
$C_s$	253.8 fF	471.5 fF
$L_b$	-	30 pH
$R_{bu}/R_{bd}$	-	1.6 Ω/0.15 Ω
$C_U/C_D$	42.1 fF/163.4 fF	42.7 fF/235.1 fF
$Q_U/Q_D$	-	131/99
R <sub>bias</sub>	-	1665 Ω
$R_{bias.DC}$	-	10.125 kΩ

Table 6.1: The circuit design parameters of the DMTL UST-IV.



Figure 6.1: 3D view of the DMTL UST-IV (z-dimension exaggerated).

Parameter	UST-IV
W	150 μm
G	150 μm
$l_{brd}$	300 µm
Wbrd	38 µm
<b>W</b> gnd	46 µm
$l_{brd-g}$	138 µm
W <sub>brd-g</sub>	6 x 48 μm
Wgo	30 µm
$l_{go}$	40 µm
WSiCr	20 µm

Table 6.2: The physical design parameters of the final design of the DMTL UST-IV.

Figure 6.2 gives the comparison of the HFSS simulations with the proposed circuit model simulations. For the comparison of the HFSS simulations and the circuit model, 16 unit section of UST-IV are cascaded and compared with the results of the 16 cascaded simulation results of the HFSS model. The circuit model parameters that are extracted from the HFSS simulations are given in the second column of Table 6.1. The simulation results again show that the model is successful in predicting the EM full-wave simulations, which is also quantitatively given in Figure 6.2-(d). Lastly, Figure 6.3 shows the S<sub>11</sub> characteristics of the final DMTL stub design for all possible  $2^9 = 512$  possible states, in which 4 + 9 UST-IVs are cascaded and terminated with an open circuit. The first 4 UST-IVs are only used in the upstate, so the number of the states of the DMTL stub does not change. The circuit that is used for the simulation of the DMTL stub also includes the transmission line that models the T-junction extension. The simulation results show that the circuit satisfies the design requirements of the RF MEMS vector modulator.



(a)  $|\,S_{11}|$  and  $|\,S_{21}|$  of DMTL UST-IV in upstate (16 unit sections cascaded).



(b)  $|S_{11}|$  and  $|S_{21}|$  of DMTL UST-IV in downstate (16 unit sections cascaded).



(c)  $\angle S_{21}$  of DMTL UST-IV in both states (16 unit sections cascaded).



(d) The error between the S-parameters of the circuit model and the HFSS simulations of DMTL UST-IV for all cases.

Figure 6.2: The comparison of the S-parameters of the circuit model and HFSS simulations for the DMTL UST-IV. The graphs are the results of cascaded 16 unit sections both for the circuit model and the HFSS simulation results (a)-(c) Magnitudes and phases of the S-parameters (d) The error between the S-parameters of the circuit model and the HFSS simulations for all cases.



Figure 6.3: The  $S_{11}$  characteristics of the DMTL stub design that is composed of 4 + 9 DMTL UST-IVs for all possible 512 states.

#### 6.2.2. The Overall RF MEMS Vector Modulator Design

Figure 6.4 shows the layout of the RF MEMS vector modulator. The design has 32 control pins, which has the same pin diagram with the RF MEMS PS. Each stub has 9 control pins, and two interconnection lines are controlled by 5 control pins. The layout of the design measures 9850  $\mu$ m × 6755  $\mu$ m<sup>2</sup>, including the 50  $\Omega$ -50  $\Omega$  transitions. These transitions are necessary in order to decrease the CPW pitch spacing as explained in Chapter 5.3.



Figure 6.4: The layout of the final RF MEMS VM design.

For the final simulation of the RF MEMS VM, the same methodology that is used during the design of the PS is followed. The method is based on cascading the ABCD parameter of the separately designed subparts, and the details were explained in Chapter 5. The states of the stubs and the interconnection lines for each insertion phase/amplitude state are found in two ways. The first one is to select the states that give the closest electrical lengths to the required value determined by the method explained in Chapter 4. The second way is to use the optimization method in which the states of the three stubs and the interconnection lines are used as the optimization variables.

The simulations results of the RF MEMS VM are shown in Figure 6.5, and the corresponding data is given in Table 6.3. The design targets of the VM are to cover 0-360° range with 45° steps and to have three amplitude states, which are -2 dB, -5 dB, and -8 dB. The design center frequency is 15 GHz. *The average phase errors is 1.31°, the average amplitude error is 0.08 dB, and the average return loss is -17.6 dB with standard deviations of 1°, 0.2 dB, and 5.1 dB, respectively for all 24 states.* The return loss is better than -10.6 dB for all 24 states. The amplitude error performance of the VM becomes worse only around -2 dB / 135° and -2 dB / 180° states, which is due to the selection of the interconnection line electrical length as explained in Chapter 5.

It should be also pointed out here that 45 ° phase and 3 dB amplitude resolutions are definitely not the lower limits for the VM design. The designed RF MEMS VM can easily be reconfigured to have a phase resolution of 5 ° with  $\pm$ 3 ° maximum phase error at the 0-360 ° range and an amplitude resolution of 0.2 dB with  $\pm$ 0.1 dB maximum amplitude error at -0.8 to -8 dB amplitude range and more than 80% of the 0-360 ° range. The amplitude range with minimum value less than -8 dB is also possible; however, the instantaneous bandwidth of the design drops rapidly as the minimum amplitude level is decreased.

Tar	get	S <sub>11</sub>	S <sub>2</sub>	21	Amp.	Phase	Stub1	Stub2	Stub3	Int. Line
Mag (dB)	Ang (°)	Mag (dB)	Mag (dB)	Ang (°)	Err. (dB)	Error (°)	State	State	State	State
-2	0	-18.6	-2.03	-1	0.03	1	319	27	0	2
-2	45	-12.5	-2.09	44.3	0.09	0.7	231	320	327	1
-2	90	-17.8	-1.98	89	0.02	1	459	463	451	0
-2	135	-23.2	-2.89	136.4	0.89	1.4	200	244	320	31
-2	180	-31.1	-2.29	178.7	0.29	1.3	48	384	400	27
-2	-135	-19.5	-1.98	-134.7	0.02	0.3	62	232	44	14
-2	-90	-19.1	-2	-89.9	0	0.1	2	511	100	21
-2	-45	-10.6	-2.07	-46.8	0.07	1.8	0	0	19	4

Table 6.3: The final simulation results of the RF MEMS VM design for 0-360° range with  $45^\circ$  steps and -2 to -8 dB amplitude range with 3 dB steps.

Avr. -19.05 -2.17 0.18 0.95

Tar	get	S <sub>11</sub>	S <sub>2</sub>	21	Amp.	Phase	Stub1	Stub2	Stub3	Int. Line
Mag (dB)	Ang (°)	Mag (dB)	Mag (dB)	Ang (°)	Err. (dB)	Error (°)	State	State	State	State
-5	0	-13.4	-4.97	-1.4	0.03	1.4	37	506	154	0
-5	45	-21.4	-5	45.7	0	0.7	111	242	175	1
-5	90	-23.1	-5	90.1	0	0.1	127	264	275	1
-5	135	-19.8	-5.12	135.3	0.12	0.3	7	86	35	31
-5	180	-12.6	-4.97	178.2	0.03	1.8	164	43	1	26
-5	-135	-22.6	-4.98	-135.2	0.02	0.2	65	442	376	11
-5	-90	-16.2	-5.01	-87.22	0.01	2.78	64	78	254	11
-5	-45	-12.2	-4.99	-41.5	0.01	3.5	387	81	336	3

Avr. -17.66 -5.01 0.03 1.35

Tar	get	S <sub>11</sub>	S <sub>2</sub>	1	Amp.	Phase	Stub1	Stub2	Stub3	Int. Line
Mag (dB)	Ang (°)	Mag (dB)	Mag (dB)	Ang (°)	Err. (dB)	Error (°)	State	State	State	State
-8	0	-16.8	-8.06	-0.7	0.06	0.7	22	476	220	0
-8	45	-13.5	-7.89	44.2	0.11	0.8	444	504	445	0
-8	90	-15.1	-7.98	93.2	0.02	3.2	271	499	190	1
-8	135	-11.5	-7.95	137.3	0.05	2.3	511	89	210	29
-8	180	-17.2	-8.03	-178.7	0.03	1.3	256	154	388	18
-8	-135	-22.5	-8.03	-136.6	0.03	1.6	54	403	388	16
-8	-90	-22	-7.98	-89.6	0.02	0.4	415	312	425	30
-8	-45	-11.1	-7.98	-47.7	0.02	2.7	444	43	203	17

	Avr.	-16.21	-7.99	0.04	1.63
Tota	l avr.	-17.64		0.08	1.31
Total s	td. dev.	5.07		0.18	0.99



Figure 6.5: The  $S_{21}$  performance of the designed RF MEMS VM for all 24 insertion phase/amplitude states.

The frequency response of the designed VM is given in Figure 6.6. The design has a narrow instantaneous bandwidth since it is based on distributed structures. It has a bandwidth of about 400 MHz for the highest amplitude states, which is -2 dB, for  $\pm 0.5$  dB amplitude error, and the performance is not limited by the phase, but the amplitude response. The bandwidth of the lowest amplitude states, which is -8 dB, is about 100 MHz for  $\pm 1$  dB amplitude error. The reason behind having such narrow bandwidths with the decreasing amplitude is due to the resonating behavior of the design. As the amplitude decreases, the design is operated more closely to the  $\lambda/2$  resonance point where the losses are maximum and the slope of the amplitude change very fast. However, it should be kept in mind that the design is completely reconfigurable, the center frequency can be shifted to any frequency that is higher than the initial design frequency, where the instantaneous bandwidth is preserved.









(c)  $|S_{11}|$  of RF MEMS VM for all -2 dB amplitude states.







(e)  $|S_{21}|$  of RF MEMS VM for all -5 dB amplitude states.













(i)  $|S_{11}|$  of RF MEMS VM for all -8 dB amplitude states.



Figure 6.6: The performance of the designed RF MEMS VM. The performance for all (a)-(b)-(c) - 2 dB (d)-(e)-(f) - 5 dB (g)-(h)-(i) - 8 dB amplitude states. (j) Amplitude linearity for all 24 states.

# 6.3. The Fabrication and the Measurement Results

The designed RF MEMS vector modulator is fabricated using the METU RF MEMS fabrication process, the details of which were given in Chapter 2. The photographs and the SEM pictures of the fabricated RF MEMS VM are given in Figure 6.7 and Figure 6.8, respectively. The RF MEMS VM measures 9850  $\mu$ m × 6755  $\mu$ m<sup>2</sup>.

The fabricated vector modulator is mounted on CSB03228 ceramic IC package with the same method given in Chapter 3. The package dimensions of CSB03228 are 1.6 in  $\times$  0.595 in. Figure 6.9 shows the photograph of the packaged RF MEMS VM. The fabricated vector modulator is then measured using the MAMS setup explained in Chapter 5.3. SOLT calibration technique is used for the measurements of the RF MEMS vector modulator in 10-30 GHz frequency bands. Additional 50  $\Omega$ -50  $\Omega$  CPW transitions are also used for the SOLT measurements on both sides the VM as explained in Chapter 5.3. The same MEMS switches are used in all types of DMTL unit sections; hence, the same actuation voltage is used in the test of the VM and all the other DMTL test structures. The actuation voltage of the MEMS switch is measured as 8 V, and all the MEMS structures are measured with 20 V bipolar waveform.

The DMTL UST-I and II, the T-junctions, and the transitions were characterized in the measurements of the RF MEMS phase shifter. So, the only remaining part in the measurement and characterization of the VM is the modeling and parameter extraction of the DMTL UST-IV, which is explained in the following subsection.

# 6.3.1. The DMTL Parameter Extraction

The characterization and the parameter extraction of DMTL UST-IV are made using the method applied during the measurements of the RF MEMS PS. In this method, dedicated DMTL structures are used for the extraction of DMTL circuit model parameters in groups. The dedicated DMTL structures are measured using the TRL calibration technique in 5-20 GHz frequency band. The details of the circuit model parameter extraction can be found in Chapter 5. The extracted circuit model parameters are given in Table 6.4, and the parameters are also compared with the simulated values within this table. Comparison of the measurement results of the DMTL, which is composed of 9 UST-IVs, with the circuit model simulations are given in Figure 6.10.



Figure 6.7: The photographs of the fabricated RF MEMS VM.



(a) Wide view of the DMTL interconnection line and the DMTL stub (UST-IV).



(b) The view of the of the DMTL stub (UST-IV).



(c) The closer view of the of the MEMS bridge and the MAM capacitor (UST-IV).

Figure 6.8: The SEM pictures of the fabricated RF MEMS VM.



Figure 6.9: The photograph of the packaged RF MEMS VM.

Sim.Meas. $Z_o(\Omega)$ 9696 $\varepsilon_{r,eff}$ 2.362.43 $\alpha(dB/m)$ 4038 $(dB/m)$ $(20 \text{ GHz})$ $(20 \text{ GHz})$ $\alpha(dB/m)$ 210210 $\alpha_{ragg}(GHz)$ 4447.5 $s(\mu m)$ 210210 $Z_{lu}(\Omega)$ 43.137.4 $Z_{ld}(\Omega)$ 20.321 $\phi_{up}(\circ)$ -12.9-14.5 $\phi_{down}(\circ)$ -28.3-27.2	
$ \begin{array}{c c c c c c c c c } Z_{o}(\varOmega) & 96 & 96 \\ \hline & \mathcal{E}_{r,eff} & 2.36 & 2.43 \\ \hline & \mathcal{L}_{a}(dB/m) & 40 & 38 \\ \hline & @20 \ GHz & @20 \ GHz \\ \hline & @20 \ GHz & @20 \ GHz \\ \hline & @20 \ GHz & 210 \\ \hline & & & & & & & \\ \hline & & & & & & & \\ \hline & & & &$	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$Z_o(\Omega)$
$\begin{array}{c} 40 & 38 \\ @20 \text{ GHz} & @20 \text{ GHz} \\ \hline \end{tabular} \\ \hline \$	$\mathcal{E}_{r,eff}$
$(ub/m)$ @20 GHz@20 GHz $aragg$ (GHz)4447.5 $s$ ( $\mu$ m)210210 $Z_{lu}$ ( $\Omega$ )43.137.4 $Z_{ld}$ ( $\Omega$ )20.321 $\phi_{up}$ (°)-12.9-14.5 $\phi_{down}$ (°)-28.3-27.2	$\alpha(dD/m)$
$p_{ragg}(GHz)$ 4447.5 $s(\mu m)$ 210210 $Z_{lu}(\Omega)$ 43.137.4 $Z_{ld}(\Omega)$ 20.321 $\phi_{up}(\circ)$ -12.9-14.5 $\phi_{down}(\circ)$ -28.3-27.2	и (и <i>Б</i> /т)
$s (\mu m)$ 210210 $Z_{lu}(\Omega)$ 43.137.4 $Z_{ld}(\Omega)$ 20.321 $\phi_{up}(\circ)$ -12.9-14.5 $\phi_{down}(\circ)$ -28.3-27.2	f <sub>Bragg</sub> (GHz)
$Z_{lu}(\Omega)$ 43.137.4 $Z_{ld}(\Omega)$ 20.321 $\phi_{up}(?)$ -12.9-14.5 $\phi_{down}(?)$ -28.3-27.2	s (µm)
$Z_{ld}(\Omega)$ 20.3 21 $\phi_{up}(\circ)$ -12.9 -14.5 $\phi_{down}(\circ)$ -28.3 -27.2	$Z_{lu}(\Omega)$
$\phi_{up}$ (°) -12.9 -14.5 $\phi_{down}$ (°) -28.3 -27.2	$Z_{ld}(\Omega)$
<i>φ</i> <sub>down</sub> ( <i>°</i> ) -28.3 -27.2	Øup ( )
	Ødown( )
$\Delta \phi$ 15.4 12.7	$\Delta \phi$
<i>C<sub>bu</sub>(fF)</i> 46.9 67.7	$C_{bu}(fF)$
<i>C<sub>bd</sub> (fF)</i> 469.2 439.8	$C_{bd}(fF)$
<i>C<sub>s</sub>(fF)</i> 471.5 363	$C_s(fF)$
<i>L<sub>b</sub>(pH)</i> 30 29.79	$L_b(pH)$
$_{bu}/R_{bd}(\Omega)$ 1.6/0.15 2/0.38	$R_{bu}/R_{bd}(\Omega)$
$C_U/C_D(fF)$ 42.7/235.1 57.1/198.9	$C_U/C_D(fF)$
<i>Q<sub>U</sub>/Q<sub>D</sub></i> 131/99 69/49	$Q_U/Q_D$
<i>R</i> <sub>bias</sub> (Ω) 1665 1215	$R_{bias}(\Omega)$

Table 6.4: The circuit model parameters of the DMTL UST-IV that are extracted from the measurements.

Considering the values given in the table above, most of the extracted values are close to the simulated values. The most import changes are the changes in the upstate capacitance of the MEMS bridge and MAM capacitance,  $C_{bu}$  and  $C_s$ , and the reasons behind this behavior was explained in detail in Chapter 5.3. The optical surface profiler measurements of the RF MEMS vector modulator is given in Figure 6.11. The figure shows that the bridge height of the MAM capacitances is 1.8 µm, which is the same as that of the MAM capacitances used in the RF MEMS phase shifter.



(a)  $|S_{11}|$  and  $|S_{21}|$  of DMTL UST-IV in upstate (9 unit sections cascaded).



(b)  $|S_{11}|$  and  $|S_{21}|$  of DMTL UST-IV in downstate (9 unit sections cascaded).



(c)  $\angle S_{21}$  of DMTL UST-IV in both states (9 unit sections cascaded).



(d) The error between the S-parameters of the circuit model and measurements of DMTL UST-IV for all cases (9 unit sections cascaded).

Figure 6.10: The comparison of the measured S-parameters and the circuit model simulations for the 9 cascaded UST-IVs. (a) Magnitudes of the S-parameters in the upstate (b) Magnitudes of the S-parameters in the downstate (c) The phases of the S-parameters for both states (d) The error between the S-parameters of the measurements and the circuit model for both states.

### 6.3.2. The RF MEMS Vector Modulator Measurements

The RF MEMS vector modulator is measured using SOLT calibration techniques 10-30 GHz frequency band. Since the VM that is used for the SOLT measurements has additional  $50 \Omega - 50 \Omega$  CPW transitions on both sides, the effects of these transitions are also included in the simulations. The multi-pin MEMS automated measurement setup (MAMS), which was explained in Chapter 3, is used for the measurements as the RF MEMS VM has 32 control pins that should be controlled simultaneously. The applied actuation voltage is 20 V bipolar waveform, which is supplied by the waveform generator card.



Figure 6.11: The optical profiler measurements of the fabricated RF MEMS vector modulator.

Figure 6.12 shows the upstate measurement of the RF MEMS VM that is measured using SOLT calibration technique. The measurements are in good agreement with the simulations up to the vicinity of 30 GHz, where the measured response starts to deviate from the simulations. The circuit parameters that are used in the simulations are extracted from the measurements of each type of DMTL structures, which were given previously in Table 5.8 and Table 6.4.


(a)  $|\,S_{11}|$  of measured RF MEMS PS compared with the simulations.



(b)  $|S_{21}|$  of measured RF MEMS PS compared with the simulations.



(c)  $\angle S_{21}$  of measured RF MEMS PS compared with the simulations.

Figure 6.12: The measured upstate performance of the RF MEMS VM (SOLT calibration used) compared with the simulations (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c)  $\angle S_{21}$ .

Figure 6.13, Figure 6.14, Figure 6.15, and Figure 6.16 show the measured performance of the VM for some sample insertion phase/amplitude states at 15 GHz, 22.5 GHz, 25 GHz, and 27.5 GHz, respectively. The state numbers of the three stubs and the interconnection line are given in decimal form, and the ones in the binary equivalent of each state number show that those unit sections are in downstate. The figures show that the measurements are in good agreement with the simulations when the VM is reconfigured at the frequencies mentioned above. The phase response is linear around the targeted frequencies for minimum ±1 GHz band. The return losses are better than -11 dB, and they stay in acceptable levels in 1 GHz band around the targeted frequencies. The insertion losses can also be guessed by the simulations by using the resistance values extracted from the specific DMTL measurements. The amplitude response is the bottleneck for the bandwidth of the RF MEMS VM, and the instantaneous bandwidth can go down to 150 MHz around the center reconfiguration frequency for some low amplitude states. However, since the VM is completely reconfigurable, the frequency of operation can be easily tuned to the desired value so that the center of the instantaneous bandwidth is centered to the frequency of operation.



(a)  $|S_{11}|$  of measured RF MEMS PS compared with the simulations.



(b)  $|S_{21}|$  of measured RF MEMS PS compared with the simulations.



(c)  $\angle S_{21}$  of measured RF MEMS PS compared with the simulations.

Figure 6.13: The measured performance of the RF MEMS VM (SOLT calibration used) that is reconfigured to (-2 dB, -90°) @ 15 GHz compared with the simulations (Stb1 = 0, Stb2 = 0, Stb3 = 0, IntLine = 8) (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c)  $\angle S_{21}$ .



(a)  $|S_{11}|$  of measured RF MEMS PS compared with the simulations.



(b)  $|S_{21}|$  of measured RF MEMS PS compared with the simulations.



(c)  $\angle S_{21}$  of measured RF MEMS PS compared with the simulations.

Figure 6.14: The measured performance of the RF MEMS VM (SOLT calibration used) that is reconfigured to (-5 dB, -135°) @ 22.5 GHz compared with the simulations (Stb1 = 32, Stb2 = 1, Stb3 = 0, IntLine = 0) (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c)  $\angle S_{21}$ .



(a)  $|S_{11}|$  of measured RF MEMS PS compared with the simulations.



(b)  $|S_{21}|$  of measured RF MEMS PS compared with the simulations.



(c)  $\angle S_{21}$  of measured RF MEMS PS compared with the simulations.

Figure 6.15: The measured performance of the RF MEMS VM (SOLT calibration used) that is reconfigured to (-8 dB, -45°) @ 25 GHz compared with the simulations (Stb1 = 192, Stb2 = 463, Stb3 = 59, IntLine = 24) (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c)  $\angle S_{21}$ .



(a)  $|S_{11}|$  of measured RF MEMS PS compared with the simulations.



(b)  $|S_{21}|$  of measured RF MEMS PS compared with the simulations.



(c)  $\angle S_{21}$  of measured RF MEMS PS compared with the simulations.

Figure 6.16: The measured performance of the RF MEMS VM (SOLT calibration used) that is reconfigured to (-6 dB, -135°) @ 27.5 GHz compared with the simulations (Stb1 = 2, Stb2 = 119, Stb3 = 401, IntLine = 18) (a)  $|S_{11}|$  (b)  $|S_{21}|$  (c)  $\angle S_{21}$ .

Amplitude States			Avr. /	S <sub>11</sub> S <sub>21</sub>		21	Amp.	Amp.	Phase
		de States	Std. Dev.	dB	dB	Lin	Error dB	Error	Error (°)
	State	Set 1	Avr.	-18.69	-3.10	0.70	0.70	0.06	4.03
N	-3.10 dB /	45° steps	Std. Dev.	9.87	0.91	0.07	0.52	0.04	5.56
Ï	State Set 2		Avr.	-16.36	-5.17	0.55	0.16	0.01	2.92
Ċ	-5.19 dB / 45° steps		Std. Dev.	7.26	0.21	0.01	0.11	0.01	4.39
<b>L</b>	State	Set 3	Avr.	-13.53	-7.56	0.42	0.71	0.03	4.94
2	-7.96 dB /	45° steps	Std. Dev.	6.41	0.84	0.04	0.55	0.03	4.16
2	Overall 24	A	vr.	-16.19	-5.27	0.56	0.52	0.03	3.96
	states	Std.	Dev.	7.93	1.99	0.13	0.50	0.03	4.61
	State	Set 1	Avr.	-20.37	-3.45	0.67	0.52	0.04	3.24
	-3.10 dB /	45° steps	Std. Dev.	5.67	0.73	0.05	0.61	0.04	3.62
4	State	Set 2	Avr.	-18.14	-5.17	0.55	0.18	0.01	3.83
Ġ	-5.19 dB /	45° steps	Std. Dev.	6.19	0.23	0.01	0.12	0.01	3.54
ы́	State	Set 3	Avr.	-17.18	-7.71	0.41	0.54	0.03	5.00
5	-7.96 dB /	45° steps	Std. Dev.	5.04	0.65	0.03	0.40	0.02	3.91
	Overall 24	A	vr.	-18.56	-5.44	0.55	0.41	0.03	4.02
	states	Std.	Dev.	5.57	1.87	0.11	0.44	0.03	3.61
	State	Set 1	Avr.	-16.87	-3.90	0.64	0.35	0.03	4.69
Ν	-3.74 dB /	45° steps	Std. Dev.	5.99	0.52	0.04	0.39	0.03	4.51
Ξ	State Set 2		Avr.	-16.44	-6.06	0.50	0.31	0.02	3.17
Ū.	-6.02 dB /	45° steps	Std. Dev.	2.96	0.35	0.02	0.11	0.01	2.29
ы.	State	Set 3	Avr.	-12.26	-9.09	0.35	0.57	0.02	2.79
2	-9.12 dB /	45° steps	Std. Dev.	4.90	0.68	0.03	0.31	0.01	2.61
	Overall 24	A	vr.	-15.19	-6.35	0.50	0.41	0.02	3.55
	states	Std.	Dev.	5.04	2.23	0.12	0.31	0.02	3.25

Table 6.5: The summary of measured performance of the RF MEMS VM.

Table 6.5 shows the summary of the measured data for RF MEMS VM for 22.5 GHz, 25 GHz, and 27.5 GHz frequencies. The RF MEMS VM is reconfigured to 8 insertion phase states for each of the tabulated 3 amplitude states, which make a total of 24 states at each frequency, to demonstrate the wide operation frequency range of the RF MEMS VM. The insertion phase states are apart by 45°. The targeted amplitude states have  $|S_{21}|$  values of 0.7, 0.55, and 0.4 for 22.5 GHz and 25 GHz, and 0.65, 0.5, and 0.35 for 27.5 GHz. The dB equivalents of the states are given in the table. The average phase errors among the measured 24 states are 3.96°, 4.02°, and 3.55° for 22.5 GHz, 25 GHz, and 27.5 GHz frequencies, respectively. The average amplitude errors are 0.52 dB, 0.41 dB, and 0.41 dB for 22.5 GHz, 25 GHz, and 27.5 GHz, and 27.5 GHz frequencies, respectively.

24 states are -16.2 dB, -18.6 dB, and -15.2 dB for 22.5 GHz, 25 GHz, and 27.5 GHz frequencies, respectively. The standard deviations of all the mentioned measurements are also given in Table 6.5.

Figure 6.17 shows the measurement results of the VM for 22.5 GHz, 25 GHz, and 27.5 GHz frequencies. The graphs show samples of insertion phase/amplitude states that can be covered at those frequencies. The graphs also include the selected 24 states, the data of which were given in Table 6.5. The measurement results in the graphs show that the design and the fabrication are both successful, and RF MEMS VM can provide 24 insertion phase/amplitude states between 22.5 GHz and 27.5 GHz.

It should be emphasized once more here that the 45 ° phase resolution and the -2 dB amplitude resolution are targeted for proof of concept purposes, and the performance is definitely not limited to these values. The figures clearly demonstrate that it is easily possible to have insertion phase steps with lower than 10 ° phase and 1 dB amplitude resolutions with average errors less than 5 ° and 0.5 dB, respectively. The author believes that the RF MEMS VM can be reconfigured to any frequency between 22.5 and 27.5 GHz frequencies.

The measurements that are reconfigured at 15 GHz are not given here since the performance is seriously affected from the change in the capacitance values, especially the increase in  $C_{bu}$ . The reason behind this behavior is the decrease in the electrical lengths of the stubs and the interconnection line. The range of the interconnection line is severely reduced, which is critical for the proper operation of the VM. The discussion about the changes in the capacitances in the circuit model was discussed earlier in Chapter 5.3, and it will be not repeated here.







Figure 6.17: The measured  $S_{21}$  performance of the RF MEMS VM (SOLT calibration used) for some sets of sample phase/amplitude pairs (a) 22.5 GHz (b) 25 GHz (c) 27.5 GHz.

The insertion loss performance of the RF MEMS VM is measured higher than the initial simulations results, which was expected from the experience obtained from the RF MEMS PS measurements. The initial design amplitude levels were -2 dB, -4.5 dB, and -8 dB where the targeted values are -3.1 dB, -5.2 dB, and -8 dB. The reasons of the increasing insertion losses were explained in Chapter 5 in detail. It is interesting to note here that even at 22.5 GHz, which is 1.5 times the initial design center frequency, there exist some states with about -2 dB insertion loss. This is because -2 dB is not maximum amplitude that the initial design could provide. The -2 dB level was selected to obtain the targeted amplitude value at each phase states so that the amplitude error remains small.

Another observation is that the instantaneous bandwidth is larger than that of the initial design, especially for the lower amplitude states. This is due to the fact that to obtain low amplitude levels with the presence of low insertion losses, the VM

circuit has to operate in the vicinity of the resonance points where the amplitude response changes very rapidly. However, as the insertion losses increase both due to the differences between the simulated and the measured response and due to the increasing frequency, the quality factor of the resonating VM circuit decreases. This results with peak points with less sharpness around the resonance point. Hence, the amplitude response of the circuit changes more slowly, which means a larger instantaneous bandwidth.

#### 6.4. Conclusion

In this chapter, the design, the fabrication, and the measurement results of the RF MEMS vector modulator that is based on the triple stub topology are presented. The measurement results show good agreement with the simulations in 15 GHz to 30 GHz frequency range. The VM has good phase error, amplitude error, and return loss performance at the range starting from 22.5 GHz up to 27.5 GHz. The wideband operation and the performance within this frequency range show that the RF MEMS VM is completely reconfigurable. However, the VM performance is severely affected from the changes in the MEMS bridge heights, and the performance is limited to only some specific states at the center design frequency of 15 GHz. In any case, the proof of concept of RF MEMS VM is demonstrated; in addition to that, the fabricated VM is ready to use in the 22.5 GHz-27.5 GHz frequency range.

## **CHAPTER 7**

## **CONCLUSIONS AND FUTURE WORK**

This thesis presented the theory, the design, and the implementation of novel reconfigurable RF MEMS circuits that are based on triple stub topology. These circuits include an impedance tuner, a phase shifter, and a vector modulator. The RF MEMS circuits are fabricated using in-house METU RF MEMS Fabrication Process, and the measurements of all of the circuits are accomplished successfully.

The fabricated RF MEMS impedance tuner is measured for 729 out of 1331 impedance points, and the measurements are in agreement with the simulations. It is demonstrated that the tuner covers the most of the Smith Chart area in 6-20 GHz band, especially between 10 GHz and 20 GHz, when the 10 dB impedance mismatch level is considered. The insertion loss of the tuner is measured to be -3.3 dB at both 10 GHz and 20 GHz. The tuner is also capable of covering the Smith Chart with 2<sup>30</sup> impedance points when more than one switch is actuated at once on each stub.

The performance of the fabricated RF MEMS phase shifter is investigated between 15 GHz and 40 GHz. The measurement results show that the phase shifter can cover 160-360° insertion phase range with 10° resolution with an average phase error of 1.7° at 15 GHz. The average insertion loss and the average return loss is measured to be -3.1 dB and -19.3 dB, respectively. It is also showed that the designed phase shifter can work up to 40 GHz with a insertion phase range about 300° starting from 22.5 GHz. The price paid is the gradually increasing insertion loss.

The performance of the fabricated RF MEMS vector modulator is examined in 10-30 GHz band. The vector modulator performance is demonstrated between 22.5 GHz and 27.5 GHz. The vector modulator is reconfigured to 8 phase states for each of the 3 amplitude states, which make a total of 24 states at each frequency. The phase state separation in 45°, and the targeted amplitude states have  $|S_{21}|$  values of 0.7, 0.55, and 0.4 for 22.5 GHz and 25 GHz, and 0.65, 0.5, and 0.35 for 27.5 GHz. The average phase errors among the measured 24 states are 3.96°, 4.02°, and 3.55° for 22.5 GHz, 25 GHz, and 27.5 GHz frequencies, respectively. The average amplitude errors are 0.52 dB, 0.41 dB, and 0.41 dB again for 22.5 GHz, 25 GHz, 25 GHz, and 27.5 GHz, and 27.5 GHz, measured 27.5 GHz, 25 GHz, 25 GHz, 25 GHz, 25 GHz, 25 GHz, 25 GHz, 25 GHz, 25 GHz, 30.41 dB, and 0.41 dB again for 22.5 GHz, 25 GHz,

Considering the measurement results that are summarized above, the following conclusions can be deduced:

- METU RF MEMS Fabrication Process is used successfully to implement complicated RF MEMS structures, which include a high number of RF MEMS components in a small wafer area. Significant increase is observed in the performance of the fabricated structures by the change of the substrate, the improvement of the first metal layer, and the improvement of the structural layer.
- The RF MEMS impedance tuner that has wide impedance coverage is implemented successfully in a very wide band with advantages brought by RF MEMS technology. The multi-pin MEMS automated measurement setup, which enables the measurement and characterization of RF MEMS structures with a high number of control pins, is developed for the measurements. The number of measured impedance points is the highest reported in the literature, which allows a complete characterization of the design.

- The theory of the triple stub topology is investigated in detail. It has been shown both analytically and experimentally that this topology is capable of providing insertion phase and amplitude control of the input signal simultaneously using only low-loss transmission lines. The mechanism of obtaining high insertion losses without any reflection is explained. It is also illustrated that maximum power transfer is possible to the transmission line-based resonators by means of direct connection instead of capacitive coupling.
- A wide band, reconfigurable RF MEMS phase shifter is demonstrated using the results extracted from the theory of triple stub topology. It is shown that the insertion phase can be controlled in a quasi-continuous manner by the presented RF MEMS phase shifter.
- A wide band, reconfigurable RF MEMS vector modulator is presented, which is again based on the results of the theory of triple stub topology. It is shown that the insertion phase can be controlled simultaneously with the amplitude in a quasi-continuous manner by using only low-loss distributed structures with the presented RF MEMS vector modulator.

Although the targets of this thesis are achieved successfully, there is always room for improvements. The following items can be investigated as future work:

 The fabrication of the entire RF MEMS designs can be fabricated again with a thinner first metal layer only in the areas underneath the MEMS bridges. This will prevent the formation of weak boundaries on the bridge, which will cease the buckling of the structures. A study has just been completed by METU RF MEMS Group, in which it is shown that the buckling is prevented with this method. Having MEMS bridge without any buckling will result with better agreement with the predicted performance, and the insertion losses are expected to decrease. Using a high-quality gold electroplating solution, which can provide a gold layer with higher conductivity, will also decrease the losses.

- The multi-pin MEMS automated measurement setup (MAMS) should be redesigned for application specific integrated circuit (ASIC) fabrication. Then, the RF MEMS structures and the setup should be placed in a hybrid package.
- The presented RF MEMS triple stub impedance tuner should be used for the implementation of real-time, autonomous impedance matching of an unknown load. The current MAMS needs a few hundred milliseconds to switch the tuner to a desired state. If the MAMS is redesigned as an ASIC as explained above, the interstate switching time will be reduced. In this case, computer or microcontroller based algorithm can be employed in order to obtain impedance matching of an unknown load. The presented RF MEMS impedance tuner allows this as the RF MEMS switches are placed uniformly on all of the stubs.
- The SiCr bias lines in the RF MEMS phase shifter and RF MEMS vector modulator should be redesigned, and the CPW ground penetrations should be avoided as much as possible. The modified designs with such bias lines are expected to have much lower insertion loss.
- The RF MEMS phase shifter and vector modulator should be investigated whether a predefined operation, for example a 4-bit phase shifter or a two amplitude state vector modulator, can be provided with less number of pins. In the current designs, the provided phase and amplitude ranges are covered in a continuous manner, so the present number of pins is necessary.

- The other alternative topologies for vector modulator, which are also based on the triple stub topology, should be implemented. The bandwidth of the simultaneous insertion phase/amplitude control is expected to increase significantly with these structures.
- The theory of triple stub topology should be investigated in more detail for the lossy case. Alternative analytical and graphical methods should be investigated for the exact solution of this problem.
- It has been shown that the set of switch states that are obtained from the optimization for a given insertion phase/amplitude can give better performance for RF MEMS phase shifter and vector modulator. An optimization method should be developed in order to find the best solution fast from the whole solution space, which currently includes 2<sup>32</sup> states.
- The RF MEMS phase shifter or the RF MEMS vector modulator can be employed in adaptive millimeterwave phase arrays. However, these structures should be redesigned with a reduced number of controls as mentioned above.
- The lifetime of the RF MEMS switches should be investigated, and a packaging solution should be proposed for the RF MEMS structures presented in this thesis. These problems are not expected to prevent the application of these circuits; however, they should be addressed carefully, and the limits should be determined before the application. Dedicated works that address both of these issues are currently investigated by METU RF MEMS Group.

Considering the summarized achievements of this thesis, it is shown that RF MEMS in an enabling technology, and it is possible to design and implement novel structures using this technology. The successful implementation of the above mentioned circuits shows that METU RF MEMS Fabrication Process, which is the only example of its kind in Türkiye, is mature. Using this fabrication process, it is possible to implement RF MEMS structures reliably and repeatedly. This result is very important in the sense that this is a national technology, and it should be used in the implementation of high performance defense applications.

## REFERENCES

- [1] http://www.its.caltech.edu/~feynman/plenty.html.
- [2] http://www.semiconductor.net/article/CA6527564.html.
- [3] L. E. Larson, R. H. Hackett, M. A. Melendes, and R. F. Lohr, "Micromachined microwave actuator (MIMAC) technology - a new tuning approach for microwave integrated circuits," *IEEE Microwave and Millimeter-Wave Monolithic Circuits Symp. Dig.*, Jun. 1991, pp. 27-30.
- [4] J. J. Yao and M. F. Chang, "A surface micromachined miniature switch for telecommunications applications with signal frequencies from DC up to 4 GHz," The 8<sup>th</sup> International Conference on Solid-State Sensors and Actuators Digest, Stockholm, Sweden, Jun. 1995, pp. 384-387.
- [5] C. Goldsmith, J. Randall, S. Eshelman, T. H. Lin, D. Denniston, S. Chen, and B. Norvell, "Characteristics of micromachined switches at microwave frequencies," *IEEE Int. Microwave Symp. Dig.*, San Francisco, CA, USA, Jun. 1996, pp. 1141 -1144.
- [6] http://www.wtc-consult.com/english/issue-4-v2-.html.
- [7] G. M. Rebeiz, *RF MEMS: Theory, Design, and Technology*. John Wiley & Sons, 2003.
- [8] Z. J. Yao, S. Chen, S. Eshelman, D. Denniston, and C. L. Goldsmith, "Micromachined low-loss microwave switches," *IEEE J. Microelectromech. Systems*, vol. 8, pp. 129–134, Jun. 1999.
- [9] J. B. Muldavin and G. M. Rebeiz, "High isolation MEMS shunt switches; Part 1: Modeling," *IEEE Trans. Microwave Theory Tech.*, vol. 48, no. 6, pp. 1045–1052, Jun. 2000.
- [10] J. B. Muldavin and G. M. Rebeiz, "High isolation MEMS shunt switches; Part 2: Design," *IEEE Trans. Microwave Theory Tech.*, vol. 48, no. 6, pp. 1053–1056, Jun. 2000.
- [11] J. Y. Park, G. H. Kim, K. W. Chung, and J. U. Bong, "Fully integrated micromachined capacitive switches for RF applications," *IEEE Int. Microwave Symp. Dig.*, Boston, MA, Jun. 2000, pp. 283-286.

- [12] D. Hyman, A. Schmitz, B. Warneke, T. Y. Hsu, J. Lam, J. Brown, J. Schaffner, A. Walston, R. Y. Loo, G. L. Tangonan, M. Mehregany, and J. Lee, "Surface micromachined RF MEMS switches on GaAs substrates," *Int. J. RF Microwave CAE*, vol. 9, no. 4, pp. 348–361, 1999.
- [13] R. E. Mihailovich, M. Kim, J. B. Hacker, E. A. Sovero, J. Studer, J. A. Higgins, and J. F. DeNatale, "MEM relay for reconfigurable RF circuits," *IEEE Microwave Wireless Comp. Lett.*, vol. 11, no. 2, pp. 53–55, Feb. 2001.
- [14] http://www.radantmems.com/radantmems/index.html.
- [15] S. Duffy, C. Bozler, S. Rabe, J. Knecht, L. Travis, P. Wyatt, C. Keast, and M. Gouker, "MEMS microswitches for reconfigurable microwave circuitry," *IEEE Microwave and Wireless Comp. Lett.*, vol. 11, no. 3, pp. 106–108, Mar. 2001.
- [16] D. J. Young and B. E. Boser, "A micromachined variable capacitor for monolithic low-noise VCOs," *Solid-State and Actuator Workshop*, Hilton Head, Jun. 1996, pp. 86–89.
- [17] A. Dec and K. Suyama, "Micromachined electro-mechanically tunable capacitors and their applications to RF IC's," *IEEE Trans. Microwave Theory Tech.*, vol. 46, no. 12, pp. 2587–2595, Dec. 1998.
- [18] A. Malczewski, S. Eshelman, B. Pillans, J. Ehmke, and C. L. Goldsmith, "X-Band RF MEMS phase shifters for phased array applications," *IEEE Microwave Guided Wave Lett.*, vol. 9, no. 12, pp. 517–519, Dec. 1999.
- [19] M. Kim, J. B. Hacker, R. E. Mihailovich, and J. F. DeNatale, "A DC-to-40 GHz four-bit RF MEMS true-time delay network," *IEEE Microwave Wireless Comp. Lett.*, vol. 11, no. 2, pp. 56–58, Feb. 2001.
- [20] G. L. Tan, R. E. Mihailovich, J. B. Hacker, J. F. DeNatale, and G. M. Rebeiz, "Low-Loss 2- and 4-Bit TTD MEMS phase shifters based on SP4T switches," *IEEE Trans. Microwave Theory Tech.*, vol. 51, no. 1, pp. 297–304, Jan. 2003.
- [21] B. Pillans, S. Eshelman, A. Malczewski, J. Ehmke, and C. Goldsmith, "Ka-band RF MEMS phase shifters," *IEEE Microwave Guided Wave Lett.*, vol. 9, no. 12, pp. 520–522, Dec. 1999.
- [22] N. S. Barker and G. M. Rebeiz, "Distributed MEMS true-time delay phase shifters and wideband switches," *IEEE Trans. Microwave Theory Tech.*, vol. 46, no. 11, pp. 1881–1890, November 1998.

- [23] N. S. Barker and G. M. Rebeiz, "Optimization of distributed MEMS transmission line phase shifters — U-band and W-band designs," *IEEE Trans. Microwave Theory Tech.*, vol. 48, no. 11, pp. 1957–1966, Nov. 2000.
- [24] Y. Liu, A. Borgioli, A. S. Nagra, and R. A. York, "K-band 3-bit low-loss distributed MEMS phase shifter," *IEEE Microwave Guided Wave Lett.*, vol. 10, no. 10, pp. 415–417, Oct. 2000.
- [25] J. S. Hayden, A. Malczewski, J. Kleber, C. L. Goldsmith, and G. M. Rebeiz, "2 and 4-bit DC-18 GHz microstrip MEMS distributed phase shifters," *IEEE Int. Microwave Symp. Dig.*, Phoenix, AZ, USA, May 2001, pp. 219-222.
- [26] H. -T. Kim, J. -H. Park, Y. -K. Kim, and Y. Kwon, "V-band low-loss and lowvoltage distributed MEMS digital phase shifter using metal-air-metal capacitors," *IEEE Int. Microwave Symp. Dig.*, Seattle, WA, Jun. 2002, pp. 341-344.
- [27] J. S. Hayden and G. M. Rebeiz, "Very low loss distributed X-band and Ka-band MEMS phase shifters using metal–air–metal capacitors," IEEE Trans. Microwave Theory Tech., vol. 51, no. 1, pp. 309–314, Jan. 2003.
- [28] J. -J. Hung, L. Dussopt, and G. M. Rebeiz, "Distributed 2- and 3-Bit W-band MEMS phase shifters on glass substrates," *IEEE Trans. Microwave Theory Tech.*, vol. 52, pp. 600-606, Feb. 2004.
- [29] N. Kingsley and J. Papapolymerou, "Organic "wafer-scale" packaged miniature 4-bit RF MEMS phase shifter," *IEEE Trans. Microwave Theory Tech.*, vol. 54, no. 3, pp. 1229-1236, Mar. 2006.
- [30] K. Van Caekenberghe and T. Vähä-Heikkilä, "An analog RF MEMS slotline truetime-delay phase shifter," *IEEE Trans. Microwave Theory and Tech.*, vol. 56, no. 9, 2151-2159, Sep. 2008.
- [31] H. -T. Kim, S. Jung, K. Kang, J. -H. Park, Y. -K. Kim, and Y. Kwon, "Low-loss analog and digital micromachined impedance tuners at the Ka-band," *IEEE Trans. Microwave Theory and Tech.*, vol. 49, no. 12, 2394-2400, Dec. 2001.
- [32] M. Unlu, K. Topalli, S. Demir, O. A. Civi, S. Koc, and T. Akin, "RF MEMS adjustable impedance matching network and adjustable power divider," *IEEE AP-S Int. Symp. Dig.*, San Antonio, TX, Jun. 2002, vol. 2, pp. 26-29.
- [33] J. Papapolymerou, K. L. Lange, C. L. Goldsmith, A. Malczewski, and J. Kleber, "Reconfigurable double-stub tuners using MEMS switches for intelligent RF front ends," *IEEE Trans. Microwave Theory Tech.*, vol. 51, no. 1, pp. 271-278, Jan. 2003.

- [34] T. Vähä-Heikkilä and G. M. Rebeiz, "A 4–18-GHz reconfigurable RF MEMS matching network for power amplifier applications," *Int. J. RF Microwave CAE*, vol. 14, no. 4, pp. 356-372, Jun. 2004.
- [35] M. Unlu, K. Topalli, H. I. Atasoy, E. U. Temocin, I. Istanbulluoglu, O. Bayraktar, S. Demir, O. A. Civi, S. Koc, and T. Akin, "A reconfigurable RF MEMS triple stub impedance matching network," *36th European Microwave Conf. Dig.*, Manchester, UK, Sep. 2006, pp. 1370-1373.
- [36] Q. Shen and N. S. Barker, "Distributed MEMS tunable matching network using minimal-contact RF-MEMS varactors," *IEEE Trans. Microwave Theory Tech.*, vol. 54, no. 6, pp. 2646-2658, Jun. 2006.
- [37] T. Vähä-Heikkilä, K. Van Caekenberghe, J. Varis, J. Tuovinen, and G. M. Rebeiz, "RF MEMS impedance tuners for 6-24 GHz applications," Int. J. RF Microwave CAE, vol. 17, no. 3, pp. 265-178, Mar. 2007.
- [38] J. Brank, J. Yao, M. Eberly, A. Malczewski, K. Varian, and C. L. Goldsmith, "RF MEMS-based tunable filters," Int. J. RF Microwave CAE, vol. 11, pp. 276-284, Sept. 2001.
- [39] J. -H. Park, H. -T. Kim, Y. Kwon, and Y. -K. Kim, "Tunable millimeter-wave filters using coplanar waveguide and micromachined variable capacitors," *Microengineering and Micromechanics*, pp. 706–712, Nov. 2001.
- [40] Y. Liu, A. Borgioli, A. S. Nagra, and R. A. York, "Distributed MEMS transmission lines for tunable filter applications," *Int. J. RF Microwave CAE*, vol. 11, pp. 254-260, Sep. 2001.
- [41] D. Peroulis, S. Pacheco, K. Sarabandi, and L. P. Katehi, "Tunable lumped components with applications to reconfigurable MEMS filters," *IEEE Int. Microwave Symp. Dig.*, Phoenix, AZ, Jun. 2001, pp. 341-344.
- [42] E. Fourn, A. Pothier, C. Champeaux, P. Tristant, A. Catherinot, P. Blondy, G. Tanne, E. Rius, C. Person, and F. Huret, "MEMS switchable interdigital coplanar filter," *IEEE Trans. Microwave Theory Tech.*, vol. 51, no. 1, pp. 320-324, Jan. 2003.
- [43] R. N. Simons, D. Chun, and L. P. B. Katehi, "Microelectromechanical systems (MEMS) actuators for antenna reconfigurability," *IEEE Int. Microwave Symp. Dig.*, May 2001, pp. 215-218.
- [44] J. C. Chiao, S.-Y. Cheng, J. L. Chang, I. M. Chio, Y. Kang, and J. Hayasaka, "MEMS reconfigurable antennas," *Int. J. RF Microwave CAE*, vol. 11, pp. 301-309, 2001.

- [45] C. W. Baek, S. Song, Y.-K. Kim C. Cheon and, and Y. Kwon, "2D mechanical beam steering antenna fabricated using MEMS technology," *IEEE Int. Microwave Symp. Dig.*, Phoenix, AZ, USA, Jun. 2001, pp. 211-214.
- [46] F. Shan, M. Lu, and B. Gao, "Rectangular increasing distance spiral slots (RIDSS) PBG applied to MEMS frequency-reconfigurable antenna design," *Microwave and Optical Tech. Lett.*, vol. 47, no. 4, pp. 394-395, Sep. 2005.
- [47] D. E. Anagnostou, G. Zheng, M. T. Chryssomallis, J. C. Lyke, G. E. Ponchak, J. Papapolymerou, and C. G. Christodoulou, "Design, fabrication, and measurements of an RF-MEMS-based self-similar reconfigurable antenna," *IEEE Trans. Antennas and Propagation*, vol. 54, no. 2, pp. 422-432, Feb. 2006.
- [48] E. Erdil, K. Topalli, M. Unlu, I. Istanbulluoglu, E. U. Temocin, H. I. Atasoy, O. Bayraktar, O. A. Civi, S. Demir, S. Koc, and T. Akin, "Reconfigurable slot antenna with fixed-fixed beam type RF MEMS capacitors," *European Conf. Antennas and Propagation*, Nice, France, 6-10 Nov. 2006.
- [49] E. Erdil, K. Topalli, M. Unlu, O. Aydin Civi, and T. Akin, "Frequency tunable microstrip patch antenna using RF MEMS technology," *IEEE Trans. Antennas* and Propagation, vol. 55, pp. 1193-1196, Apr. 2007.
- [50] N. Kingsley, G. E. Ponchak, and J. Papapolymerou "Reconfigurable RF MEMS phased array antenna integrated within a liquid crystal polymer (LCP) systemon-package," *IEEE Trans. Antennas and Propagation*, vol. 56, no. 1, pp. 108-118, Jan. 2008.
- [51] C.W. Jung, M. -J. Lee and F. De Flaviis, "Reconfigurable dual-band antenna with high frequency ratio (1.6:1) using MEMS switches," *Electronics Letters*, vol. 44, no. 2, pp. 76-77, Jan. 2008.
- [52] K. Topalli, O. Aydin Civi, S. Demir, S. Koc, and T. Akin, "A monolithic phased array using 3-bit distributed RF MEMS phase shifters," *IEEE Trans. Antennas* and Propagation, vol. 56, no. 2, pp. 270-277, Feb. 2008.
- [53] C. T. C. Nguyen, L. P. B. Katehi, and G. M. Rebeiz, "Micromachined devices for wireless communications," *Proc. IEEE*, vol. 86, pp. 1756-1768, 1998.
- [54] G. M. Rebeiz, L. P. B. Katehi, T. M. C. Weller, C. Y. Chi, and R. Robertson, "Micromachined membrane filters for microwave and millimeter-wave applications," *Int. J. RF Microwave CAE*, vol. 7, pp. 149-166, 1997.
- [55] H. A. C. Tilmans, W. De Raedt, and E. Beyne, "MEMS for wireless communications: 'from RF-MEMS components to RF-MEMS-SiP'," J. Micromech. Microeng., vol. 13, pp. 139-163, 2003.

- [56] http://www.avagotech.com.
- [57] C. Goldsmith, J. Maciel, and J. Mckillop, "Demonstrating reliability," *IEEE Microwave Magazine*, vol. 8, no. 6, pp. 56-60, Dec. 2007.
- [58] C. L. Goldsmith, D. I. Forehand, Z. Peng, J. C. M. Hwang, and J. L. Ebel, "High-cycle life testing of RF MEMS switches," *IEEE Int. Microwave Symp. Dig.*, Honolulu, HI, USA, June 2007, pp. 1805–1808.
- [59] D. I. Forehand and C. L. Goldsmith, "Zero-level packaging for RF MEMS switches," *Govt. Microcircuit Applications and Critical Tech Conf.*, San Diego, CA, USA, Mar. 2006, paper 4.03.
- [60] J. Bouchaud, B. Knoblich, and H. Wicht, "Will RF MEMS live up their promise?," 36th European Microwave Conf. Dig., Manchester, UK, Sep. 2006, pp. 1076-1079.
- [61] C. Bozler, R. Drangmeister, S. Duffy, M. Gouker, J. Knecht, L. Kushner, R. Parr, S. Rabe, and L. Travis, "MEMS microswitch arrays for reconfigurable distributed microwave components," *IEEE Int. Microwave Symp. Dig.*, Boston, MA, USA, Jun. 2000, pp. 153–156.
- [62] W. Bischof, "Variable impedance tuner for MMIC's," *IEEE Microwave Guided Wave Lett.*, vol. 4, no. 6, pp. 172-174, 1994.
- [63] C. E. McIntosh, R. D. Pollard, and R. E. Miles, "Novel MMIC source-impedance tuners for on-wafer microwave noise-parameter measurements," *IEEE Trans. Microwave Theory Tech.*, vol. 47, no. 2, pp. 125-131, Feb. 1999.
- [64] R. N. de Lima, B. Huyart, E. Bergeault, and L. Jallet, "MMIC impedance matching system," *Electronics Lett.*, vol. 36, no. 16., Aug. 2000.
- [65] D. Pienkowski and W. Wiatr, "Broadband electronic impedance tuner," Int. Conf. Microwaves, Radar Wireless Comm., Gdansk, Poland, May 2002, pp. 310-313.
- [66] Z. Zhou and K. L. Melde, "Frequency agility of broadband antennas integrated with a reconfigurable RF impedance tuner," *IEEE Antennas and Wireless Propagation Lett.*, vol. 6, pp. 56-59, 2007.
- [67] J. -C. Chiao, Y. Fu, D. Choudhury, and L. -Y. Lin, "MEMS millimeterwave components," *IEEE Int. Microwave Symp. Dig.*, Anaheim, CA, USA, Jun. 1999, pp. 463-466.
- [68] H. -T. Kim, S. Jung, K. Kang, J. -H. Park, Y. -K. Kim, and Y. Kwon, "Low-loss analog and digital micromachined impedance tuners at the Ka-band," *IEEE Trans. Microwave Theory Tech.*, vol. 49, no. 12, pp. 2394-2400, Dec. 2001.

- [69] M. Unlu, K. Topalli, S. Demir, O. A. Civi, S. Koc, and T. Akin, "RF MEMS adjustable impedance matching network and adjustable power divider," *IEEE* AP-S Int. Symp. Dig., San Antonio, TX, Jun. 2002, vol. 2, pp. 26-29.
- [70] J. Papapolymerou, K. L. Lange, C. L. Goldsmith, A. Malczewski, and J. Kleber, "Reconfigurable double-stub tuners using MEMS switches for intelligent RF front-ends," *IEEE Trans. Microwave Theory Tech.*, vol. 51, no. 1, pp. 271-278, Jan. 2003.
- [71] G. Zheng, P. L. Kirby, S. Pajic, A. Pothier, P. Blondy, J. Papapolymerou, and Z. Popovic, "A monolithic reconfigurable tuner with ohmic contact MEMS switches for efficiency optimization of X-band power amplifiers," *Silicon Monolithic Integrated Circuits in RF Systems Top. Meet. Dig.*, Sep. 2004, pp. 159-162.
- [72] T. Vähä-Heikkilä and G. M. Rebeiz, "A 4–18-GHz reconfigurable RF MEMS matching network for power amplifier applications," *Int. J. RF Microwave CAE*, vol. 14, no. 4, pp. 356-372, Jun. 2004.
- [73] D. Qiao, R. Molfino, S. M. Lardizabal, B. Pillans, P. M. Asbeck, and G. Jerinic, "An intelligently controlled RF power amplifier with a reconfigurable MEMS-varactor tuner," *IEEE Trans. Microwave Theory Tech.*, vol. 53, no. 3, pp. 1089-1095, Mar. 2005.
- [74] T. Vähä-Heikkilä, J. Varis, J. Tuovinen, and G. M. Rebeiz, "A 20-50 GHz RF MEMS single-stub impedance tuner," *IEEE Microwave Wireless Comp. Lett.*, vol. 15, no. 4, pp. 205-207, Apr. 2005.
- [75] Y. Lu, L. P. B. Katehi, and D. Peroulis, "High-power MEMS varactors and impedance tuners for millimeter-wave applications," *IEEE Trans. Microwave Theory Tech.*, vol. 53, no. 11, pp. 3672-3678, Nov. 2005.
- [76] Q. Shen and N. S. Barker, "Distributed MEMS tunable matching network using minimal-contact RF-MEMS varactors," *IEEE Trans. Microwave Theory Tech.*, vol. 54, no. 6, pp. 2646-2658, Jun. 2006.
- [77] T. Vähä-Heikkilä, K. Van Caekenberghe, J. Varis, J. Tuovinen, and G. M. Rebeiz, "RF MEMS impedance tuners for 6-24 GHz applications," Int. J. RF Microwave CAE, vol. 17, no. 3, pp. 265-278, May 2007.
- [78] M. Unlu, K. Topalli, H. I. Atasoy, E. U. Temocin, I. Istanbulluoglu, O. Bayraktar, S. Demir, O. A. Civi, S. Koc, and T. Akin, "A reconfigurable RF MEMS triple stub impedance matching network," *36th European Microwave Conf. Dig.*, Manchester, UK, Sep. 2006, pp. 1370-1373.

- [79] C. R. Boyd, "An accurate analog ferrite phase shifter," *Microwave Symp. Dig.*, 1971, pp. 104-105.
- [80] W. E. Hord Jr, C. R. Boyd, and D. Diaz, "A new Type of fast-switching dualmode ferrite phase shifter," *IEEE Trans. Microwave Theory Tech.*, vol. 35, no. 12, pp. 1219-1225, Dec. 1987.
- [81] J. D. Adam, L. E. Davis, G. F. Dionne, E. F. Schloemann, and S. N. Stitzer, "Ferrite devices and materials," *IEEE Trans. Microwave Theory Tech.*, vol. 50, no. 3, pp. 721-737, Mar. 2002.
- [82] M. J. Schindler and M. E. Miller, "A 3-bit K/Ka band MMIC phase shifter," IEEE Microwave and Millimeter-Wave Monolithic Circuits Symp. Dig., New York, NY, USA, 1988, pp. 95-98.
- [83] D. C. Boire and R. Marion, "A high-performance 6–18 GHz five bit MMIC phase shifter," IEEE GaAs IC Symp. Dig., Oct. 1995, pp. 267-270.
- [84] M. Aust, H. Wang, R. Carandang, K. Tan, C. H. Chen, T. Trinh, R. Esfandiari, and H. C. Yen, "GaAs monolithic components development for Q-band phased array application," *IEEE Int. Microwave Symp. Dig.*, Albuquerque, NM, USA, Jun. 1992, pp. 703-706.
- [85] A. W. Jacomb-Hood, D. Seielstad, and J. D. Merrill, "A three-bit monolithic phase shifter at V-band," *IEEE Microwave and Millimeter-Wave Monolithic Circuits Symp. Dig.*, Jun. 1987, pp. 81–84.
- [86] T. A. Murphy, J. W. Gipprich, M. E. Hines, and D. Kruger, "A 6-bit GMIC phase shifter for 6–18 GHz," *IEEE Int. Microwave Symp. Dig.*, Albuquerque, NM, USA, Jun. 1992, pp. 1171-1174.
- [87] D. Teeter, R. Wohlert, B. Cole, G. Jackson, E. Tong, P. Saledas, M. Adlerstein, M. Schindler, and S. Shanfield, "Ka-band GaAs HBT PIN diode switches and phase shifters," *IEEE Int. Microwave Symp. Dig.*, San Diego, CA, USA, 1994, pp. 451-454.
- [88] S. Weinreb, W. Berk, S. Duncan, and N. Byer, "Monolithic varactor 360° phase shifters for 75–110 GHz," Int. Semiconductor Device Research Conf. Dig., Charlottesville, VA, USA, Dec. 1993.
- [89] A. S. Nagra, J. Xu, E. Erker, and R. A. York, "Monolithic GaAs phase shifter circuit with low insertion loss and continuous 0-360° phase shift at 20 GHz," *IEEE Microwave Guided Wave Lett.*, vol. 9, no. 1, pp. 31-33, Jan. 1999.

- [90] A. S. Nagra and R. A. York, "Distributed analog phase shifters with low insertion loss," *IEEE Trans. Microwave Theory Tech.*, vol. 47, no. 9, pp. 1705-1711, Sep. 1999.
- [91] M. Tuckman, "IQ Vector Modulator-The Ideal Control Component?," Microwave Systems News, pp. 105-115, 1988.
- [92] V. Cannistraro, J. C. Liou, and S. McCarter, "Direct modulation lowers VSAT equipment costs," *Microwaves & RF*, vol. 29, p. 99-102, Aug. 1990.
- [93] S. Kumar, "Directly modulated VSAT transmitters," *Microwave Journal*, vol. 33, pp. 255-264, 1990.
- [94] A. Truitt, J. Cerney, J. S. Mason, T. I. Inc, and T. X. Dallas, "A 0.3 to 3 GHz monolithic vector modulation for adaptive array systems," *IEEE Int. Microwave Symp. Dig.*, Long Beach, CA, USA, Jun. 1989, pp. 421-422.
- [95] G. B. Norris, D. C. Boire, G. St. Onge, C. Wutke, C. Barratt, W. Coughlin, and J. Chickanosky, "A fully monolithic 4-18 GHz digital vector modulator," *IEEE Int. Microwave Symp. Dig.*, Dallas, TX, USA, May 1990, pp. 789-792.
- [96] S. Lucyszyn, T. Sewell, and I. D. Robertson, "Multi-level digital modulation performed directly at carrier frequency," 25th European Microwave Conf. Dig., Bologna, Italy, 1995, pp. 673-676.
- [97] S. Nam, U. Karacaoglu, C. Oztek-Yerli, and I. D. Robertson, "GaAs MMIC chipset for 17/18 GHz indoor radio LANs," 26th European Microwave Conf. Dig., Prague, Czech Rep., Oct. 1996, pp. 960-963.
- [98] Y. -S. Dai and D. -G. Fang, "A novel UWB high performance digital/analog compatible vector modulator module with MMIC-based devices," *IEEE Asia-Pacific Microwave Conf. Proc.*, Suzhou, China, Dec. 2005.
- [99] L. M. Devlin and B. J. Minnis, "A versatile vector modulator design for MMIC," IEEE Int. Microwave Symp. Dig., Dallas, TX, USA, May 1990, pp. 519-521.
- [100] D. C. W. Lo, H. Wang, B. R. Allen, G. S. Dow, K. W. Chang, M. Biedenbender, R. Lai, S. Chen, and D. Yang, "Novel monolithic multifunctional balanced switching low-noise amplifiers," *IEEE Trans. Microwave Theory Tech.*, vol. 42, no.12, pp. 2629-2634, Dec. 1994.
- [101] A. E. Ashtiani, S. Nam, A. d'Espona, S. Lucyszyn, and I. D. Robertson, "Direct multilevel carrier modulation using millimeter-wave balanced vector modulators," *IEEE Trans. Microwave Theory Tech.*, vol. 46, no. 12, pp. 2611-2619, Dec. 1998.

- [102] S. Nam, N. Shala, K. S. Ang, A. E. Ashtiani, T. Gokdemir, I. D. Robertson, and S. P. Marsh, "Monolithic millimeter-wave balanced bi-phase amplitude modulator inGaAs/InGaP HBT technology," *IEEE Int. Microwave Symp. Dig.*, 1999, Anaheim, CA, USA, 1999, pp. 243-246.
- [103] T. Lodhi, D. L. Edgar, H. McLelland, S. Ferguson, K. Elgaid, C. R. Stanley, and I. G. Thayne, "A 77 GHz coplanar waveguide MMIC BPSK vector modulator realized using InP technology," 22<sup>nd</sup> GaAs IC Symp. Dig., Seattle, WA, USA, 2000, pp. 183-186.
- [104] D. S. McPherson and S. Lucyszyn, "Vector modulator for W-band software radar techniques," *IEEE Trans. Microwave Theory Tech.*, vol. 49, no. 8, pp. 1451-1461, Aug. 2001.
- [105] C. Y. Ng, M. Chongcheawchamnan, and I. D. Robertson, "Miniature Ka-band I/Q vector modulator using 3D-MMIC technology," 33rd European Microwave Conf. Dig., Munich, Germany, Oct. 2003, pp. 841-844.
- [106] M. Chongcheawchamnan, S. Bunnjaweht, D. Kpogla, D. Lee, and I. D. Robertson, "Microwave IQ vector modulator using a simple technique for compensation of FET parasitics," *IEEE Trans. Microwave Theory Tech.*, vol. 50, no. 6, pp. 1642-1646, Jun. 2002.
- [107] H. Y. Chang, T. W. Huang, H. Wang, Y. C. Wang, P. C. Chao, and C. H. Chen, "Broad-band HBT BPSK and IQ modulator MMICs and millimeter-wave vector signal characterization," *IEEE Trans. Microwave Theory Tech.*, vol. 52, no. 3, pp. 908-919, Mar. 2004.
- [108] H. Y. Chang, P. S. Wu, T. W. Huang, H. Wang, Y. C. Tsai, and C. H. Chen, "An ultra compact and broadband 15-75 GHz BPSK modulator using 0.13-μm CMOS process," *IEEE Int. Microwave Symp. Dig.*, Long Beach, CA, USA, Jun. 2005, pp. 41-44.
- [109] H. Chang, P. Wu, T. Huang, H. Wang, C. Chang, and J. G. J. Chern, "Design and analysis of CMOS broad-band compact high-linearity modulators for gigabit microwave/millimeter-Wave applications," *IEEE Trans. Microwave Theory Tech.*, vol. 54, no. 1, pp. 20-30, Jan. 2006.
- [110] A. P. Freundorfer, K. Hamed, Y. Sun, Y. Antar, P. Frank, and D. Sawatzky, "A direct digital 2 Gb/s modulator/demodulator experiment in GaAs HBT at 30 GHz," *IEEE Asia-Pacific Microwave Conf. Proc.*, Yokohama, Japan, Dec. 2006, pp. 1611-1664.
- [111] H. Y. Chang, H. Wang, and W. Lin, "A miniature 35-110 GHz modified reflection-type BPSK modulator using 65-nm CMOS technology," *IEEE Int. Microwave Symp. Dig.*, Honolulu, HI, USA, Jun. 2007, pp. 2165-2168.

- [112] H. Y. Chang, "Design of broadband highly linear IQ modulator using a 0.5 μm E/D-PHEMT process for millimeter-wave applications," *IEEE Microwave Wireless Comp. Lett.*, vol. 18, no. 7, pp. 491-493, Jul. 2008.
- [113] Y. Hou, L. Li, R. Qian, and X. Sun, "An efficient technique for designing highperformance millimeter-wave vector modulators with low temperature drift," *IEEE Trans. Microwave Theory Tech.*, vol. 56, no. 12, pp. 3100-3107, Dec. 2008.
- [114] R. Pyndiah, P. Jean, R. Leblanc, and J. C. Meunier, "GaAs monolithic direct linear (1-2.8) GHz QPSK modulator," 19th European Microwave Conf. Dig., London, UK, Sep. 1989, pp. 597-602.
- [115] F. L. M. Van Den Bogaart and R. A. Pyndiah "10-14 GHz linear MMIC vector modulator with less than 0.1 dB and 0.8 degrees amplitude and phase error," *IEEE Int. Microwave Symp. Dig.*, Dallas, TX, USA, May 1990, pp. 465-468.
- [116] I. Telliez, A. M. Couturier, C. Rumelhard, C. Versnaeyen, P. Champion, and D. Fayol, "A compact, monolithic microwave demodulator-modulator for 64-QAM digital radio links," *IEEE Trans. Microwave Theory Tech.*, vol. 39, no. 12, pp. 1947-1954, Dec. 1991.
- [117] K. Yamamoto, K. Maemura, N. Andoh, and Y. Mitsui, "A 1. 9-GHz-band GaAs direct-quadrature modulator IC with a phase shifter," *IEEE J. Solid-State Circuits*, vol. 28, pp. 994-1000, no. 10, Oct. 1993.
- [118] A. Boveda and J. I. Alonso, "A 0.7-3 GHz GaAs QPSK/QAM direct modulator," *IEEE J. Solid-State Circuits*, vol. 28, no. 12, pp. 1340-1349, Dec. 1993.
- [119] P. Boutet, J. Dubouloy, M. Soulard, and J. Pinho, "Fully Integrated QPSK Linear Vector Modulator for Space Applications in Ku Band," 28th European Microwave Conf. Dig., Amsterdam, Netherlands, 1998, pp. 389-392.
- [120] Y. Hamada, K. Maruhashi, M. Ito, S. Kishimoto, T. Morimoto, and K. Ohata, "A 60-GHz-band compact IQ modulator MMIC for ultra-high-speed wireless communication," *IEEE Int. Microwave Symp. Dig.*, San Francisco, CA, USA, 2006, pp. 1701-1704.
- [121] J. -M. Wu, F. -Y. Han, T. -S. Horng, and J. Lin, "Direct-conversion quadrature modulator MMIC design with a new 90° phase shifter including package and PCB effects for W-CDMA applications," *IEEE Trans. Microwave Theory Tech.*, vol. 54, no. 6, pp. 2691-2698, Jun. 2006.
- [122] J. -H. Tsai and T. -W. Huang "35-65-GHz CMOS broadband modulator and demodulator with sub-harmonic pumping for MMW wireless gigabit applications," *IEEE Trans. Microwave Theory Tech.*, vol. 55, no. 10, pp. 2075-2085, Oct. 2007.

- [123] K. W. Hamed, A. P. Freundorfer, Y. M. M. Antar, P. Frank, and D. Sawatzky, "A high-bit rate Ka-band direct conversion QPSK demodulator," *IEEE Microwave Wireless Comp. Lett.*, vol. 18, no. 5, pp. 365-367, May 2008.
- [124] C. Wagner, M. Hartmann, A. Stelzer, and H. Jaeger, "A fully differential 77-GHz active IQ modulator in a Silicon-Germanium technology," *IEEE Microwave Wireless Comp. Lett.*, vol. 18, no. 5, pp. 362-364, May 2008.
- [125] A. Primerose, J. Graffeuil, L. Lapierre, J. Sombrin, J. C. Lalaurie, and J. Larroque, "High bit rate four phase MMIC remodulation demodulator and modulator," *Proc. European Gallium Arsenide and Related III-V Compounds Apps. Symp.*, Apr. 1992.
- [126] G. Burnsed, D. Kalcik, and J. Sherman, "C-band analog monolithic vector modulator," *IEEE Microwave and Millimeter-Wave Monolithic Circuits Symp. Dig.*, Orlando, FL, USA, May 1995, pp. 91-94.
- [127] J. Grajal, J. Gismero, M. Mahfoudi, and F. A. Petz, "A 1.4-2.7-GHz analog MMIC vector modulator for a crossbar beamforming network," *IEEE Trans. Microwave Theory Tech.*, vol. 45, no. 10, pp. 1705-1714, Oct. 1997.
- [128] F. Ellinger, U. Lott, and W. Bachtold, "An antenna diversity MMIC vector modulator for HIPERLAN with low power consumption and calibration capability," *IEEE Trans. Microwave Theory Tech.*, vol. 49, no. 5, pp. 964-969, May 2001.
- [129] M. Altuntaş, "MMIC vector modulator design," M. Sc. thesis, Electrical and Electronics Eng. Dept., METU, Ankara, 2004.
- [130] P. S. Wu, H. Y. Chang, M. D. Tsai, T. W. Huang, and H. Wang, "New miniature 15-20-GHz continuous-phase/amplitude control MMICs using 0.18-μm CMOS technology," *IEEE Trans. Microwave Theory Tech.*, vol. 54, no. 1, pp. 10-19, Jan. 2006.
- [131] P. S. Wu, H. Y. Chang, M. F. Lei, B. J. Huang, H. Wang, C. M. Yu, and J. G. J. Chern, "A 40-74 GHz amplitude/phase control MMIC using 90-nm CMOS technology," *European Microwave Integrated Circuit Conf.*, Munich, Germany, Oct. 2007, pp. 115-118.
- [132] M. Goldfarb, J. B. Cole, A. Platzker, R. Div, R. Co, and M. A. Lexington, "A novel MMIC biphase modulator with variable gain using enhancement-mode FETs suitable for 3 V wireless applications," *IEEE Microwave and Millimeter-Wave Monolithic Circuits Symp. Dig.*, San Diego, CA, USA, May 1994, pp. 99-102.

- [133] W. Philibert and R. Verbiest, "A subharmonically pumped I/Q vector modulator MMIC for Ka band satellite communication," *IEEE Radio Frequency Integrated Circuits Symp. Dig.*, Boston, MA, USA, Jun. 2000, pp. 183-186.
- [134] G. J. Carchon, D. M. M. -P. Schreurs, W. De Raedt, P. Van Loock, and B. K. J. C. Nauwelaers "A direct Ku-band linear subharmonically pumped BPSK and I/Q vector modulator in multi-layer thin-film MCM-D," *IEEE Trans. Microwave Theory Tech.*, vol. 49, no. 8, pp. 1374-1382, Aug. 2001.
- [135] J. Robert, "Low cost Ka band IQ vector modulator with flip chip diodes," *31<sup>st</sup> European Microwave Conf. Dig.*, London, UK, 2001, pp. 1-5.
- [136] L. Wang, X. H. Tang, F. Xiao, and R. H. Liu, "A low cost X-band HMIC vector modulator," *IEEE Asia-Pacific Microwave Conf. Proc.*, Suzhou, China, Dec. 2005.
- [137] C. E. Saavedra, "A BPSK modulator using a ring-hybrid and HFET switches," IEEE Asia-Pacific Microwave Conf. Proc., Yokohama, Japan, Dec. 2006, pp. 1665-1668.
- [138] N. A. Riza, "Analog vector modulation-based widely tunable frequency photonic beamformer for phased-array antennas," *IEEE Trans. Microwave Theory Tech.*, vol. 45, no. 8, pp. 1508-1512, Aug. 1997.
- [139] W. D. Jemison, A. J. Kreuzberger, and E. Funk, "Microwave photonic vector modulator for high-speed wireless digital communications," *IEEE Microwave Wireless Comp. Lett.*, vol. 12, no. 4, pp. 125-127, Apr. 2002.
- [140] N. S. Barker and G. M. Rebeiz, "Distributed MEMS transmission-line BPSK modulator," *IEEE Microwave and Guided Wave Lett.*, vol. 10, no. 5, pp. 198-200, May 2000.
- [141] B. R. Jackson and C. E. Saavedra, "2.4 GHz direct-digital binary phase shift keying modulator using MEMS switch," *Electronics Lett.*, vol. 40, no. 24, pp. 1539-1541, Nov. 2004.
- [142] K. Topallı, "A monolithic phased array using RF MEMS technology," Ph. D. dissertation, Electrical and Electronics Eng. Dept., METU, Ankara, 2007.
- [143] http://www.quartzunlimited.com/Pages/Properties/electric.html.
- [144] M. Ünlü, "An adjustable impedance matching network using RF MEMS technology," M. Sc. thesis, Electrical and Electronics Eng. Dept., METU, Ankara, 2003.
- [145] E. Erdil, "Tunable frequency microstrip antennas by RF-MEMS technology,"
   M. Sc. thesis, Electrical and Electronics Eng. Dept., METU, Ankara, 2005.

- [146] PI2737 polyimide datasheet, http://www.hdmicrosystems.com.
- [147] H. İ. Atasoy, "Design and fabrication of RF MEMS switches and instrumentation for performance evaluation," M. Sc. thesis, Electrical and Electronics Eng. Dept., METU, Ankara, 2007.
- [148] Hüseyin Sağkol, "A phase shifter using RF MEMS technology," M. Sc. thesis, Electrical and Electronics Eng. Dept., METU, Ankara, 2002.
- [149] İpek İstanbulluoğlu, "A novel modeling methodology and performance improvement technique for DMTL phase shifters," M. Sc. thesis, Electrical and Electronics Eng. Dept., METU, Ankara, 2006.
- [150] Engin Ufuk Temoçin, "Design and implementation of microwave lumped components and system integration using MEMS technology," M. Sc. thesis, Electrical and Electronics Eng. Dept., METU, Ankara, 2006.
- [151] D. M. Mattox, Handbook of Physical Vapor Deposition (PVD) Processing Film Formation, Adhesion, Surface Preparation and Contamination Control. Noyes Publications, 1998.
- [152] J. A. Thornton, "The microstructure of sputter-deposited coatings," J. Vac. Sci. Technol. A, vol. 4, no. 6, pp. 3059-3065, Nov. 1986.
- [153] E. S. Machlin, *Materials Science in Microelectronics-The Relationships Between Thin Film Processing and Structure*. Giro, 1995.
- [154] C. R. M. Grovenor, H. T. G. Hentzell, D. A. Smith, "Development of grain structure during growth of metallic films," *Acta Metallurgica*, vol. 32, no. 5, pp. 773-781, May 1984.
- [155] C. V. Thompson, "Grain growth in thin films," *Annual Review of Materials Science*, vol. 20, pp. 245-268, Aug. 1990.
- [156] M. Ohring, *Materials Science of Thin Films Deposition & Structure*. Academic Press, 2002.
- [157] http://www.bestec.de.
- [158] R. A. Powell and S. Rossnagel, Thin Films, PVD for Microelectronics: Sputter Deposition Applied to Semiconductor Manufacturing, Vol. 26. Academic Press, 1999.
- [159] http://www.pgo.com/pdf/pyrex.pdf.

- [160] J. A. Hoffman and D. W. Hoffman, "Internal stresses in titanium, nickel, molybdenum, and tantalum films deposited by cylindrical magnetron sputtering," J. Vac. Sci. Technol. A, vol. 14, no. 1, pp. 164-168, Jan. 1977.
- [161] D. W. Hoffman and J. A. Hoffman, "Effects of substrate orientation and rotation on internal stresses in sputtered metal films," J. Vac. Sci. Technol. A, vol. 15, no. 2, pp. 134-137, Mar. 1979.
- [162] D. W. Hoffman, "Stress and property control in sputtered metal films without substrate bias," *Thin Solid Films*, vol. 107, no. 4, pp. 353–358, Sep. 1983.
- [163] J. A. Thornton and D.W. Hoffman, "Stress related effects in thin films," *Thin Solid Films*, vol. 171, no. 1, pp. 5–31, Apr. 1989.
- [164] D. W. Hoffman, "Perspective on stresses in magnetron-sputtered thin films," J. Vac. Sci. Technol. A, vol. 12, no. 4, pp. 953-961, Jul. 1994.
- [165] M. T. A. Saif and N. C. MacDonald, "Planarity of large MEMS," J. Microelectromech. Syst., vol. 5, pp. 79–97, Jun. 1996.
- [166] L. B. Freund and S. Suresh, *Thin Film Materials: Stress, Defect Formation, and Surface Evolution.* Cambridge University Press, 2003.
- [167] W. Fang and J. A. Wickert, "Determining mean and gradient residual stresses in thin films using micromachined cantilevers," J. Micromech. Microeng., vol. 6, pp. 301–309, Sept. 1996.
- [168] L. Lin, A. P. Pisano, and R. T. Howe, "A micro strain gauge with mechanical amplifier," *J. Microelectromech. Syst.*, vol. 9, pp. 313–321, Dec. 1997.
- [169] Y. H. Min and Y. K. Kim, "In situ measurement of residual stress in micromachined thin films using a specimen with composite-layered cantilevers," J. Micromech. Microeng., vol. 10, pp. 314–321, Sep. 2000.
- [170] Coventorware, Coventor Inc., Cary, NC, USA, 2001.
- [171] http://www.bsac.eecs.berkeley.edu/cfm.
- [172] R. E. Collin, *Foundations for Microwave Engineering*. New York: McGraw-Hill, 1992.
- [173] D. M. Pozar, *Microwave Engineering, Second Edition*. John Wiley & Sons, 2001.
- [174] G. L. Ragan, *Microwave Transmission Circuits*. New York: McGraw-Hill, 1948, pp. 466–472.

- [175] Microwave Office 2002. Appl. Wave Res. Inc., El Segundo, CA, USA, 2002.
- [176] HFSS v9.2, Ansoft Corporation, Pittsburgh, PA, USA, 2002.
- [177] A. K. Goel, *High-Speed VLSI Interconnections: Modeling, Analysis, and Simulation.* New York: Wiley, 1994.
- [178] K. C. Gupta, *Microstrip Lines and Slotlines*. Artech House, 1996.
- [179] R. Simons, *Coplanar waveguide Circuits, Components, and Systems*. John Wiley & Sons, 2001.
- [180] M. J. W. Rodwell, S. T. Allen, R. Y. Yu, M. G. Case, U. Bhattacharya, M. Reddy, E. Carman, M. Kamegawa, Y. Konishi, J. Pusl, and R. Pullela, "Active and nonlinear wave propagation devices in ultrafast electronics and optoelectronics," *Proc. IEEE*, vol. 82, no. 7, pp. 1037–1059, Jul. 1994.
- [181] J. S. Hayden, "High-performance digital X-band and Ka-band distributed MEMS phase shifters," Ph. D. dissertation, Dept. of Electrical Eng., The Univ. of Michigan, Ann Arbor, MI, USA, 2002.
- [182] G. M. Rebeiz, "Phase-noise analysis of MEMS-based circuits and phase shifters," *IEEE Trans. Microwave Theory Tech.*, vol. 50, no. 5, pp. 1316-1323, May 2002.
- [183] N. S. Barker, "Distributed MEMS transmission lines," Ph. D. dissertation, Dept. of Electrical Eng., The Univ. of Michigan, Ann Arbor, MI, USA, 1999.
- [184] K. Topalli, M. Unlu, S. Demir, O. A. Civi, S. Koc, and T. Akin, "New approach for modelling distributed MEMS transmission lines," *IEE. Proc. Microwaves Antennas and Propagation*, vol. 153, no. 2, pp. 152-162, Apr. 2006.
- [185] J. B. Muldavin, "Design and analysis of series and shunt MEMS switches," Ph. D. dissertation, Dept. of Electrical Eng., The Univ. of Michigan, Ann Arbor, MI, USA, 2001.

# APPENDIX

# **DETAILED FABRICATION PROCESS FLOW**

This appendix presents the detailed fabrication flow of the METU RF MEMS Fabrication process. The entire fabrication process is done at the fabrication facilities of the METU MEMS center on 4" quartz substrate.

STEP NO	PROCESS	SPECIFICATION
1	WAFER CLEANING	1. Soak in piranha solution ( $H_2SO_4$ : $H_2O_2$ 1:1.2), Duration: 2 × 15 min 2. DI-water rinse (min. 3 × 10 min cycles in automated rinser). 3. Dry by $N_2$ .
2	SURFACE FORMATION	1. Soak in BHF, Duration: 30 s.
		2. DI-water rinse (min. $3 \times 30$ s cycles in automated rinser). 3. Dry by N <sub>2</sub> .
3	SiCr SPUTTERING	Dehydration, N <sub>2</sub> , 110°C, 30 min.
	Resistive layer deposition	Power: 100 W, Rate: 0.05/0.1 A/s.
		Pressure: $10 \times 10^{\circ}$ mBar, Flow: 12.6 sccm Ar.
		Temperature: 120 °C, Distance: 130 mm.
		nme: 4000 s, Endpoint: 200 A.
		Weasured Unickness. 1000 A.
4	LITHOGRAPHY	1. Dehydration, $N_2$ , 110°C, 10 min.
	MASK #1 for SiCr Etch	2. Spin Primer (HMDS), 500 rpm 7 sec. + 4000 rpm 30 sec.
		3. Spin S1813 PR, 500 rpm 7 sec. + 4000 rpm, 30 sec.
		4. Softbake 115°C, 70 s, hotplate.
		5. EBR @ 2000 rpm.
		<ol> <li>Align &amp; Expose, 4.2 sec @ 13 mW/cm<sup>2</sup> (Constant intensity).</li> </ol>
		7. Develop, MF-319, 60 s.
		8. DI-water rinse, 1:30 min + 1:30 min.
		9. Dry by N <sub>2</sub> .
		10. Inspect alignment marks.
		11. Hardbake @ 110°C, 10 min, N <sub>2</sub> , oven.
		12. O <sub>2</sub> plasma @ 300 W, 0.4 Torr, 30 s.
5	SiCr ETCH	1. Etch in BHF, (Duration depends on the layer, $\sim$ 3-6 min).
	Resistive layer patterning	2. DI-water rinse (min. $3 \times 30$ s cycles in automated rinser).
		3. Dry by N <sub>2</sub>

		4. Inspection (Check the metals on the anchors)
6	PHOTORESIST STRIP	<ol> <li>Soak in ACE (10 min) or wash the wafer with ACE.</li> <li>Soak in IPA (5 min) or wash the wafer with IPA.</li> <li>Soak in SVC-175 baths @ 80°C, Duration: 2 × 10 min.</li> <li>DI-water rinse (min. 5 × 10 min cycles in automated rinser).</li> <li>Dry by N<sub>2</sub>.</li> <li>Inspection.</li> </ol>
7a	<b>Ti SPUTTERING</b> First metal layer deposition	Dehydration, N <sub>2</sub> , 110°C, 30 min. Power: 300 W, Rate: 1.1 Å/s. Pressure: 2 × 10 <sup>-3</sup> mBar, Flow: 2.1 sccm Ar. Temperature: 150 °C, Distance: 130 mm. Time: 75 s, Endpoint: 80 Å. Measured thickness: 200 Å.
7b	Au SPUTTERING First metal layer deposition	Power: 100 W, Rate: 1.1 Å/s. Pressure: 20 × 10 <sup>-3</sup> mBar, Flow: 25 sccm Ar. Temperature: 150 °C, Distance: 130 mm. Time: 2000 s, Endpoint: 2500 Å. Measured thickness: 9000 Å.
8	LITHOGRAPHY MASK #2 for Ti/Au Etch	<ol> <li>Dehydration, N<sub>2</sub>, 110°C, 10 min.</li> <li>Spin Primer(HMDS), 500 rpm 7 sec. + 4000 rpm 30 sec.</li> <li>Spin SPR220-3 PR, 500 rpm 7 sec. + 4000 rpm, 30 sec.</li> <li>Softbake 115°C, 90 s, hotplate.</li> <li>EBR @ 2000 rpm.</li> <li>Align &amp; Expose, 12.5 sec @ 13 mW/cm<sup>2</sup> (Constant intensity).</li> <li>Develop, MF-24A, 90 s.</li> <li>DI-water rinse, 1:30 min + 1:30 min.</li> <li>Dry by N<sub>2</sub>.</li> <li>Inspect alignment marks.</li> <li>Hardbake @ 110°C, 10 min, N<sub>2</sub>, oven.</li> <li>O<sub>2</sub> plasma @ 300 W, 0.4 Torr, 30 s.</li> </ol>
9a	Au ETCH First metal layer patterning	<ol> <li>Etch in Au Etchant, Duration: ~2 min.</li> <li>DI-water rinse (min. 3 × 30 s cycles in automated rinser).</li> <li>Dry by N<sub>2</sub>.</li> <li>Inspection.</li> </ol>
9b	<b>Ti ETCH</b> First metal layer patterning	<ol> <li>Etch in Ti Etchant (HF:H<sub>2</sub>O<sub>2</sub>:DI 1:1:640), Duration: ~1 min.</li> <li>DI-water rinse (min. 3 × 30 s cycles in automated rinser).</li> <li>Dry by N<sub>2</sub>.</li> <li>Inspection.</li> </ol>
10	PHOTORESIST STRIP	<ol> <li>Soak in ACE (10 min) or wash the wafer with ACE.</li> <li>Soak in IPA (5 min) or wash the wafer with IPA.</li> <li>Soak in SVC-175 baths @ 80°C, Duration: 2 × 10 min.</li> <li>DI-water rinse (min. 5 × 10 min cycles in automated rinser).</li> <li>Dry by N<sub>2</sub>.</li> <li>Inspection.</li> </ol>
11a	Ti SPUTTERING Backside metal deposition	Dehydration, N₂, 110°C, 30 min. Power: 300 W, Rate: 1.1 Å/s. Pressure: 2 × 10 <sup>-3</sup> mBar, Flow: 2.1 sccm Ar. Temperature: RT, Distance: 130 mm. Time: 75 s, Endpoint: 80 Å.

Measured thickness: 200 Å.

11b	Au SPUTTERING Backside metal deposition	Power: 300 W, Rate: 4 Å/s. Pressure: 20 × 10 <sup>-3</sup> mBar, Flow: 25 sccm Ar. Temperature: 150 °C, Distance: 130 mm. Time: 150 s, Endpoint: 600 Å. Measured thickness: 2000 Å.
12	PECVD SIN DEPOSITION	Dehydration, N <sub>2</sub> , 120°C 30 min. MF, stress free recipe @ 300 °C, Duration: 23 min. Meas. thickness: 2900 Å
13	LITHOGRAPHY MASK #3 for SiN RIE	<ol> <li>Dehydration, N<sub>2</sub>, 110°C, 10 min.</li> <li>Spin Primer(HMDS), 500 rpm 7 sec. + 4000 rpm 30 sec.</li> <li>Spin SPR220-3 PR, 500 rpm 12.5 sec. + 4000 rpm, 30 sec.</li> <li>Softbake 115°C, 90 s, hotplate.</li> <li>EBR @ 2000 rpm.</li> <li>Align &amp; Expose, 12.5 sec @ 13 mW/cm<sup>2</sup> (Constant intensity).</li> <li>Develop, MF-24A, 90 s.</li> <li>DI-water rinse, 1:30 min + 1:30 min.</li> <li>Dry by N<sub>2</sub>.</li> <li>Inspect alignment marks.</li> <li>Hardbake @ 110°C, 10 min, N<sub>2</sub>, oven.</li> <li>O<sub>2</sub> plasma @ 300 W, 0.4 Torr, 30 s.</li> </ol>
14	SiN RIE Isolation layer patterning	Pressure: 70 mTorr, Power: 250 W. Flow: 60 sccm CF <sub>4</sub> , 22 sccm CHF <sub>3</sub> , 10 sccm O <sub>2</sub> . Recipe: 1_SiN, Duration: 60 s.
15	PHOTORESIST STRIP	1. Soak in ACE (10 min) or wash the wafer with ACE (Mandatory). 2. Soak in IPA (5 min) or wash the wafer with IPA (Mandatory). 3. Soak in SVC-175 baths @ 80°C, Duration: $2 \times 10$ min. 4. DI-water rinse (min. $5 \times 10$ min cycles in automated rinser). 5. Dry by N <sub>2</sub> . 6. Inspection.
16	LITHOGRAPHY MASK #4 for Sacrificial Layer Formation	<ol> <li>Dehydration, N<sub>2</sub>, 110°C, 10 min.</li> <li>Spin PI2737 polyimide, 500 rpm 7 sec. + 4500 rpm, 30 sec.</li> <li>Softbake 75°C, 4:30 min hotplate.</li> <li>Align &amp; Expose, 20 sec @ 13 mW/cm<sup>2</sup> (Constant intensity).</li> <li>Develop in DE9040, 2:30 min.</li> <li>Rinse in PA401R, 30 s.</li> <li>Dry by N<sub>2</sub>.</li> <li>Inspect alignment marks.</li> <li>Skip O<sub>2</sub> plasma.</li> <li>Curing: Ramp 50-to-220 °C, hold @ 220°C for 115 min, Cool down to 50° C in 20 min.</li> </ol>
17	Au SPUTTERING Structural metal layer deposition	Dehydration, N₂, 110°C, 10 min. Power: 213 W, Rate: 2.7 Å/s. Pressure: 10 × 10 <sup>-3</sup> mBar, Flow: 22 sccm Ar. Temperature: RT, Distance: 130 mm. Time: 4 × 300 s, Endpoint: 4000 Å. Measured thickness: 11000 Å.
18	LITHOGRAPHY MASK #5 for <b>Anchor Fill</b>	1. Dehydration, N <sub>2</sub> , 110°C, 10 min. 2. Spin Primer(HMDS), 500 rpm 7 sec. + 2000 rpm 30 sec.
	Electroplating	<ol> <li>Spin SPR220-7 PR, 500 rpm 7 sec. + 2000 rpm, 30 sec.</li> <li>Softbake 115°C, 90 s, hotplate.</li> <li>EBR @ 2000 rpm.</li> <li>Align &amp; Expose, 55 sec @ 13 mW/cm<sup>2</sup> (Constant intensity).</li> <li>Develop, MF-24A, 2 min.</li> <li>DI-water rinse, 1:30 min + 1:30 min.</li> <li>Dry by N<sub>2</sub>.</li> <li>Inspect alignment marks.</li> <li>Hardbake @ 110°C, 10 min, N<sub>2</sub>, oven.</li> <li>O<sub>2</sub> plasma @ 300 W, 0.4 Torr, 30 s.</li> </ol>
----	--	--
19	Au ELECTROPLATING Strengthening layer selective deposition	Average current: 6 mA DC, Plating area: 2.53 cm <sup>2</sup> . Duty cycle: 2 ms ON/98 ms OFF. Pump: ON, Reciprocator: OFF, Temperature: 32 °C. Duration: 20 min.
20	PHOTORESIST STRIP	1. Soak in ACE (10 min) or wash the wafer with ACE. 2. Soak in IPA (5 min) or wash the wafer with IPA. 3. DI-water rinse (min. $3 \times 10$ min cycles in automated rinser). 4. Dry by N <sub>2</sub> . 5. Inspection.
21	LITHOGRAPHY MASK #6 for Structural Layer Etch	<ol> <li>Dehydration, N<sub>2</sub>, 110°C, 10 min.</li> <li>Spin Primer(HMDS), 500 rpm 7 sec. + 2000 rpm 30 sec.</li> <li>Spin SPR220-7 PR, 500 rpm 7 sec. + 2000 rpm, 30 sec.</li> <li>Softbake 115°C, 90 s, hotplate.</li> <li>EBR @ 2000 rpm.</li> <li>Align &amp; Expose, 55 sec @ 13 mW/cm<sup>2</sup> (Constant intensity).</li> <li>Develop, MF-24A, 2 min.</li> <li>DI-water rinse, 1:30 min + 1:30 min.</li> <li>Dry by N<sub>2</sub>.</li> <li>Inspect alignment marks.</li> <li>Hardbake @ 110°C, 10 min, N<sub>2</sub>, oven.</li> <li>O<sub>2</sub> plasma @ 300 W, 0.4 Torr, 30 s.</li> </ol>
22	Au ETCH Structural layer patterning	<ol> <li>Etch in Au Etchant, Duration: ~2:30 min.</li> <li>DI-water rinse (min. 3 × 30 s cycles in automated rinser).</li> <li>Dry by N<sub>2</sub>.</li> <li>Inspection.</li> </ol>
23	<b>Ti ETCH</b> Backside metal stripping	<ol> <li>Etch in Ti Etchant (HF:H<sub>2</sub>O<sub>2</sub>:DI 1:1:640), Duration: ~1 min.</li> <li>DI-water rinse (min. 3 × 30 s cycles in automated rinser).</li> <li>Dry by N<sub>2</sub>.</li> <li>Inspection.</li> </ol>
24	DICING	Separating the samples by dicing.
25	SACRIFICAL LAYER ETCH	<ol> <li>Sacrificial layer wet etching in EKC-265 @ 50 °C Stirrer @ 250 rpm, Duration: 16 h.</li> <li>DI-water rinse (min. 2 × 30 min).</li> <li>IPA rinse (min. 2 × 15 min).</li> </ol>
26	CRITICAL POINT DRYING	<ol> <li>Release the structures using critical point drying.</li> <li>Inspection both by optical microscope and optical profiler.</li> </ol>

# **CURRICULUM VITAE**

## **Personal Information**

Surname, Name	ÜNLÜ, Mehmet
Date of birth Place of birth	15.10.1979 Ankara/TURKEY
Nationality	Turkish (T.C.)
Current Position	<i>Researcher,</i> METU-MEMS Center, Middle East Technical University, Ankara, TURKEY
Phone	+90 312 210 23 40 / +90 533 540 39 40
Home Address	Ziya Bey Cad. Petrol Sit. 21/27 Balgat, 06520, Ankara

## Education

2003-	Ph. D. Department of Electrical and Electronics Engineering, Middle East Technical University, Ankara, TURKEY (Cum. GPA: 3.81/4.00)
2001-2003	MSc. Department of Electrical and Electronics Engineering, Middle East Technical University, Ankara, TURKEY (Cum. GPA: 3.74/4.00)
1997-2001	B.Sc. Department of Electrical and Electronics Engineering, Middle East Technical University, Ankara, TURKEY (Cum. GPA: 3.11/4.00)
1996-1997	Ankara Ömer Seyfettin Lisesi (Cum. GPA: 4.51/5.00)
1990-1996	Ankara Gazi Anadolu Lisesi

#### **Work Experience**

Oct. 2008-	<i>Researcher,</i> METU-MEMS Center, Middle East Technical University, Ankara, TURKEY		
Jul. 2001- Sep. 2008	<i>Research and Teaching Assistant</i> , Department of Electrical and Electronics Engineering, Middle East Technical University, Ankara, TURKEY		
Sep.2000- Jul. 2001	Part time engineer, ASELSAN A.Ş, Ankara, TURKEY		
Summer 2000	Intern, TELSIM A.Ş., İstanbul, TURKEY		
Summer 1999	Intern, ASELSAN A.Ş, Ankara, TURKEY		

#### **Foreign Languages**

English (Fluent), German (Basic)

### Hobbies

Playing Ud, playing basketball, Classical Turkish Music, and travelling.

### **International Journal Publications**

- 1. E. Erdil, K. Topalli, **M. Unlu**, O. Aydin Civi, and T. Akin, "Frequency tunable microstrip patch antenna using RF MEMS technology," *IEEE Trans. Antennas and Propagation*, vol.55, no.4, pp. 1193-1196, Apr. 2007.
- 2. K. Topalli, **M. Unlu**, S. Demir, O. Aydin Civi, S. Koc, and T. Akin, "A new approach for modelling Distributed MEMS Transmission Lines," *IEE Proc. Microwaves, Antennas & Propagation*, vol. 153, pp. 152-162, Apr. 2006.

### **International Conference Publications**

 K. Topalli, O. Bayraktar, M. Unlu, O. Aydin Civi, S. Demir, and T. Akin, "Reconfigurable antennas by RF MEMS technology," 30<sup>th</sup> ESA Antenna Workshop on Antennas for Earth Observation, Science, Telecommunications and Navigation Space Missions, pp. 497-500, Noordwijk, The Netherlands, 27-30 May 2008.

- K. Topalli, M. Unlu, I. Istanbulluoglu, H. I. Atasoy, O. Bayraktar, O. Aydin Civi, Ş. Demir, S. Koc, and T. Akin, "Reconfigurable RF MEMS devices and antennas," 5<sup>th</sup> International Conference on Electrical and Electronics Engineering (ELECO 2007), pp.204-209, Bursa, Turkey, 5-9 Dec. 2007.
- M. Unlu, H. I. Atasoy, S. Demir, O. Aydin Civi, S. Koc, and T. Akin, "Double stub and triple stub RF MEMS impedance tuners," *Proceedings of 8<sup>th</sup> International Symposium on RF MEMS and RF Microsystems*, MEMSWAVE, Barcelona, Spain, pp. 169-172, 26-29 Jun. 2007.
- 4. O. Bayraktar, K. Topalli, **M. Unlu**, O. Aydin Civi, S. Demir, and T. Akin, "Beam switching reflectarray using RF MEMS technology," *European Conference on Antennas and Propagation*, Edinburgh, UK, 11-16 Nov. 2007, pp. 1-6.
- O. Bayraktar, K. Topalli, M. Unlu, I. Istanbulluoglu, E. U. Temocin, H. I. Atasoy, O. A. Civi, S. Demir, S. Koc, T. Akin, "Reconfigurable reflectarray using RF MEMS technology," *European Conference on Antennas and Propagation*, Nice, France, 6-10 Nov. 2006, on CD-ROM.
- E. Erdil, K. Topalli, M. Unlu, I. Istanbulluoglu, E. U. Temocin, H. I. Atasoy, O. Bayraktar, O. A. Civi, S. Demir, S. Koc, T. Akin, "Reconfigurable slot antenna with fixed-fixed beam type RF MEMS capacitors," *European Conference on Antennas and Propagation*, Nice, France, 6-10 Nov. 2006.
- K. Topalli, M. Unlu, O. Aydin Civi, S. Demir, S. Koc, and T. Akin, "A monolithic phased array using 3-bit RF MEMS phase shifters," 2006 IEEE AP-S International Symposium and USNC/URSI National Radio Science Meeting, Albuquerque, New Mexico, USA, pp. 517-520, Jul. 2006.
- M. Unlu, K. Topalli, H. I. Atasoy, E. U. Temocin, I. Istanbulluoglu, O. Bayraktar, S. Demir, O. A. Civi, S. Koc, and T. Akin, "A tunable RF MEMS triple stub impedance matching network," *European Microwave Conference 2006*, Manchester, United Kingdom, pp. 1370-1373, 10-15 Sep. 2006.
- H. I. Atasoy, K. Topalli, M. Unlu, I. Istanbulluoglu, O. Bayraktar, E. U. Temocin, S. Demir, O. A. Civi, S. Koc, and T. Akin, "Investigation of on-wafer TRL calibration accuracy dependence on transitions and probe positioning," *European Microwave Conference 2006*, Manchester, United Kingdom, pp. 1582-1585, 10-15 Sep. 2006.
- M. Secmen, K. Topalli, M. Unlu, E. Erdil, O. Aydin Civi, and A. Hizal; "Dual wideband antenna analysis for linear FMCW radar applications," 2006 IEEE 14<sup>th</sup> Signal Processing and Communications, pp. 1-4, 17-19 Apr. 2006, Antalya, Turkey

- H. I. Atasoy, K. Topalli, M. Unlu, I. Istanbulluoglu, E. U. Temocin, O. Bayraktar, S. Demir, O. Aydin-Civi, S. Koc, and T. Akin, "Metal-to-metal and capacitive contact RF MEMS shunt switch structures," 7<sup>th</sup> Workshop on MEMS for MilimeterWave Communications (MEMSWAVE), pp. 154-156, Orvieto, Italy, 27-30 Jun. 2006.
- M. Unlu, K. Topalli, H. I. Atasoy, E. U. Temocin, I. Istanbulluoglu, O. Bayraktar, S. Demir, O. Aydin-Civi, S. Koc, and T. Akin, "A reconfigurable RF MEMS impedance tuner for impedance matching of a two-stage low noise amplifier," 7<sup>th</sup> Workshop on MEMS for MilimeterWave Communications (MEMSWAVE), pp. 115-118, Orvieto, Italy, 27-30 Jun. 2006.
- S. Onat, M. Unlu, L. Alatan, S. Demir, and T. Akin, "Design of a re-configurable dual frequency microstrip antenna with integrated RF MEMS switches," *IEEE Antennas and Propagation Society International Symposium*, Volume: 2A, pp. 384- 387, 3-8 Jul. 2005.
- E. Erdil, S. Onat, K. Topalli, I. Istanbulluoglu, M. Unlu, S. Demir, L. Alatan, O. Aydin Civi, S. Koc, and T. Akin, "Reconfigurable antenna structures using MEMS technology," 6<sup>th</sup> Workshop on MEMS for MilimeterWave Communications (MEMSWAVE), pp. 35-38, Switzerland, 23-24 Jun. 2005.
- M. Unlu, K. Topalli, S. Demir, O. Aydin Civi, S. Koc, and T. Akin, "A Parametric Study On Modeling Distributed Mems Transmission Lines", 34<sup>th</sup> European Microwave Conference, pp.1157-1160, Amsterdam, The Netherlands, Oct. 11-15, 2004.
- M. Unlu, K. Topalli, S. Demir, O. Aydin Civi, S. Koc, and T. Akin, "Design and modeling of a reconfigurable triple stub impedance matching network using Distributed MEMS Transmission Lines," 5<sup>th</sup> Workshop on MEMS for MilimeterWave Communications (MEMSWAVE), pp. 117-120, Uppsala, Sweden, 30 Jun.-2 Jul. 2004.
- K. Topalli, M. Unlu, S. Demir, O. Aydin Civi, S. Koc, and T. Akin, "A parametric modeling study on Distributed MEMS Transmission Lines," 5<sup>th</sup> Workshop on MEMS for MilimeterWave Communications (MEMSWAVE), pp. 13-16 Uppsala, Sweden, 30 Jun.-2 Jul. 2004.
- K. Topalli, M. Unlu, H. Sagkol, S. Demir, O. Aydin Civi, S. Koc, and T. Akin, "A new model for Distributed MEMS Transmission Lines," 2004 AP-S International Symposium and USNC/URSI National Radio Science Meeting, pp. 2855-2858, Monterey, California, 20-26 Jun. 2004.
- 19. **M. Unlu**, K. Topalli, H. Sagkol, S. Demir, O. Aydin Civi, S. Koc, and T. Akin, "New MEMS switch structures for antenna applications," *2002 IEEE AP-S International*

*Symposium and USNC/URSI National Radio Science Meeting*, p. 134, San Antonio, Texas, 16-21 Jun. 2002.

- K. Topalli, H. Sagkol, M. Unlu, S. Onat, O. Aydin Civi, S. Demir, S. Koc, L. Alatan, and T. Akin, "Microwave circuit components and reconfigurable antennas using RF MEMS technology," *Proceeding of 4<sup>th</sup> COST284 Management Committee Meeting & Workshop with INICA*, Berlin, Germany, 19 Sep. 2003.
- M. Unlu, K. Topalli, H. Sagkol, S. Demir, O. Aydin Civi, S. Koc, and T. Akin, "RF MEMS adjustable impedance matching network and adjustable power divider," 2002 IEEE AP-S International Symposium and USNC/URSI National Radio Science Meeting, pp. 26-29, San Antonio, Texas, 16-21 Jun. 2002.
- H. Sagkol, K. Topalli, M. Unlu, O. Aydin Civi, S. Koc, S. Demir, and T. Akin, "A monolithic phased array with RF MEMS technology," 2002 IEEE AP-S International Symposium and USNC/URSI National Radio Science Meeting, pp. 760-763, San Antonio, Texas, 16-21 Jun. 2002.

#### **National Conference Publications**

- E. U. Temoçin, K. Topallı, M. Ünlü, I. Istanbulluoğlu, H. I. Atasoy, Ö. Bayraktar, Ş. Demir, Ö. Aydın Çivi, S. Koç ve T. Akın, "Eşdüzlemsel dalga klavuzundan mikroşerit hatta geçis üzerine çalısmalar," National URSI Symposium Turkey, pp. 448-450, Hacettepe, Ankara, Türkiye, 6-8 Eylül 2006 (in Turkish).
- Ö. Bayraktar, K. Topallı, M. Ünlü, I. Istanbulluoğlu, E. U. Temoçin, H. I. Atasoy, Ş. Demir, Ö. Aydın Çivi, S. Koç ve T. Akın, "RF MEMS teknolojisi ile yansıtıcı dizi anteni tasarımı," National URSI Symposium Turkey, pp. 438-440, Hacettepe, Ankara, Türkiye, 6-8 Eylül 2006 (in Turkish).
- H. I. Atasoy, M. Ünlü, K. Topallı, I. Istanbulluoğlu, E. U. Temoçin, Ö. Bayraktar, Ş. Demir, Ö. Aydın Çivi, S. Koç ve T. Akın, "TRL kalibrasyonunda uç konumlandırmasi ve geçislerin etkileri," National URSI Symposium Turkey, pp. 426-428, Hacettepe, Ankara, Türkiye, 6-8 Eylül 2006 (in Turkish).
- I. Istanbulluoğlu, K. Topallı, E. U. Temoçin, H. I. Atasoy, M. Ünlü, Ö. Bayraktar, Ş. Demir, Ö. Aydın Çivi, S. Koç ve T. Akın, "RF MEMS teknolojisi kullanılarak tasarlanmış ayarlanabilir bant geçiren bağlanmış-hat süzgeç," National URSI Symposium Turkey, pp. 416-418, Hacettepe, Ankara, Türkiye, 6-8 Eylül 2006 (in Turkish).
- 5. K. Topallı, **M. Ünlü**, Ö. Aydın Çivi, Ş. Demir, S. Koç ve T. Akın, "RF MEMS teknolojisi ile tek parça faz dizili anten yapısı," National URSI Symposium Turkey, pp. 413-415, Hacettepe, Ankara, Türkiye, 6-8 Eylül 2006 (in Turkish).

- H. I. Atasoy, K. Topalli, M. Ünlü, I. Istanbulluoğlu, E. U. Temoçin, Ö. Bayraktar, Ş. Demir, Ö. Aydın Çivi, S. Koç ve T. Akın, "RF MEMS teknolojisi ile kapasitif ve metal-metal bağlantılı paralel anahtar yapıları", National URSI Symposium Turkey, pp. 410-412, Hacettepe, Ankara, Türkiye, 6-8 Eylül 2006 (in Turkish).
- M. Ünlü, K. Topallı, Ş. Demir, Ö. Aydın Çivi, S. Koç ve T. Akın, "RF MEMS teknolojisi ile ayarlanabilir empedans uyumlama devresi," National URSI Symposium Turkey, pp. 407-409, Hacettepe, Ankara, Türkiye, 6-8 Eylül 2006 (in Turkish).
- 8. E. Erdil, K. Topallı, **M. Ünlü**, Ö. Aydın Çivi ve T. Akın, "RF MEMS kapasitörler ile rezonans frekansı ayarlanabilen antenler," National URSI Symposium Turkey, pp. 404-406, Hacettepe, Ankara, Türkiye, 6-8 Eylül 2006 (in Turkish).
- K. Topalli, M. Ünlü, Ş. Demir, Ö. Aydın Çivi, S. Koç ve T. Akın, "RF MEMS teknolojisi ile mikroşerit ve eşdüzlemsel dalga kılavuzu hat üstünde paralel anahtar yapıları," National URSI Symposium Turkey, pp. 370-372, 8-10 September 2004 (in Turkish).
- 10. **M. Ünlü**, K. Topallı, Ş. Demir, Ö. Aydın Çivi, S. Koç ve T. Akın, "Sıralanmış MEMS iletim hatları için yeni bir model," National URSI Symposium Turkey, pp. 435-437, 8-10 September 2004 (in Turkish).
- 11. K. Topallı, H. Sağkol, **M. Ünlü**, Ö. Aydin Çivi, S. Demir, S. Koç ve T. Akın, "RF MEMS teknolojisi ile faz dizili anten," National URSI Symposium Turkey, pp. 134-137, September 2002 (in Turkish).
- H. Sağkol, M. Ünlü, K. Topallı, Ö. Aydin Çivi, Ş. Demir, S. Koç ve T. Akın, "RF MEMS teknolojisi ile mikrodalga faz kaydırıcıları," National URSI Symposium Turkey, pp. 138-141, September 2002 (in Turkish).
- M. Ünlü, H. Sağkol, K. Topallı, Ö. Aydin Çivi, Ş. Demir, S. Koç ve T. Akın, "RF MEMS teknolojisi ile ayarlanabilir empedans uyumlama ve güç bölücü devreleri," National URSI Symposium Turkey, pp. 142-145, September 2002 (in Turkish).
- 14. **M. Ünlü**, H. Sağkol, K. Topallı, Ö. Aydin Çivi, Ş. Demir, S. Koç ve T. Akın, "Anten uygulamaları için yeni MEMS anahtar yapıları," National URSI Symposium Turkey, pp. 146-149, September 2002 (in Turkish).